



University of Groningen

Distributed Linear Quadratic Control and Filtering

Jiao, Junjie

DOI: 10.33612/diss.135590405

IMPORTANT NOTE: You are advised to consult the publisher's version (publisher's PDF) if you wish to cite from it. Please check the document version below.

Document Version Publisher's PDF, also known as Version of record

Publication date: 2020

Link to publication in University of Groningen/UMCG research database

Citation for published version (APA): Jiao, J. (2020). Distributed Linear Quadratic Control and Filtering: a suboptimality approach. University of Groningen. https://doi.org/10.33612/diss.135590405

Copyright

Other than for strictly personal use, it is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license (like Creative Commons).

The publication may also be distributed here under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license. More information can be found on the University of Groningen website: https://www.rug.nl/library/open-access/self-archiving-pure/taverneamendment.

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Downloaded from the University of Groningen/UMCG research database (Pure): http://www.rug.nl/research/portal. For technical reasons the number of authors shown on this cover page is limited to 10 maximum.

Distributed Linear Quadratic Control and Filtering:

a suboptimality approach

Junjie Jiao



The research described in this dissertation has been carried out at the Faculty of Science and Engineering (FSE), University of Groningen, The Netherlands, within the Bernoulli Institute for Mathematics, Computer Science and Artificial Intelligence.



The research described in this dissertation is part of the research program of the Dutch Institute of Systems and Control (DISC). The author has successfully completed the educational program of the Graduate School DISC.



The research reported in this dissertation was financially supported by the China Scholarship Council (CSC), the Chinese Ministry of Education.

Printed by Ipskamp Printing Cover designed by Qingqing Cai



Distributed Linear Quadratic Control and Filtering:

a suboptimality approach

PhD thesis

to obtain the degree of PhD at the University of Groningen on the authority of the Rector Magnificus Prof. C. Wijmenga and in accordance with the decision by the College of Deans.

This thesis will be defended in public on

Friday 23 October 2020 at 12.45 hours

by

Junjie Jiao

born on 22 July 1993 in Anhui, China

Supervisors

Prof. dr. H.L. Trentelman Prof. dr. M.K. Camlibel

Assessment Committee

Prof. dr. C.W. Scherer Prof. dr. J. Lunze Prof. dr. A.J. van der Schaft To my parents

献给我亲爱的父母, 焦圣中、李敏

Acknowledgment

This thesis would not have existed without the help and support of many people. I would like to take this opportunity to express my gratitude and appreciation to them.

First and foremost, I would like to express my utmost gratitude to my supervisors, Harry Trentelman and Kanat Camlibel, for all their kind support and careful guidance in the last four years. Harry, I have learned so much from you, from doing research to writing papers, response letters and reviews. I wish I would have also learned from you a bit of a good sense of humor. You have changed the way that I view research and life, and for this I am eternally grateful. Kanat, you are one of the sharpest minds I have ever met. You have always been patient, supportive and encouraging. I have also enjoyed a lot from our scientific discussions as well as casual conversations. I am incredibly fortunate to have you as my co-supervisor.

I would also like to express my thanks to the members of the reading committee, Professor Carsten Scherer, Professor Jan Lunze and Professor Arjan van der Schaft, for reading the draft of my thesis and making valuable comments.

I thank Jaap and Daili for accepting to be my paranymphs. Jaap, thank you for all your kind help. I also had a lot of fun for our 'intense competition' while playing Diplomacy. Daili, I really enjoyed conquering the Tour du Mont Blanc with you. Thanks also for being a reliable partner for running, swimming and bouldering, which makes my life in Groningen more exciting.

I am also grateful to Henk and Tessa for translating the summary of my thesis into Dutch. Thank you also for inviting me to your wedding. It was a wonderful experience for me to be part of it. Best wishes for you both! A special debt is owned to Qingqing Cai for designing the thesis cover for me. I wish you a lot of happiness and success in your future career!

Of course I also want to thank my colleagues of SCAA, DTPA and SMS groups. Bart, many thanks for inviting me to be the co-organizer for the SCAA colloquia. Stephan, thank you very much for your insightful comments on my research. Jiajia, it was great to share this 'PhD burden' with you, and congratulations with your faculty position! Tjerk, it was my honor to be one of your paranymphs (together with Ronald). Mark, thanks for building the 'Fish-day' website. I miss it already while writing down this line. I would also like to thank all other members of SCAA, DTPA and SMS groups, I very much enjoyed and learned a lot from our lunch breaks and chat from time to time. Also, I want to thank our secretaries for all their kind help. In particular, I want to thank Ineke and Elina.

I would also like to thank my girlfriend Ziwei for her incredible patience and invaluable support during the last year. I enjoy every moment we have spent together, and I look forward to what the future holds for us. I want to express my deepest gratitude to my parents for their unconditional and endless support and encouragement during the last 27 years. (Hopefully, they would be willing to continue supporting me unconditionally.) I would like to say some more words to them in Chinese, since they are still practicing Chinese:

Finally, I want to close the acknowledgment by thanking everyone that has encountered in my life.

Junjie Jiao Groningen September 22, 2020

Contents

A	cknov	vledgements	vi				
1	Intr	oduction	1				
	1.1	Background	1				
	1.2	Outline of this thesis	5				
	1.3	Publications during the PhD project	6				
	1.4	Notation	7				
	1.5	Graph theory	7				
2	A suboptimality approach to distributed linear quadratic optimal con-						
	trol		11				
	2.1	Introduction	11				
	2.2	Problem formulation	13				
	2.3	Linear quadratic suboptimal control for linear systems	15				
		2.3.1 Quadratic performance analysis for autonomous linear sys-					
		tems	16				
		2.3.2 Linear quadratic suboptimal control for linear systems	17				
	2.4	Distributed linear quadratic suboptimal control for multi-agent					
		systems	19				
	2.5	Simulation example	26				
	2.6	Conclusions	29				
3	Distributed linear quadratic tracking control: a suboptimality approach						
	3.1	Introduction	31				
	3.2	Quadratic performance analysis for autonomous linear systems	33				
	3.3	Problem formulation	33				
	3.4	Distributed suboptimal tracking control for leader-follower multi-					
		agent systems	36				
	3.5	Simulation example	41				
	3.6	Conclusions	44				

4	Dist	ributed linear quadratic control: compute locally and act globally	47				
	4.1	Introduction	47				
	4.2	The general form of a distributed linear quadratic cost functional .	49				
	4.3	Centralized computation of the optimal control gain	52				
	4.4	Towards decentralized computation	53				
	4.5	The discounted linear quadratic tracking problem	54				
	4.6	Consensus analysis	56				
	4.7	Simulation example	60				
	4.8	Conclusions	62				
5	A suboptimality approach to distributed \mathcal{H}_2 optimal control by state						
	feed	lback	63				
	5.1	Introduction	63				
	5.2	Problem formulation	65				
	5.3	\mathcal{H}_2 suboptimal control for linear systems by static state feedback .	67				
		5.3.1 H_2 performance analysis for linear systems with disturbance					
		inputs	67				
		5.3.2 H_2 suboptimal control for linear systems with control and					
		disturbance inputs	68				
	5.4	Distributed \mathcal{H}_2 suboptimal protocols by static state feedback	70				
	5.5	Conclusions	75				
6	Dist	ributed \mathcal{H}_2 suboptimal control by dynamic output feedback	77				
	6.1	Introduction	77				
	6.2	Problem formulation	79				
	6.3	\mathcal{H}_2 suboptimal control for linear systems by dynamic output feedback	82				
	6.4	Distributed \mathcal{H}_2 suboptimal protocols by dynamic output feedback .	86				
	6.5	Simulation example	92				
	6.6	Conclusions	94				
7	\mathcal{H}_2 suboptimal output synchronization of heterogeneous multi-agent						
	syst	ems	95				
	7.1	Introduction	95				
	7.2	Problem formulation	97				
	7.3	Preliminary results	.00				
		7.3.1 \mathcal{H}_2 suboptimal control for linear systems by dynamic output					
		feedback	.00				
		7.3.2 Output synchronization of heterogeneous linear multi-agent					
		systems	.02				
	7.4	\mathcal{H}_2 suboptimal output synchronization protocols by dynamic out-	<u> </u>				
		put teedback	.03				

	7.5	\mathcal{H}_2 suboptimal output synchronization protocols by state feedback	107		
	7.6	Simulation example	109		
	7.7	Conclusions	113		
8	\mathcal{H}_2 a	and \mathcal{H}_∞ suboptimal distributed filter design for linear systems	115		
	8.1	Introduction	115		
	8.2	Preliminaries	117		
		8.2.1 Detectability and detectability decomposition	117		
		8.2.2 \mathcal{H}_2 and \mathcal{H}_∞ performance of linear systems	118		
	8.3	Problem formulation	119		
	8.4	\mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filter design $\ldots \ldots \ldots \ldots$	122		
		8.4.1 \mathcal{H}_2 suboptimal distributed filter design	127		
		8.4.2 \mathcal{H}_{∞} suboptimal distributed filter design	131		
	8.5	Simulation example	133		
	8.6	Conclusions	138		
9	Con	clusions and future research	139		
	9.1	Conclusions	139		
	9.2	Future research	141		
Bi	Bibliography				
Su	Summary				
Sa	Samenvatting				

Introduction

1.1 Background

Collective behavior of groups of individuals, for example animals, is ubiquitous in nature [80]. Examples are schooling of fish, flocking of birds, swarming of insects and pack hunting of wolves. These examples of collective animal behavior are believed to have many benefits. For instance, from a *protection-from-predator* perspective, one hypothesis states that fish schools or bird flocks may thwart predators [64]. This is because the many moving preys create a sensory overload on the visual channel of the predators and, consequently, it becomes difficult for the predators to pick out individual prey from the groups. Another hypothesis is that the collective behavior of animals (e.g. fish or birds) may save energy when swimming or flying together [6, 20]. These interesting phenomena have attracted much attention from researchers in many scientific disciplines, ranging from biological science [80] to physics [110], computer science [89] and control engineering [14, 34, 82].

In addition to the fact that we humans desire to understand this beautiful and fascinating collective behavior in nature, there also exist broad practical applications that are urgently calling for investigation in the area of control of interconnected systems. As examples, we mention satellite formation flying [79, 101], intelligent transportation systems [4, 5], distributed sensor networks [81] and power grids [17]. Due to the physical constraints on the interaction between these interconnected systems such as limited computational resources, local communication and local sensing capabilities, control problems in these application areas are very challenging.

Motivated by the above observations, in the past two decades, researchers in the systems and control community have put much effort into studying the problem of *distributed control of multi-agent systems*, see e.g. [11, 34, 54, 60, 85, 87, 94, 103, 111]. In this problem framework, a multi-agent system is a system that consists of a number of local systems, called the *agents*. Each agent exchanges information

with the other agents according to a given communication topology. In this way, these agents together form a *network*. Consequently, the overall behavior of a network is determined not only by the behavior of the agents, but also by the communication between these agents, see e.g. [71, 85, 111]. Such communication topologies are represented by *graphs*. In general, a graph contains *nodes* being connected by *edges*, where each node represents an agent, and the edges represent the communication between these agents. It turns out that tools from algebraic graph theory are useful for tackling problems in the area of multi-agent systems [21, 63].

The essential idea of distributed control for multi-agent systems is that, while each agent makes use of only information obtained from local interactions according to the communication graph, the agents together will still achieve a common goal. Two typical examples are *consensus* and *synchronization*. Within the problem of consensus, the dynamics of the agents is often described by single or double integrators, and by proposing distributed control laws, the agents agree on a certain (possibly nonzero) constant value [29, 54, 85, 87, 111]. On the other hand, in the context of synchronization, the dynamics of the agents is typically characterized by a general higher dimensional linear or nonlinear system, and the proposed distributed protocols guarantee that the states or the outputs of the agents all converge to a common time-varying trajectory [28, 54, 94, 99, 103, 122]. If relative state information of the agents is available, it is often possible to design distributed protocols using static state feedback [10, 85, 87, 111]. However, if the models of the agents are described by higher dimensional dynamics, very often only relative output information is available instead of relative state information. In this case, the controlled multi-agent network may want to achieve synchronization by using dynamic output feedback based distributed protocols, see e.g. [54, 94, 103, 115].

In the literature on consensus and synchronization, based on whether all agents take equally important roles, multi-agent systems can be categorized into two types, namely, *leaderless* multi-agent systems and *leader-follower* multi-agent systems. In the leaderless case, all agents are equally important in the sense that they reach an agreement which depends on the dynamics of all agents, see e.g. [37, 54, 85, 87, 94, 103, 104]. On the other hand, in the leader-follower case there is (in most cases) one agent that takes the dominant position, called the *leader*, and the other agents are called the *followers*. The leaders are often taken as autonomous systems [35, 75, 122], or systems with unknown inputs [29, 56, 61]. Problems of consensus or synchronization for leader-follower systems are often referred to as *distributed tracking control problems*. The goal of a distributed tracking control problem is then to design distributed protocols for the followers such that their dynamics tracks that of the leader [29, 78].

For multi-agent systems, the models of the agents are not necessarily required to be identical. Depending on whether the agents have the same dynamics, multiagent systems can also be classified into the following two subclasses, called *homogeneous* multi-agent systems and *heterogeneous* multi-agent systems. For homogeneous systems, the system models of the agents are identical, and here most existing work in the literature deals with *state* consensus or synchronization [54, 85, 103]. However, for heterogeneous systems, due to their very nature that the system models of the agents are allowed to be distinct, in particular their state space dimensions may even be different. For these systems it is therefore more natural and interesting to consider the collective *output* behavior [22, 46, 59, 99, 115].

While designing distributed control laws for a multi-agent system, one may not only want the controlled multi-agent network to achieve consensus or synchronization, but also would like the overall network to minimize a certain *optimality* criterion [7, 10, 23, 31, 41, 122]. Such problems are referred to as *distributed optimal control problems*. Within this framework, one of the important problems is the *distributed linear quadratic optimal control problem*. In the context of distributed linear quadratic optimal control, a global linear quadratic cost functional is introduced for a multi-agent system with *given initial states*. The objective is to design distributed control laws such that the given linear quadratic cost functional is minimized while the agents reach consensus or synchronization. Due to the particular form of distributed control laws, which capture the structure of the communication between the agents, the distributed linear quadratic optimal control problem is non-convex and very difficult to solve. It is also unclear whether in general a closed form solution exists.

As a consequence, the existing work in the literature on distributed linear quadratic optimal control either deals with suboptimality versions of this problem [7, 74, 76, 97, 98], or considers *special cases*, such as single integrator agent dynamics [10] and inverse optimality [75, 77, 123]. In Chapter 2 of this thesis, we investigate a suboptimality version of this problem. Given a leaderless multi-agent system and an associated global linear quadratic cost functional, we establish a design method for computing distributed control laws that guarantee the associated cost to be smaller than a given upper bound and achieve synchronization for the controlled network. In Chapter 3, we extend the results in Chapter 2 on distributed linear quadratic control for leaderless multi-agent systems to the case of distributed linear quadratic tracking control for leader-follower multi-agent systems. Both in the above distributed linear quadratic control problem and the distributed linear quadratic tracking problem, our computation of the proposed distributed control laws uses so-called global information, in the sense that, in order to compute the distributed control laws, knowledge of the entire network graph is required. To remove the dependence on this global information, in Chapter 4, for leaderless multi-agent systems with single integrator agent dynamics, we provide a decentralized method for computing distributed suboptimal control laws that do not involve global information.

Another important problem within the framework of distributed optimal control is the *distributed* \mathcal{H}_2 optimal control problem. In the context of distributed \mathcal{H}_2 optimal control, the dynamical model of a multi-agent system contains external disturbance inputs. An \mathcal{H}_2 cost functional is then introduced to quantify the influence of the disturbance inputs on the performance output of the overall network. The distributed \mathcal{H}_2 optimal control problem is the problem of finding a distributed protocol that minimizes the associated \mathcal{H}_2 cost while the controlled network achieves consensus or synchronization, see e.g. [36, 39, 53, 55, 112]. As before, due to the fact that the proposed distributed protocols have a particular structure imposed by the network graph, the problem of distributed \mathcal{H}_2 optimal control is a non-convex optimization problem. Again, it is unclear whether in general a closed form solution exists. Therefore, in Chapter 5 of this thesis, instead of considering the *actual* distributed \mathcal{H}_2 optimal control problem, we study a version of this problem that requires only suboptimality. Given a homogeneous multiagent system and an associated \mathcal{H}_2 cost functional, we provide a design method for obtaining distributed protocols using *static relative state* information such that the associated \mathcal{H}_2 cost is smaller than an a priori given upper bound and the controlled network achieves state synchronization. In Chapter 6, we generalize the results in Chapter 5 on static relative state feedback to the general case of *dynamic relative output feedback.* The results in Chapters 5 and 6 on distributed \mathcal{H}_2 suboptimal control of homogeneous multi-agent systems are then further generalized in Chapter 7 to the case of *heterogeneous* multi-agent systems.

In parallel to the development of control design for consensus and synchronization of multi-agent systems, recent years have also witnessed an increasing interest in problems of distributed state estimation for spatially constrained largescale systems. Applications can be found in power grids [32], industrial plants [107] and wireless sensor networks [86]. Due to physical constraints on the monitored systems, the measured output of a system is often not available to one single sensor. Consequently, standard estimation methods do not directly apply anymore. It might however be possible to monitor the state of a system by means of a sensor network. Such a sensor network consists a number of local sensors, where each of these has access to a certain portion of the measured output of the system. Each sensor then makes use of its obtained output portion to generate an estimate, and communicates this local estimate to the other local sensors according to a given communication graph. In this way, the states of all local sensors will reach synchronization to a common trajectory, which is then an estimate of the state of the measured system. Problems of monitoring the state of a spatially constrained system by a sensor network are often referred to as distributed state estimation problems.

The distributed state estimation problem has been studied mainly in two

research directions, namely, distributed observer design [26, 27, 49, 65, 88, 113, 114] and distributed filtering [40, 47, 81, 83, 84, 105, 106]. In the distributed observer design problem, the system is noise/disturbance free and is monitored by a number of local sensors, called *local observers*. Each local observer makes use of its measured output portion of the monitored system and then communicates with the other local observers according to the given communication graph. In this way, the local observers together form a *distributed observer*. The aim is to design a distributed observer such that all local observers reconstruct the state of the system. On the other hand, in the context of distributed filtering, the dynamic model of the monitored system contains noise/disturbance inputs and its output is observed by a number of local sensors, which are referred to as *local filters*. Similarly, each local filter makes use of its measured output portion and then exchanges information with the other local filters according to the given communication graph. In this way, these local filters together form a *distributed filter*. The goal of the distributed filtering problem is to design a distributed filter such that the states of all local filters track that of the system and, in addition, this distributed filter is optimal with respect to a certain cost functional. A typical problem appearing in this research direction is the *distributed Kalman filtering problem*, see e.g. [81, 83, 84].

In the literature on distributed filtering, most of the existing work deals with stochastic versions of this problem. In Chapter 8 of this thesis, however, we consider two *deterministic* versions of the distributed optimal filtering problem for linear systems, more specifically, the distributed \mathcal{H}_2 and \mathcal{H}_∞ optimal filtering problems. The distributed \mathcal{H}_2 and \mathcal{H}_∞ optimal filtering problems are the problems of designing local filter gains such that the \mathcal{H}_2 or \mathcal{H}_∞ norm of the transfer matrix from the disturbance input to the output estimation error is minimized while all local filters reconstruct the full system state asymptotically. Again, due to their *non-convex* nature, these problems are in general very challenging and it is not clear whether solutions exist. Therefore, instead in Chapter 8 we address suboptimality versions of these problems. In particular, we provide conceptual algorithms for obtaining \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filters, respectively. The resulting distributed filters guarantee that the \mathcal{H}_2 or \mathcal{H}_∞ norm of the transfer matrix from the disturbance input to the output estimation error is smaller than an a priori given upper bound, while all local filters reconstruct the full system state asymptotically.

1.2 Outline of this thesis

The organization of this thesis is as follows. Chapters 2 - 4 are concerned with the distributed linear quadratic optimal control problem. In Chapter 2, we study a *suboptimality* version of the distributed linear quadratic optimal control problem for

leaderless homogeneous multi-agent systems. In Chapter 3, we extend the results in Chapter 2 on distributed linear quadratic suboptimal control for leaderless multi-agent systems to the case of distributed linear quadratic suboptimal tracking control for *leader-follower* multi-agent systems. The computation of the local control gains in Chapters 2 and 3 requires complete knowledge of the eigenvalues of the Laplacian matrix or of a given positive definite matrix associated with the communication graph interconnecting the agents, often called *global information*. In Chapter 4, we aim at removing this dependence on global information. For multiagent systems with single integrator agent dynamics, we establish a *decentralized* computation method for computing suboptimal local control gains. Chapters 5 - 7 deal with the distributed \mathcal{H}_2 suboptimal control problem. In Chapter 5, we study this problem for *homogeneous* multi-agent systems by *static relative state feedback*, and the results are then generalized in Chapter 6 to the case of *dynamic relative* output feedback. In Chapter 7, we further generalize the results in Chapters 5 and 6, and investigate the distributed \mathcal{H}_2 suboptimal control problem for *heterogeneous* multi-agent systems. In Chapter 8, we study \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filtering problems for linear systems. In Chapter 9, we formulate the conclusions of this thesis, and discuss directions for possible future research.

1.3 Publications during the PhD project

Journal papers

- J. Jiao, H. L. Trentelman and M. K. Camlibel, "A suboptimality approach to distributed linear quadratic optimal control", *IEEE Transactions on Automatic Control*, Volume: 65, Issue: 3, 2020. (Chapter 2)
- J. Jiao, H. L. Trentelman and M. K. Camlibel, "Distributed linear quadratic optimal control: compute locally and act globally", *IEEE Control Systems Letters*, Volume: 4, Issue: 1, 2020. (Chapter 4)
- J. Jiao, H. L. Trentelman and M. K. Camlibel, "A suboptimality approach to distributed *H*₂ control by dynamic output feedback", Automatica, Volume 121, 109164, 2020. (Chapter 6)
- J. Jiao, H. L. Trentelman and M. K. Camlibel, "*H*₂ suboptimal output synchronization of heterogeneous multi-agent systems", submitted for publication in *Systems and Control Letters*, 2020. (Chapter 7)
- J. Jiao, H. L. Trentelman and M. K. Camlibel, "*H*₂ and *H*_∞ suboptimal distributed filter design for linear systems", submitted for publication in *IEEE Transactions on Automatic Control*, 2020. (Chapter 8)

Conference papers

- J. Jiao, H. L. Trentelman and M. K. Camlibel, "Distributed linear quadratic tracking control for leader-follower multi-agent systems: a suboptimality approach", 21st IFAC World Congress, 2020. (Chapter 3)
- J. Jiao, H. L. Trentelman and M. K. Camlibel, "A suboptimality approach to distributed H₂ optimal control", 7th IFAC Workshop on Distributed Estimation and Control in Networked Systems (NecSys18), 2018. (Chapter 5)

1.4 Notation

In this section, we will introduce some basic notation that will be used throughout this thesis.

We denote by \mathbb{R} the field of real numbers and by \mathbb{R}^n the space of *n* dimensional vectors over \mathbb{R} . For $x \in \mathbb{R}^n$, its Euclidean norm is defined by $||x|| := \sqrt{x^{\top}x}$. For a given r > 0, we denote by $B(r) := \{x \in \mathbb{R}^n \mid ||x|| \leq r\}$ the closed ball of radius r. We write $\mathbf{1}_N$ for the n dimensional column vector with all its entries equal to 1. We denote by $\mathbb{R}^{n \times m}$ the space of real $n \times m$ matrices. For a given matrix A, we write A^{\top} to denote its transpose and A^{-1} its inverse (if exists). For a symmetric matrix P, we denote P > 0 ($P \ge 0$) if it is positive (semi-)definite and P < 0 if its negative definite. We denote the identity matrix of dimension $n \times n$ by I_n . A matrix is called Hurwitz if all its eigenvalues have negative real parts. The trace of a square matrix A is denoted by tr(A). We denote by $diag(d_1, d_2, \ldots, d_n)$ the $n \times n$ diagonal matrix with d_1, d_2, \ldots, d_n on the diagonal. Given matrices $R_i \in \mathbb{R}^{m \times m}$, i = 1, 2, ..., n, we denote by blockdiag (R_i) the $nm \times nm$ block diagonal matrix with R_1, R_2, \ldots, R_n on the diagonal and we denote by $col(R_i)$ the $nm \times m$ column block matrix $(R_1^{\top}, R_2^{\top}, \dots, R_n^{\top})^{\top}$. The Kronecker product of two matrices A and *B* is denoted by $A \otimes B$. For a linear map $A : \mathcal{X} \to \mathcal{Y}$, the kernel and image of A are denoted by ker(A) := { $x \in \mathcal{X} \mid Ax = 0$ } and im(A) := { $Ax \mid x \in \mathcal{X}$ }, respectively.

1.5 Graph theory

We will now review some basic concepts and elementary results on graph theory that will be used.

A directed weighted graph is denoted by $\mathcal{G} = (\mathcal{V}, \mathcal{E}, \mathcal{A})$ with node set $\mathcal{V} = \{1, 2, ..., N\}$ and edge set $\mathcal{E} = \{e_1, e_2, ..., e_M\}$ satisfying $\mathcal{E} \subset \mathcal{V} \times \mathcal{V}$, and where $\mathcal{A} = [a_{ij}]$ is the adjacency matrix with nonnegative elements a_{ij} , called the edge weights. If $(i, j) \in \mathcal{E}$ we have $a_{ji} > 0$. If $(i, j) \notin \mathcal{E}$ we have $a_{ji} = 0$. Given a graph \mathcal{G} , a directed path from node 1 to node p is a sequence of edges (k, k + 1),

k = 1, 2, ..., p - 1. A directed weighted graph is called *strongly connected* if for any pair of distinct nodes *i* and *j*, there exists a directed path from *i* to *j*. A graph is called undirected if $(i, j) \in \mathcal{E}$ implies $(j, i) \in \mathcal{E}$. It is called *simple* if $a_{ii} = 0$ for all *i*. A simple undirected graph is called *connected* if for each pair of nodes *i* and *j* there exists a directed path from *i* to *j*.

Given a graph \mathcal{G} , the degree matrix of \mathcal{G} is the diagonal matrix, given by $\mathcal{D} = \text{diag}(\mathsf{d}_1, \mathsf{d}_2, \dots, \mathsf{d}_N)$ with $\mathsf{d}_i = \sum_{j=1}^N a_{ij}$. The Laplacian matrix is defined as $L := \mathcal{D} - \mathcal{A}$. If \mathcal{G} is a directed weighted graph, the associated Laplacian matrix L has a zero eigenvalue corresponding to the eigenvector $\mathbf{1}_N$. If moreover \mathcal{G} is strongly connected, then the eigenvalue 0 has multiplicity one, and all the other eigenvalues lie in the open right half-plane. The Laplacian matrix of an undirected graph is symmetric and has only real nonnegative eigenvalues. A simple undirected weighted graph is connected if and only if its Laplacian matrix L has eigenvalue 0 with multiplicity one. In that case, there exists an orthogonal matrix U such that

$$U^{\top}LU = \Lambda = \operatorname{diag}(0, \lambda_2, \dots, \lambda_N) \tag{1.1}$$

with $0 < \lambda_2 \leq \cdots \leq \lambda_N$. Moreover, we can take $U = \begin{pmatrix} \frac{1}{\sqrt{N}} \mathbf{1}_N & U_2 \end{pmatrix}$ with $U_2 U_2^\top = I_N - \frac{1}{N} \mathbf{1}_N \mathbf{1}_N^\top$.

A simple undirected weighted graph obviously has an even number of edges M. Define $K := \frac{1}{2}M$. For such graph, an associated incidence matrix $R \in \mathbb{R}^{N \times K}$ is defined as a matrix $R = (r_1, r_2, \ldots, r_K)$ with columns $r_k \in \mathbb{R}^N$. Each column r_k corresponds to exactly one pair of edges $e_k = \{(i, j), (j, i)\}$, and the *i*th and *j*th entry of r_k are equal to 1 or -1, while they do not take the same value. The remaining entries of r_k are equal to 0. We also define the matrix

$$W = \operatorname{diag}(\mathbf{w}_1, \mathbf{w}_2, \dots, \mathbf{w}_K) \tag{1.2}$$

as the $K \times K$ diagonal matrix, where W_k is the weight on each of the edges in e_k for k = 1, 2, ..., K. The relation between the Laplacian matrix and the incidence matrix is captured by

$$L = RWR^{\top}.$$

For connected simple undirected graphs, we review the following result [29]:

Lemma 1.1. Let G be a connected simple undirected graph with Laplacian matrix L. Let g_1, g_2, \ldots, g_N be non-negative real numbers with at least one $g_i > 0$. Define $G = \text{diag}(g_1, g_2, \ldots, g_N)$. Then the matrix L + G is positive definite.

For strongly connected weighted directed graphs, we review the following result [8, 62]:

Lemma 1.2. Let G be a strongly connected weighted directed graph with Laplacian matrix L. Then there exists a unique row vector $\theta = (\theta_1, \theta_2, \dots, \theta_N)$, where $\theta_1, \theta_2, \dots, \theta_N$ are all

positive real numbers, such that $\theta L = 0$ and $\theta \mathbf{1}_N = N$. Define $\Theta := \operatorname{diag}(\theta_1, \theta_2, \dots, \theta_N)$, then the matrix $\mathcal{L} := \Theta L + L^\top \Theta$ is a positive semi-definite matrix.

Note that ΘL is the Laplacian matrix of the balanced directed graph obtained by adjusting the weights in the original graph. The matrix \mathcal{L} is the Laplacian matrix of the undirected graph obtained by taking the union of the edges and their reversed edges in this balanced directed graph.

2

A suboptimality approach to distributed linear quadratic optimal control

This chapter is concerned with a suboptimality version of the distributed linear quadratic optimal control problem for multi-agent systems. Given a multi-agent system with identical agent dynamics and an associated global quadratic cost functional, our objective is to design distributed control laws that achieve synchronization and whose cost is smaller than an a priori given upper bound, for all initial states of the network that are bounded in norm by a given radius. A centralized design method is provided to compute such suboptimal controllers, involving the solution of a single Riccati inequality of dimension equal to the dimension of the agent dynamics, and the smallest nonzero and the largest eigenvalue of the Laplacian matrix. Furthermore, we relax the requirement of exact knowledge of the smallest nonzero and largest eigenvalue of the Laplacian matrix by using only lower and upper bounds on these eigenvalues. Finally, a simulation example is provided to illustrate our design method.

2.1 Introduction

In this chapter, we study the distributed linear quadratic optimal control problem for multi-agent networks. This problem deals with a number of identical agents represented by a finite dimensional linear input-state system, and an undirected graph representing the communication between these agents. Given is also a quadratic cost functional that penalizes the differences between the states of neighboring agents and the size of the local control inputs. The distributed linear quadratic control problem is the problem of finding a distributed diffusive control law that minimizes this cost functional, while achieving synchronization for the controlled network. This problem is non-convex and difficult to solve, and a closed form solution has not been provided in the literature up to now. It is also unknown under what conditions a distributed diffusive optimal control law exists in general [74]. Therefore, instead of addressing the problem formulated above, in the present chapter we will study a *suboptimality* version of this optimal control problem. In other words, our aim will be to design distributed diffusive suboptimal control laws that guarantee the controlled network to reach synchronization.

The distributed linear quadratic control problem has attracted extensive attention in the last decade, and has been studied from many different angles. For example, in [70, 109, 119] it was shown that if the quadratic cost functional involves the differences of states of neighboring agents, then, necessarily, the optimal control laws must be distributed and diffusive. However, these references do not address the problem of designing the optimal control laws. In [7], a design method was introduced for computing distributed suboptimal *stabilizing* controllers for decoupled linear systems. In this reference, the authors consider a global linear quadratic cost functional which contains terms that penalize the states and inputs of each agent and also the relative states between each agent and its neighboring agents. In [104, 122], methods were established for designing distributed *synchronizing* control laws for linear multi-agent systems, where the control laws are derived from the solution of an algebraic Riccati equation of dimension equal to the state space dimension of the agents. However, in these references, cost functionals were not taken explicitly into consideration.

The distributed linear quadratic optimal control problem was also addressed in [10] for multi-agent systems with *single integrator* agent dynamics. The authors obtained an expression for the optimal control law, with the optimal feedback gain given in terms of the initial conditions of *all* agents. In addition, in [98] a distributed optimal control problem was considered from the perspective of cooperative game theory. In that paper, the problem being studied was solved by transforming it into a maximization problem for linear matrix inequalities, taking into consideration the structure of the Laplacian matrix. For related work we also mention [15, 19, 73, 117], to name a few.

Also, in [76], a hierarchical control approach was introduced for linear leaderfollower multi-agent systems. For the case that the weighting matrices in the cost functional are chosen to be of a special form, two suboptimal controller design methods were given in this reference. In addition, in [75], an inverse optimal control problem was addressed both for leader-follower and leaderless multi-agent systems. For a particular class of digraphs, the authors showed that distributed optimal controllers exist, and can be obtained if the weighting matrices are assumed to be of a special form, capturing the graph information. For other work related to distributed inverse optimal control, we refer to [77, 123].

In this chapter, our objective is to design distributed diffusive control laws that guarantee the controlled network to reach synchronization while the associated cost is smaller than an a priori given upper bound. The main contributions of this chapter are the following:

- We present a design method for computing distributed diffusive suboptimal control laws, based on computing a positive definite solution of a single Riccati inequality of dimension equal to the dimension of the agent dynamics. In the computation of the local control gain, the smallest nonzero eigenvalue and the largest eigenvalue of the Laplacian matrix are involved.
- For the case that exact information on the smallest nonzero eigenvalue and the largest eigenvalue of the Laplacian matrix is not available, we establish a design method using only lower and upper bounds on these Laplacian eigenvalues.

The remainder of this chapter is organized as follows. In Section 2.2, we formulate the distributed linear quadratic suboptimal control problem. Section 2.3 presents the analysis and design of linear quadratic suboptimal control for linear systems, collecting preliminary classical results for treating the actual distributed suboptimal control problem for multi-agent systems. Then, in Section 2.4, we study the distributed suboptimal control problem for linear multi-agent systems. To illustrate our results, a simulation example is provided in Section 2.5. Finally, in Section 2.6 we formulate some conclusions.

2.2 **Problem formulation**

In this chapter, we consider a multi-agent system consisting of N identical agents. It will be a standing assumption that the underlying graph is simple, undirected and connected. The corresponding Laplacian matrix is denoted by L. The dynamics of the identical agents is represented by the continuous-time linear timeinvariant (LTI) system given by

$$\dot{x}_i(t) = Ax_i(t) + Bu_i(t), \quad x_i(0) = x_{i0}, \quad i = 1, 2, \dots, N,$$
(2.1)

where $A \in \mathbb{R}^{n \times n}$, $B \in \mathbb{R}^{n \times m}$, and $x_i \in \mathbb{R}^n$, $u_i \in \mathbb{R}^m$ are the state and input of the agent *i*, respectively, and x_{i0} is its initial state. Throughout this chapter, we assume that the pair (A, B) is stabilizable.

We consider the infinite horizon distributed linear quadratic optimal control problem for multi-agent system (2.1), where the global cost functional integrates the weighted quadratic difference of states between every agent and its neighbors, and also penalizes the inputs in a quadratic form. Thus, the cost functional considered in this chapter is given by

$$J(u_1, u_2, \dots, u_N) = \int_0^\infty \frac{1}{2} \sum_{i=1}^N \sum_{j \in \mathcal{N}_i} (x_i - x_j)^\top Q(x_i - x_j) + \sum_{i=1}^N u_i^\top R u_i \, dt, \quad (2.2)$$

where $Q \ge 0$ and R > 0 are given real weighting matrices.

We can rewrite multi-agent system (2.1) in compact form as

$$\dot{\mathbf{x}} = (I_N \otimes A)\mathbf{x} + (I_N \otimes B)\mathbf{u}, \quad \mathbf{x}(0) = \mathbf{x}_0$$
(2.3)

with $\mathbf{x} = (x_1^{\top}, x_2^{\top}, \dots, x_N^{\top})^{\top}$, $\mathbf{u} = (u_1^{\top}, u_2^{\top}, \dots, u_N^{\top})^{\top}$, where $\mathbf{x} \in \mathbb{R}^{nN}$, $\mathbf{u} \in \mathbb{R}^{mN}$ contain the states and inputs of all agents, respectively. Note that, although the agents have identical dynamics, we allow the initial states of the individual agents to differ. These initial states are collected in the joint vector of initial states $\mathbf{x}_0 = (x_{10}^{\top}, x_{20}^{\top}, \dots, x_{N0}^{\top})^{\top}$. Moreover, we can also write the cost functional (2.2) in compact form as

$$J(\mathbf{u}) = \int_0^\infty \mathbf{x}^\top (L \otimes Q) \mathbf{x} + \mathbf{u}^\top (I_N \otimes R) \mathbf{u} \, dt.$$
(2.4)

The distributed linear quadratic optimal control problem is the problem of minimizing for all initial states \mathbf{x}_0 the cost functional (2.4) over all distributed diffusive control laws that achieve synchronization. By a distributed diffusive control law we mean a control law of the form

$$\mathbf{u} = (L \otimes K)\mathbf{x},\tag{2.5}$$

where $K \in \mathbb{R}^{m \times n}$ is an identical feedback gain for all agents. The adjective diffusive refers to the fact that the input of each agent depends on the relative state variables with respect to its neighbors. The control law (2.5) is distributed in the sense that the local gains for all agents are identical.

By interconnecting the agents using this control law, we obtain the overall network dynamics

$$\dot{\mathbf{x}} = (I_N \otimes A + L \otimes BK)\mathbf{x}.$$
(2.6)

Foremost, we want the control law to achieve synchronization:

Definition 2.1. We say the network reaches synchronization using control law (2.5) if for all i, j = 1, 2, ..., N and for all initial states x_{i0} and x_{j0} , we have

$$x_i(t) - x_j(t) \to 0 \text{ as } t \to \infty.$$

As a function of the to-be-designed local feedback gain K, the cost functional (2.4) can be rewritten as

$$J(K) = \int_0^\infty \mathbf{x}^\top \left(L \otimes Q + L^2 \otimes K^\top R K \right) \mathbf{x} \, dt.$$
(2.7)

In other words, the distributed linear quadratic optimal control problem is the

problem of minimizing the cost functional (2.7) over all $K \in \mathbb{R}^{m \times n}$ such that the controlled network (2.6) reaches synchronization.

Due to the distributed nature of the control law (2.5) as imposed by the network topology, the distributed linear quadratic optimal control problem is a *non-convex* optimization problem. It is therefore difficult, if not impossible, to find a *closed form solution* for an optimal controller, or such optimal controller may not even exist. Therefore, as announced in the introduction, in this chapter we will study and resolve a version of this problem involving the design of distributed suboptimal control laws.

More specifically, let $B(r) = {\mathbf{x} \in \mathbb{R}^{nN} | ||\mathbf{x}|| \leq r}$ be the closed ball of radius r in the joint state space \mathbb{R}^{nN} of the network (2.3). Then, for system (2.3) with initial states in such a closed ball of a given radius, we want to design a distributed diffusive controller such that synchronization is achieved and, for all initial states in the given ball, the associated cost is smaller than an a priori given upper bound. Thus, we will consider the following problem:

Problem 2.1. Consider the multi-agent system (2.3) and associated cost functional given by (2.7). Let r > 0 be a given radius and let $\gamma > 0$ be an a priori given upper bound for the cost. The problem is to find a distributed diffusive controller of the form (2.5) such that the controlled network (2.6) reaches synchronization, and for all $\mathbf{x}_0 \in B(r)$ the associated cost (2.7) is smaller than the given upper bound, i.e., $J(K) < \gamma$.

Remark 2.1. Note that we could also have formulated the alternative problem of finding a suboptimal controller for a *single*, *given*, *initial state* \mathbf{x}_0 . In fact, this would be closer to the classical linear quadratic optimal control problem, which is usually formulated as the problem of minimizing the cost functional for a *given* initial state \mathbf{x}_0 . In that context, however, the optimal controller is a state feedback that turns out to be optimal *for all initial states*. In order to capture in our problem formulation this property of being optimal for all initial states, we have formulated Problem 2.1 in terms of initial states contained in a ball of a given radius.

Before we address Problem 2.1, we will first briefly discuss the linear quadratic suboptimal control problem for a single linear system. This will be the subject of the next section.

2.3 Linear quadratic suboptimal control for linear systems

In this section, we consider a linear quadratic suboptimal control problem for single linear systems. The results presented in this section are standard and can be found scattered over the literature, see e.g. [33, 100, 102]. Exact references are

however hard to give and therefore, in order to make this chapter self-contained, we will collect the required results here and provide their proofs.

We will first analyze the quadratic performance of a given autonomous system. Subsequently, we will discuss how to design suboptimal control laws for a linear system with inputs.

2.3.1 Quadratic performance analysis for autonomous linear systems

Consider the autonomous linear system

$$\dot{x}(t) = \bar{A}x(t), \quad x(0) = x_0,$$
(2.8)

where $\overline{A} \in \mathbb{R}^{n \times n}$ and $x \in \mathbb{R}^n$ is the state. We consider the quadratic performance of system (2.8), given by

$$J = \int_0^\infty x^\top \bar{Q}x \, dt, \tag{2.9}$$

where $\bar{Q} \ge 0$ is a given real weighting matrix. Note that the performance *J* is finite if system (2.8) is stable, i.e., \bar{A} is Hurwitz.

We are interested in finding conditions such that the performance (2.9) of system (2.8) is smaller than a given upper bound. For this, we have the following lemma:

Lemma 2.2. Consider system (2.8) with the corresponding quadratic performance (2.9). The performance (2.9) is finite if system (2.8) is stable, i.e., \overline{A} is Hurwitz. In this case, it is given by

$$J = x_0^{\top} Y x_0, (2.10)$$

where Y is the unique positive semi-definite solution of

$$\bar{A}^{\top}Y + Y\bar{A} + \bar{Q} = 0.$$
 (2.11)

Alternatively,

$$J = \inf\{x_0^{\top} P x_0 \mid P > 0 \text{ and } \bar{A}^{\top} P + P \bar{A} + \bar{Q} < 0\}.$$
 (2.12)

Proof. The fact that the quadratic performance (2.9) is given by the quadratic expression (2.10) involving the Lyapunov equation (2.11) is well-known.

We will now prove (2.12). Let *Y* be the solution to Lyapunov equation (2.11) and let *P* be a positive definite solution to the Lyapunov inequality in (2.12). Define X := P - Y. Then we have

$$\bar{A}^{\top}(X+Y) + (X+Y)\bar{A} + \bar{Q} < 0.$$

So consequently,

$$\bar{A}^{\top}X + X\bar{A} < 0.$$

Since \overline{A} is Hurwitz, it follows that X > 0. Thus, we have P > Y and hence $J \leq x_0^\top P x_0$ for any positive definite solution P to the Lyapunov inequality.

Next we will show that for any $\epsilon > 0$ there exists a positive definite matrix P_{ϵ} satisfying the Lyapunov inequality such that $P_{\epsilon} < Y + \epsilon I$, and consequently $x_0^{\top} P_{\epsilon} x_0 \leq J + \epsilon ||x_0||^2$. Indeed, for given ϵ , take P_{ϵ} equal to the unique positive definite solution of

$$\bar{A}^{\top}P + P\bar{A} + \bar{Q} + \epsilon I = 0.$$

Clearly then, $P_{\epsilon} = \int_{0}^{\infty} e^{\bar{A}^{\top} t} (\bar{Q} + \epsilon I) e^{\bar{A}t} dt$, so $P_{\epsilon} \downarrow Y$ as $\epsilon \downarrow 0$. This proves our claim.

The following theorem now yields *necessary* and *sufficient* conditions such that, for a given upper bound $\gamma > 0$, the quadratic performance (2.9) satisfies $J < \gamma$.

Theorem 2.3. Consider system (2.8) with the associated quadratic performance (2.9). For given $\gamma > 0$, we have that \overline{A} is Hurwitz and $J < \gamma$ if and only if there exists a positive definite matrix P satisfying

$$\bar{A}^{\top}P + P\bar{A} + \bar{Q} < 0, \tag{2.13}$$

$$x_0^{\dagger} P x_0 < \gamma. \tag{2.14}$$

Proof. (if) Since there exists a positive definite solution to the Lyapunov inequality (2.13), it follows that \overline{A} is Hurwitz. Take a positive definite matrix P satisfying the inequalities (2.13) and (2.14). By Lemma 2.2, we then immediately have $J \leq x_0^{\top} P x_0 < \gamma$.

(only if) If \overline{A} is Hurwitz and $J < \gamma$, then, again by Lemma 2.2, there exists a positive definite solution P to the Lyapunov inequality (2.13) such that $J \leq x_0^{\top} P x_0 < \gamma$.

In the next subsection, we will discuss the suboptimal control problem for a linear system with inputs.

2.3.2 Linear quadratic suboptimal control for linear systems

In this section, we consider the linear time-invariant system given by

$$\dot{x}(t) = Ax(t) + Bu(t), \quad x(0) = x_0,$$
(2.15)

where $A \in \mathbb{R}^{n \times n}$, $B \in \mathbb{R}^{n \times m}$, and $x \in \mathbb{R}^n$, $u \in \mathbb{R}^m$ are the state and the input, respectively, and x_0 is a given initial state. Assume that the pair (A, B) is stabilizable.

The associated cost functional is given by

$$J(u) = \int_0^\infty x^\top Q x + u^\top R u \, dt, \qquad (2.16)$$

where $Q \ge 0$ and R > 0 are given weighting matrices that penalize the state and input, respectively.

Given $\gamma > 0$ and initial state x_0 , we want to find a state feedback control law u = Kx such that the closed system

$$\dot{x}(t) = (A + BK)x(t)$$
 (2.17)

is stable and the corresponding cost

$$J(K) = \int_0^\infty x^\top (Q + K^\top R K) x \, dt$$
 (2.18)

satisfies $J(K) < \gamma$.

The following theorem gives a sufficient condition for the existence of such control law.

Theorem 2.4. Consider the system (2.15) with initial state x_0 and associated cost functional (2.16). Let $\gamma > 0$. Suppose that there exists a positive definite P satisfying

$$A^{\top}P + PA - PBR^{-1}B^{\top}P + Q < 0, \tag{2.19}$$

$$x_0^\top P x_0 < \gamma. \tag{2.20}$$

Let $K := -R^{-1}B^{\top}P$. Then the controlled system (2.17) is stable and the control law u = Kx is suboptimal, i.e., $J(K) < \gamma$.

Proof. Substituting $K := -R^{-1}B^{\top}P$ into (2.17) yields

$$\dot{x}(t) = (A - BR^{-1}B^{\top}P)x(t), \quad x(0) = x_0.$$
 (2.21)

Since P satisfies (2.19), it should also satisfy

$$(A - BR^{-1}B^{\top}P)^{\top}P + P(A - BR^{-1}B^{\top}P) + Q + PBR^{-1}B^{\top}P < 0,$$

which implies that $A - BR^{-1}B^{\top}P$ is Hurwitz, i.e., the closed system (2.21) is stable. Consequently, the corresponding cost is finite and equal to

$$J(K) = \int_0^\infty x^\top (Q + PBR^{-1}B^\top P)x \, dt.$$

Since (2.20) holds, by taking $\overline{A} = A - BR^{-1}B^{\top}P$ and $\overline{Q} = Q + PBR^{-1}B^{\top}P$ in

Theorem 2.3, we immediately have $J(K) < \gamma$.

In the next section we will apply the above results to tackle the distributed linear quadratic suboptimal control problem as formulated in Problem 2.1.

2.4 Distributed linear quadratic suboptimal control for multi-agent systems

Again consider the multi-agent system with the dynamics of the identical agents represented by

$$\dot{x}_i(t) = Ax_i(t) + Bu_i(t), \quad x_i(0) = x_{i0}, \quad i = 1, 2, \dots, N,$$
(2.22)

where $A \in \mathbb{R}^{n \times n}$, $B \in \mathbb{R}^{n \times m}$, and $x_i \in \mathbb{R}^n$, $u_i \in \mathbb{R}^m$ are the state and input of the *i*-th agent, respectively, and x_{i0} its initial state. We assume that the pair (A, B) is stabilizable.

Denoting $\mathbf{x} = (x_1^{\top}, x_2^{\top}, \dots, x_N^{\top})^{\top}$, $\mathbf{u} = (u_1^{\top}, u_2^{\top}, \dots, u_N^{\top})^{\top}$, we can rewrite the multi-agent system in compact form as

$$\dot{\mathbf{x}} = (I_N \otimes A)\mathbf{x} + (I_N \otimes B)\mathbf{u}, \quad \mathbf{x}(0) = \mathbf{x}_0.$$
(2.23)

The cost functional we consider was already introduced in (2.4). We repeat it here for convenience:

$$J(\mathbf{u}) = \int_0^\infty \mathbf{x}^\top (L \otimes Q) \mathbf{x} + \mathbf{u}^\top (I_N \otimes R) \mathbf{u} \, dt, \qquad (2.24)$$

where $Q \ge 0$ and R > 0 are given real weighting matrices.

As formulated in Problem 2.1, given a desired upper bound $\gamma > 0$, for multiagent system (2.23) with initial states contained in the closed ball B(r) of given radius r we want to design a control law of the form

$$\mathbf{u} = (L \otimes K)\mathbf{x} \tag{2.25}$$

such that the controlled network

$$\dot{\mathbf{x}} = (I_N \otimes A + L \otimes BK)\mathbf{x} \tag{2.26}$$

reaches synchronization and, moreover, for all $\mathbf{x}_0 \in B(r)$ the associated cost

$$J(K) = \int_0^\infty \mathbf{x}^\top \left(L \otimes Q + L^2 \otimes K^\top R K \right) \mathbf{x} \, dt \tag{2.27}$$

is smaller than the given upper bound, i.e., $J(K) < \gamma$.

Let the matrix $U \in \mathbb{R}^{\bar{N} \times N}$ be an orthogonal matrix that diagonalizes the Laplacian *L*. Define $\Lambda := U^{\top}LU = \text{diag}(0, \lambda_2, \dots, \lambda_N)$. To simplify the problem given above, by applying the state and input transformations $\bar{\mathbf{x}} = (U^{\top} \otimes I_n)\mathbf{x}$ and $\bar{\mathbf{u}} = (U^{\top} \otimes I_m)\mathbf{u}$ with $\bar{\mathbf{x}} = (\bar{x}_1^{\top}, \bar{x}_2^{\top}, \dots, \bar{x}_N^{\top})^{\top}$, $\bar{\mathbf{u}} = (\bar{u}_1^{\top}, \bar{u}_2^{\top}, \dots, \bar{u}_N^{\top})^{\top}$, system (2.23) becomes

$$\dot{\mathbf{x}} = (I_N \otimes A)\bar{\mathbf{x}} + (I_N \otimes B)\bar{\mathbf{u}}, \quad \bar{\mathbf{x}}(0) = \bar{\mathbf{x}}_0$$
(2.28)

with $\bar{\mathbf{x}}_0 = (U^{\top} \otimes I_n) \mathbf{x}_0$. Clearly, (2.25) is transformed to

$$\bar{\mathbf{u}} = (\Lambda \otimes K)\bar{\mathbf{x}},\tag{2.29}$$

and the controlled network (2.26) transforms to

$$\dot{\bar{\mathbf{x}}} = (I_N \otimes A + \Lambda \otimes BK) \,\bar{\mathbf{x}}.\tag{2.30}$$

In terms of the transformed variables, the cost (2.27) is given by

$$J(K) = \int_0^\infty \sum_{i=1}^N \bar{x}_i^\top (\lambda_i Q + \lambda_i^2 K^\top R K) \bar{x}_i \, dt.$$
(2.31)

Note that the transformed states \bar{x}_i and inputs \bar{u}_i , i = 2, 3, ..., N appearing in system (2.30) and cost (2.31) are decoupled from each other, so that we can write system (2.30) and cost (2.31) as

$$\dot{\bar{x}}_1 = A\bar{x}_1,\tag{2.32}$$

$$\dot{\bar{x}}_i = (A + \lambda_i BK)\bar{x}_i, \quad i = 2, 3, \dots, N,$$
(2.33)

and

$$J(K) = \sum_{i=2}^{N} J_i(K)$$
 (2.34)

with

$$J_{i}(K) = \int_{0}^{\infty} \bar{x}_{i}^{\top} (\lambda_{i}Q + \lambda_{i}^{2}K^{\top}RK)\bar{x}_{i} dt, \quad i = 2, 3, \dots, N.$$
 (2.35)

Note that $\lambda_1 = 0$, and that therefore (2.32) does not contribute to the cost J(K).

We first record a well-known fact (see [54, 103]) that we will use later:

Lemma 2.5. Consider the multi-agent system (2.23). Then the controlled network reaches synchronization with control law (2.25) if and only if, for i = 2, 3, ..., N, the systems (2.33) are stable.

Thus we have transformed the problem of distributed suboptimal control for system (2.23) into the problem of finding a feedback gain $K \in \mathbb{R}^{m \times n}$ such that

the systems (2.33) are stable and $J(K) < \gamma$. Moreover, since the pair (A, B) is stabilizable, there exists such a feedback gain *K* [103].

The following lemma gives a necessary and sufficient condition for a given feedback gain *K* to make all systems (2.33) stable and such that $J(K) < \gamma$ is satisfied for given initial states.

Lemma 2.6. Let *K* be a feedback gain. Consider the systems (2.33) with given initial states $\bar{x}_{20}, \bar{x}_{30}, \ldots, \bar{x}_{N0}$ and associated cost functionals (2.34) and (2.35). Let $\gamma > 0$. Then all systems (2.33) are stable and $J(K) < \gamma$ if and only if there exist positive definite matrices P_i satisfying

$$(A + \lambda_i BK)^\top P_i + P_i (A + \lambda_i BK) + \lambda_i Q + \lambda_i^2 K^\top RK < 0,$$
(2.36)

$$\sum_{i=2}^{N} \bar{x}_{i0}^{\top} P_i \bar{x}_{i0} < \gamma, \qquad (2.37)$$

for $i = 2, 3, \ldots, N$, respectively.

Proof. (if) Since (2.37) holds, there exist sufficiently small $\epsilon_i > 0$, i = 2, 3, ..., N such that $\sum_{i=2}^{N} \gamma_i < \gamma$ where $\gamma_i := \bar{x}_{i0}^\top P_i \bar{x}_{i0} + \epsilon_i$. Because there exists P_i such that (2.36) and $\bar{x}_{i0}^\top P_i \bar{x}_{i0} < \gamma_i$ holds for all i = 2, 3, ..., N, by taking $\bar{A} = A + \lambda_i BK$ and $\bar{Q} = \lambda_i Q + \lambda_i^2 K^\top RK$, it follows from Theorem 2.3 that all systems (2.33) are stable and $J_i(K) < \gamma_i$ for i = 2, 3, ..., N. Since $J(K) = \sum_{i=2}^{N} J_i(K)$, this implies that $J(K) < \sum_{i=2}^{N} \gamma_i < \gamma$.

(only if) Since $J(K) < \gamma$ and $J(K) = \sum_{i=2}^{N} J_i(K)$, there exist sufficiently small $\epsilon_i > 0, i = 2, 3, ..., N$ such that $\sum_{i=2}^{N} \gamma_i < \gamma$ where $\gamma_i := J_i(K) + \epsilon_i$. Because all systems (2.33) are stable and $J_i(K) < \gamma_i$ for i = 2, 3, ..., N, by taking $\bar{A} = A + \lambda_i BK$ and $\bar{Q} = \lambda_i Q + \lambda_i^2 K^{\top} RK$, it follows from Theorem 2.3 that there exist positive definite P_i such that (2.36) and $\bar{x}_{i0}^{\top} P_i \bar{x}_{i0} < \gamma_i$ hold for all i = 2, 3, ..., N. Since $\sum_{i=2}^{N} \gamma_i < \gamma$, this implies that $\sum_{i=2}^{N} \bar{x}_{i0}^{\top} P_i \bar{x}_{i0} < \sum_{i=2}^{N} \gamma_i < \gamma$.

Lemma 2.6 establishes a necessary and sufficient condition for a given feedback gain *K* to stabilize all systems (2.33) and to satisfy, for given initial states of these systems, $J(K) < \gamma$. However, Lemma 2.6 does not yet provide a method to compute such *K*. In the following we present a method to find such *K*.

Lemma 2.7. Consider the multi-agent system (2.23) with associated cost functional (2.27). Let \mathbf{x}_0 be a given initial state for the multi-agent system. Let $\gamma > 0$. Let c be any real number such that $0 < c < \frac{2}{\lambda_N}$. We distinguish two cases:

(i) if $\frac{2}{2} < 1 \le \frac{2}{2}$ (2.28)

$$\frac{2}{\lambda_2 + \lambda_N} \leqslant c < \frac{2}{\lambda_N},\tag{2.38}$$

then there exists P > 0 satisfying the Riccati inequality

$$A^{\top}P + PA + (c^2\lambda_N^2 - 2c\lambda_N)PBR^{-1}B^{\top}P + \lambda_NQ < 0.$$
(2.39)

(ii) if

$$0 < c < \frac{2}{\lambda_2 + \lambda_N},\tag{2.40}$$

then there exists P > 0 satisfying

$$A^{\top}P + PA + (c^{2}\lambda_{2}^{2} - 2c\lambda_{2})PBR^{-1}B^{\top}P + \lambda_{N}Q < 0.$$
(2.41)

In both cases, if in addition P satisfies

$$\mathbf{x}_0^{\top} \left((I_N - \frac{1}{N} \mathbf{1}_N \mathbf{1}_N^{\top}) \otimes P \right) \mathbf{x}_0 < \gamma,$$
(2.42)

then the controlled network (2.26) with $K := -cR^{-1}B^{\top}P$ reaches synchronization and with the initial state \mathbf{x}_0 we have $J(K) < \gamma$.

Proof. We will only give the proof for case (i) above. Using the upper and lower bounds on *c* given by (2.38), it can be verified that $c^2 \lambda_i^2 - 2c\lambda_i \leq c^2 \lambda_N^2 - 2c\lambda_N < 0$ for i = 2, 3, ..., N. It is then easily seen that (2.39) has many positive definite solutions. Since also $\lambda_i \leq \lambda_N$, any such solution *P* is a solution to the N - 1 Riccati inequalities

$$A^{\top}P + PA + (c^{2}\lambda_{i}^{2} - 2c\lambda_{i})PBR^{-1}B^{\top}P + \lambda_{i}Q < 0, \quad i = 2, 3, \dots, N.$$
 (2.43)

Equivalently, P also satisfies the Lyapunov inequalities

$$(A - c\lambda_{i}BR^{-1}B^{\top}P)^{\top}P + P(A - c\lambda_{i}BR^{-1}B^{\top}P) +\lambda_{i}Q + c^{2}\lambda_{i}^{2}PBR^{-1}B^{\top}P < 0, \quad i = 2, 3, \dots, N.$$
(2.44)

Next, recall that $\bar{\mathbf{x}} = (U^{\top} \otimes I_n)\mathbf{x}$ with $U = \begin{pmatrix} \frac{1}{\sqrt{N}} \mathbf{1}_N & U_2 \end{pmatrix}$. From this it is easily seen that $(\bar{x}_{20}^{\top}, \bar{x}_{30}^{\top}, \cdots, \bar{x}_{N0}^{\top})^{\top} = (U_2^{\top} \otimes I_n)\mathbf{x}_0$. Also, $U_2U_2^{\top} = I_N - \frac{1}{N}\mathbf{1}_N\mathbf{1}_N^{\top}$. Since (2.42) holds, we have

$$\mathbf{x}_{0}^{\top} \left(U_{2} U_{2}^{\top} \otimes P \right) \mathbf{x}_{0} < \gamma \Leftrightarrow$$
$$\left(\left(U_{2}^{\top} \otimes I_{n} \right) \mathbf{x}_{0} \right)^{\top} \left(I_{N-1} \otimes P \right) \left(\left(U_{2}^{\top} \otimes I_{n} \right) \mathbf{x}_{0} \right) < \gamma \quad \Leftrightarrow$$
$$\left(\bar{x}_{20}^{\top}, \bar{x}_{30}^{\top}, \cdots, \bar{x}_{N0}^{\top} \right) \left(I_{N-1} \otimes P \right) \left(\bar{x}_{20}^{\top}, \bar{x}_{30}^{\top}, \cdots, \bar{x}_{N0}^{\top} \right)^{\top} < \gamma,$$

which is equivalent to

$$\sum_{i=2}^{N} \bar{x}_{i0}^{\top} P \bar{x}_{i0} < \gamma.$$
(2.45)

Taking $P_i = P$ for i = 2, 3, ..., N and $K := -cR^{-1}B^{\top}P$ in inequalities (2.36) and (2.37) immediately gives us inequalities (2.44) and (2.45). Then it follows from Lemma 2.6 that all systems (2.33) are stable and $J(K) < \gamma$. Furthermore, it follows from Lemma 2.5 that the controlled network (2.26) reaches synchronization.

We will now apply Lemma 2.7 to establish a solution to Problem 2.1. Indeed, the next main theorem gives a condition under which, for given radius r and upper bound γ , distributed diffusive suboptimal control laws exist, and explains how these can be computed.

Theorem 2.8. Consider the multi-agent system (2.23) with associated cost functional (2.27). Let r > 0 be a given radius and let $\gamma > 0$ be an a priori given upper bound for the cost. Let *c* be any real number such that $0 < c < \frac{2}{\lambda_N}$. We distinguish two cases:

(*i*) *if*

$$\frac{2}{\lambda_2 + \lambda_N} \leqslant c < \frac{2}{\lambda_N},\tag{2.46}$$

then there exists P > 0 satisfying the Riccati inequality

$$A^{\top}P + PA + (c^2\lambda_N^2 - 2c\lambda_N)PBR^{-1}B^{\top}P + \lambda_NQ < 0.$$
(2.47)

(ii) if

$$0 < c < \frac{2}{\lambda_2 + \lambda_N},\tag{2.48}$$

then there exists P > 0 satisfying

$$A^{\top}P + PA + (c^{2}\lambda_{2}^{2} - 2c\lambda_{2})PBR^{-1}B^{\top}P + \lambda_{N}Q < 0.$$
(2.49)

In both cases, if in addition P satisfies

$$P < \frac{\gamma}{r^2} I, \tag{2.50}$$

then the controlled network (2.26) with $K := -cR^{-1}B^{\top}P$ reaches synchronization and $J(K) < \gamma$ for all $\mathbf{x}_0 \in B(r)$.

Proof. Again, we only give the proof for case (i) above. Let P > 0 satisfy (2.47) and (2.50) holds. Our aim is to prove that (2.42) is satisfied for all $\mathbf{x}_0 \in B(r)$. First note that

$$\frac{1}{N} \mathbf{1}_N \mathbf{1}_N^\top \otimes P = \frac{1}{N} (\mathbf{1}_N \otimes P^{\frac{1}{2}}) (\mathbf{1}_N \otimes P^{\frac{1}{2}})^\top,$$
which is therefore positive semi-definite. Now, for all $\mathbf{x}_0 \in B(r)$ we have

$$\mathbf{x}_0^\top \left((I_N - \frac{1}{N} \mathbf{1}_N \mathbf{1}_N^\top) \otimes P \right) \mathbf{x}_0$$
$$\leq \mathbf{x}_0^\top (I_N \otimes P) \mathbf{x}_0 < \frac{\gamma}{r^2} \mathbf{x}_0^\top \mathbf{x}_0 \leq \gamma.$$

By Lemma 2.7 then, the controlled network (2.26) with the given *K* reaches synchronization and $J(K) < \gamma$ for all $\mathbf{x}_0 \in B(r)$.

Remark 2.9. Theorem 2.8 states that after choosing *c* satisfying the inequality (2.46) for case (i) and finding a positive definite *P* satisfying (2.47) and (2.50), the distributed control law with local gain $K = -cR^{-1}B^{\top}P$ is γ -suboptimal for all initial states of the network in the closed ball with radius *r*. By (2.50), the smaller the solution *P* of (2.47), the smaller the quotient $\frac{\gamma}{r^2}$ is allowed to be, leading to a smaller upper bound and a larger radius. The question then arises: how should we choose the parameter *c* in (2.46) so that the Riccati inequality (2.47) allows a positive definite solution that is as small as possible. In fact, one can find a positive definite solution *P*(*c*, ϵ) to (2.47) by solving the Riccati equation

$$A^{\top}P + PA - PB\bar{R}(c)^{-1}B^{\top}P + \bar{Q}(\epsilon) = 0$$
(2.51)

with $\bar{R}(c) = \frac{1}{-c^2 \lambda_N^2 + 2c\lambda_N} R$ and $\bar{Q}(\epsilon) = \lambda_N Q + \epsilon I_n$ where c is chosen as in (2.46) and $\epsilon > 0$. If c_1 and c_2 as in (2.46) satisfy $c_1 \leq c_2$, then we have $\bar{R}(c_1) \leq \bar{R}(c_2)$, so, clearly, $P(c_1, \epsilon) \leq P(c_2, \epsilon)$. Similarly, if $0 < \epsilon_1 \leq \epsilon_2$, we immediately have $\bar{Q}(\epsilon_1) \leq \bar{Q}(\epsilon_2)$. Again, it follows that $P(c, \epsilon_1) \leq P(c, \epsilon_2)$. Therefore, if we choose $\epsilon > 0$ very close to 0 and $c = \frac{2}{\lambda_2 + \lambda_N}$, we find the 'best' solution to the Riccati inequality (2.47) in the sense explained above.

Likewise, if *c* satisfies (2.48) corresponding to case (ii), it can be shown that if we choose $\epsilon > 0$ very close to 0 and c > 0 very close to $\frac{2}{\lambda_2 + \lambda_N}$, we find the 'best' solution to the Riccati inequality (2.49) in the sense explained above.

In Theorem 2.8, in order to compute a suitable feedback gain K, one needs to know λ_2 and λ_N , the smallest nonzero eigenvalue (the *algebraic connectivity*) and the largest eigenvalue of the Laplacian matrix, exactly. This requires so-called global information on the network graph which might not always be available. There exist algorithms to estimate λ_2 in a distributed way, yielding lower and upper bounds, see e.g. [2]. Moreover, also an upper bound for λ_N can be obtained in terms of the maximal node degree of the graph, see [1]. Then the question arises: can we still find a suboptimal controller reaching synchronization, using as information only a *lower bound* for λ_2 and an *upper bound* for λ_N ? The answer to this question is affirmative, as shown in the following theorem.

Theorem 2.10. Let a lower bound for λ_2 be given by $l_2 > 0$ and an upper bound for λ_N be given by L_N . Let r > 0 be a given radius and let $\gamma > 0$ be an a priori given upper bound for the cost. Choose *c* such that

$$\frac{2}{l_2 + L_N} \leqslant c < \frac{2}{L_N}.\tag{2.52}$$

Then there exists P > 0 such that

$$A^{\top}P + PA + (c^{2}L_{N}^{2} - 2cL_{N})PBR^{-1}B^{\top}P + L_{N}Q < 0.$$
(2.53)

If, in addition, P satisfies

$$P < \frac{\gamma}{r^2}I,\tag{2.54}$$

then the controlled network with local gain $K = -cR^{-1}B^{\top}P$ reaches synchronization and $J(K) < \gamma$ for all initial states $\mathbf{x}_0 \in B(r)$.

Furthermore, if we choose c such that

$$0 < c < \frac{2}{l_2 + L_N},\tag{2.55}$$

then there exists P > 0 such that

$$A^{\top}P + PA + (c^2 l_2^2 - 2c l_2) PBR^{-1}B^{\top}P + L_N Q < 0.$$
(2.56)

If, in addition, P satisfies (2.54), then the controlled network with $K := -cR^{-1}B^{\top}P$ reaches synchronization and $J(K) < \gamma$ for all $\mathbf{x}_0 \in B(r)$.

Proof. A proof can be given along the lines of the proof of Theorem 2.8. \Box

Remark 2.11. Note that also in Theorem 2.10 the question arises how to choose c > 0 such that the Riccati inequalities (2.53) and (2.56) admit a positive definite solution that is as small as possible. Following the same ideas as in Remark 2.9, if we choose $\epsilon > 0$ very close to 0 and c > 0 equal to $\frac{2}{l_2+L_N}$ in (2.53) (respectively very close to $\frac{2}{l_2+L_N}$ in (2.56)), we find the 'best' solution to the Riccati inequalities (2.53) and (2.56).

Moreover, one may also ask the question: can we compare, with the same choice for *c*, solutions to (2.53) with solutions to (2.47), and also solutions to (2.56) with solutions to (2.49)? The answer is affirmative. Choose *c* that satisfies both conditions (2.46) and (2.52). One can then check that the computed positive definite solution to (2.53) is indeed 'larger' than that to (2.47) as explained in Remark 2.9. A similar remark holds for the positive definite solutions to (2.56) and corresponding solutions to (2.49) if *c* satisfies both (2.48) and (2.55). We conclude that if, instead of using the exact values λ_2 and λ_N , we use a lower bound, respectively upper bound

for these eigenvalues, then the computed distributed control law is suboptimal for 'less' initial states of the agents.

Remark 2.12. As a final remark, we note that the theory developed in this chapter carries over unchanged to the case of undirected *weighted* graphs. In that case the expression for cost functional (2.2) should be changed to

$$J(u_1, u_2, \dots, u_N) = \int_0^\infty \frac{1}{2} \sum_{i=1}^N \sum_{j=1}^N a_{ij} (x_i - x_j)^\top Q(x_i - x_j) + \sum_{i=1}^N u_i^\top R u_i \, dt,$$

in which $\mathcal{A} = [a_{ij}]$ is the *weighted* adjacency matrix. Denoting the corresponding weighted Laplacian matrix by L, also this cost functional can be represented in compact form by (2.4), and the subsequent development will remain the same.

2.5 Simulation example

In this section, we will use a simulation example borrowed from [76] to illustrate the proposed design method for distributed suboptimal controllers. Consider a group of 8 linear oscillators with identical dynamics

$$\dot{x}_i = Ax_i + Bu_i, \quad x_i(0) = x_{i0}, \quad i = 1, 2, \dots, 8$$
 (2.57)

with

$$A = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix}, \quad B = \begin{pmatrix} 0 \\ 1 \end{pmatrix}$$

Assume the underlying graph is the undirected line graph with Laplacian matrix

$$L = \begin{pmatrix} 1 & -1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\ -1 & 2 & -1 & 0 & 0 & 0 & 0 & 0 \\ 0 & -1 & 2 & -1 & 0 & 0 & 0 & 0 \\ 0 & 0 & -1 & 2 & -1 & 0 & 0 & 0 \\ 0 & 0 & 0 & -1 & 2 & -1 & 0 & 0 \\ 0 & 0 & 0 & 0 & -1 & 2 & -1 & 0 \\ 0 & 0 & 0 & 0 & 0 & -1 & 2 & -1 \\ 0 & 0 & 0 & 0 & 0 & 0 & -1 & 1 \end{pmatrix}.$$

We consider the cost functional

$$J(\mathbf{u}) = \int_0^\infty \mathbf{x}^\top (L \otimes Q) \mathbf{x} + \mathbf{u}^\top (I_8 \otimes R) \mathbf{u} \, dt$$
 (2.58)



Figure 2.1: Plots of the state vector $\mathbf{x}^1 = (x_{1,1}, x_{2,1}, \dots, x_{8,1})$ (upper plot) and $\mathbf{x}^2 = (x_{1,2}, x_{2,2}, \dots, x_{8,2})$ (lower plot) of the 8 decoupled oscillators without control

where the matrices Q and R are chosen to be

$$Q = \begin{pmatrix} 2 & 0\\ 0 & 1 \end{pmatrix}, \quad R = 1.$$

Let the desired upper bound for the cost functional (2.58) be given as $\gamma = 3$. Our goal is to design a control law $\mathbf{u} = (L \otimes K)\mathbf{x}$ such that the controlled network reaches synchronization and the associated cost is less than γ for all initial states \mathbf{x}_0 in a closed ball B(r) with radius r. The radius r will be specified later on in this example.

In this example, we adopt the control design method given in case (i) of Theorem 2.8. The smallest nonzero and largest eigenvalue of *L* are $\lambda_2 = 0.0979$ and $\lambda_8 = 3.8478$. First, we compute a positive definite solution *P* to (2.47) by solving the Riccati equation

$$A^{\top}P + PA + (c^2\lambda_8^2 - 2c\lambda_8)PBR^{-1}B^{\top}P + \lambda_8Q + \epsilon I_2 = 0$$

with $\epsilon > 0$ chosen small as mentioned in Remark 2.9. Here we choose $\epsilon = 0.001$. Moreover, we choose $c = \frac{2}{\lambda_2 + \lambda_8} = 0.5$, which is the 'best' choice as mentioned in



Figure 2.2: Plots of the state vector $\mathbf{x}^1 = (x_{1,1}, x_{2,1}, \dots, x_{8,1})$ (upper plot) and $\mathbf{x}^2 = (x_{1,2}, x_{2,2}, \dots, x_{8,2})$ (lower plot) of the controlled oscillator network

Remark 2.9. Then, by solving (2.5) in Matlab, we obtain

$$P = \begin{pmatrix} 12.1168 & 3.1303\\ 3.1303 & 8.3081 \end{pmatrix}$$

Correspondingly, the local feedback gain is then equal to

$$K = \begin{pmatrix} -1.5652 & -4.1541 \end{pmatrix}$$

We now compute the radius r of a ball B(r) of initial states for which the distributed control law $\mathbf{u} = (L \otimes K)\mathbf{x}$ is suboptimal, i.e. J(K) < 3. We compute that the largest eigenvalue of P is equal to 13.8765. Hence for every radius r such that $\frac{3}{r^2} > 13.8765$ the inequality (2.54) holds. Thus, the distributed controller with local gain K is suboptimal for all \mathbf{x}_0 with $\|\mathbf{x}_0\| \leq r$ with r < 0.4650.

As an example, the following initial states of the agents satisfy this norm bound: $x_{10}^{\top} = (-0.08 \ 0.11)$, $x_{20}^{\top} = (0.12 \ -0.08)$, $x_{30}^{\top} = (0.09 \ -0.14)$, $x_{40}^{\top} = (-0.12 \ 0.04)$, $x_{50}^{\top} = (0.07 \ -0.16)$, $x_{60}^{\top} = (-0.11 \ 0.12)$, $x_{70}^{\top} = (0.15 \ -0.16)$, $x_{80}^{\top} = (-0.05 \ -0.14)$. The plots of the eight decoupled oscillators without control are shown in Figure 2.1. Figure 2.2 shows that the controlled network of oscillators reaches synchronization.

2.6 Conclusions

In this chapter, we have studied a distributed linear quadratic suboptimal control problem for undirected linear multi-agent networks. We have considered a multi-agent system with identical linear agent dynamics and an associated global quadratic cost functional. For these, we have provided a design method to compute distributed diffusive control laws whose cost is bounded by a given upper bound for all initial states in a closed ball of a given radius, and such that the controlled network reaches synchronization. The computation of the local control gain involves finding solutions of a single Riccati inequality, whose dimension is equal to the dimension of the agent dynamics, and also involves the smallest nonzero and largest eigenvalue of the Laplacian matrix. As an extension, we have removed the requirement of having exact knowledge on the smallest nonzero and largest eigenvalue of the Laplacian matrix by, instead, using only lower and upper bounds for these eigenvalues.

3 Distributed linear quadratic tracking control: a suboptimality approach

In this chapter, we extend the results in Chapter 2 on distributed linear quadratic control for *leaderless* multi-agent systems to the case of distributed linear quadratic tracking control for *leader-follower* multi-agent systems. Given one autonomous leader and a number of homogeneous followers, we introduce an associated global quadratic cost functional. We assume that the leader shares its state information with at least one of the followers and the communication between the followers is represented by a connected simple undirected graph. Our objective is to design distributed control laws such that the controlled network reaches tracking consensus and, moreover, the associated cost is smaller than a given tolerance for all initial states bounded in norm by a given radius. We establish a centralized design method for computing such suboptimal control laws, involving the solution of a single Riccati inequality of dimension equal to the dimension of the local agent dynamics, and the smallest and the largest eigenvalue of a given positive definite matrix involving the underlying graph. The proposed design method is illustrated by a simulation example.

3.1 Introduction

Distributed control for multi-agent systems has drawn much attention in the past two decades due to its practical applications, e.g., formation control, intelligent transportation systems and power grids. In the literature, basically two types of multi-agent systems are considered, namely leaderless multi-agent systems and leader-follower multi-agent systems. In the leaderless case, the local agents reach agreement which depends on the dynamics of all agents [85, 103]. In the leaderfollower case, the states or the outputs of the followers track that of the leader [29, 78]. One of the attractive directions in distributed control for multi-agent systems is to design distributed control laws that minimize certain global or local performances, while reaching an agreement for the controlled network.

In the past, quite some work has been devoted to distributed linear quadratic

(LQ) optimal control for leaderless multi-agent systems. In [104], an LQR based method was used to design distributed synchronizing control laws for a multi-agent system, without taking any performance into consideration. In [7], suboptimal distributed stabilizing control laws were established for a multi-agent system with general agent dynamics with respect to an associated global cost functional, while in [10], the optimal synchronizing control gain was computed for leaderless multi-agent systems with single integrator agent dynamics. In the meantime, the distributed LQ control problem was also considered in [98] by utilizing a game theoretic approach, in [75] by adopting an inverse optimal approach, and later in [37] by employing a suboptimality approach. For other papers related to this topic, see also [38].

On the other hand, distributed LQ tracking control for leader-follower multiagent systems has also attracted much attention. In [122], distributed synchronizing control laws were established using an LQR based approach without optimizing any performance. Later on, in [13], distributed suboptimal control laws were proposed for achieving guaranteed cost. In [76], a hierarchical LQR based method was presented to design suboptimal synchronizing control laws for leader-follower systems, and an inverse optimal approach was introduced in [75], see also [123].

In the present chapter we extend the results from [37] on distributed LQ control for leaderless multi-agent systems to the case of distributed LQ tracking control for leader-follower multi-agent systems. Given a leader-follower system with one autonomous leader and a number of followers, we introduce an associated global quadratic cost functional. We assume that the leader shares its state information with at least one of the followers, and the communication between the followers is represented by a connected simple undirected graph. Our aim is then to design distributed diffusive control laws such that the controlled network reaches tracking synchronization, i.e., the states of the followers track the state of the leader asymptotically and the associated cost is smaller than an a priori given upper bound.

The outline of this chapter is as follows. Section 3.2 provides some preliminaries on quadratic performance analysis for autonomous linear systems. In Section 3.3, we formulate the distributed linear quadratic suboptimal tracking control problem for leader-follower multi-agent systems. We then address this distributed suboptimal tracking control problem in Section 3.4. A simulation example is presented in Section 3.5 to illustrate our design method. Finally, Section 3.6 concludes this chapter.

3.2 Quadratic performance analysis for autonomous linear systems

In this subsection, we will analyze the quadratic performance of a linear autonomous system. Consider the autonomous system

$$\dot{x}(t) = \bar{A}x(t), \quad x(0) = x_0$$
(3.1)

where $\overline{A} \in \mathbb{R}^{n \times n}$ and $x \in \mathbb{R}^n$ is the state. We consider the quadratic performance of system (3.1), given by

$$J = \int_0^\infty x^\top(t) \bar{Q}x(t) dt$$
(3.2)

where $\bar{Q} \ge 0$ is a given real weighting matrix. Note that the performance *J* is finite if system (3.1) is asymptotically stable, i.e., \bar{A} is Hurwitz.

The following well-known result ([37, 100]) provides a *necessary* and *sufficient* condition such that, for a given tolerance $\gamma > 0$, the performance (3.2) satisfies $J < \gamma$.

Theorem 3.1. Consider system (3.1) with associated performance (3.2). For given $\gamma > 0$, we have that \overline{A} is Hurwitz and $J < \gamma$ if and only if there exists P > 0 satisfying

$$\bar{A}^{\top}P + P\bar{A} + \bar{Q} < 0, \tag{3.3}$$

$$x_0^{\dagger} P x_0 < \gamma. \tag{3.4}$$

In the next section, we will formulate the problem that we will address in this chapter.

3.3 **Problem formulation**

In this chapter, we consider a leader-follower multi-agent system, consisting of one leader and N followers. The dynamics of the leader is represented by the linear time-invariant autonomous system

$$\dot{x}_r(t) = Ax_r(t), \quad x_r(0) = x_{r0}.$$
(3.5)

where $A \in \mathbb{R}^{n \times n}$, $x_r \in \mathbb{R}^n$ is the state of the leader and x_{r0} is its initial state. The dynamics of the followers are identical and represented by the linear time-invariant systems

$$\dot{x}_i(t) = Ax_i(t) + Bu_i(t), \quad x_i(0) = x_{i0}, \quad i = 1, 2, \dots, N$$
(3.6)

where $A \in \mathbb{R}^{n \times n}$, $B \in \mathbb{R}^{n \times m}$, and $x_i \in \mathbb{R}^n$, $u_i \in \mathbb{R}^m$ are the state and input of follower *i*, respectively, and x_{i0} is its initial state. Throughout this chapter, we assume that the pair (A, B) is stabilizable. Moreover, we make the following two standard assumptions regarding the communication between the leader and the followers:

Assumption 3.1. We assume that at least one follower receives the state information of the leader.

Assumption 3.2. We also assume that the underlying graph G of the communication between the followers is a connected simple undirected graph.

We consider the infinite horizon distributed linear quadratic tracking control problem for the leader-follower system (3.5) and (3.6), where the global cost functional integrates the weighted quadratic difference of states between every follower and its neighbors and the weighted quadratic difference of states between the leader and the followers communicating with the leader, and where the cost functional also penalizes the inputs in a quadratic form.

Note that, as mentioned in Assumption 3.1, at least one follower receives the state information of the leader. Thus, the leader-follower system (3.5) and (3.6) can be interconnected by a distributed diffusive control law of the form

$$u_i(t) = K \sum_{j=1}^N a_{ij}(x_i(t) - x_j(t)) + K g_i(x_i(t) - x_r(t))$$
(3.7)

where a_{ij} is the ij-th entry of the adjacency matrix \mathcal{A} of the underlying graph \mathcal{G} , $K \in \mathbb{R}^{m \times n}$ is an identical feedback gain for all followers and we have $g_i > 0$ for at least one i = 1, 2, ..., N. Accordingly, the cost functional considered in this chapter is given by

$$J(u_1, u_2, \dots, u_N) = \int_0^\infty \frac{1}{2} \sum_{i=1}^N \sum_{j=1}^N a_{ij} (x_i - x_j)^\top Q(x_i - x_j) + \sum_{i=1}^N g_i (x_i - x_r)^\top Q(x_i - x_r) + \sum_{i=1}^N u_i^\top R u_i \, dt$$
(3.8)

where $Q \ge 0$ and R > 0 are given real weighting matrices of suitable dimensions.

The distributed linear quadratic tracking problem is then the problem of minimizing the cost functional (3.8) for all initial states x_{r0} and x_{i0} , i = 1, 2, ..., N over all distributed diffusive control laws (3.7) such that the states of all followers track the state of the leader asymptotically. In that case we say the network reaches *tracking synchronization*: **Definition 3.1.** We say the control law (3.7) achieves tracking synchronization for the leader-follower system (3.5) and (3.6) if for all i = 1, 2, ..., N and for all initial states x_{r0} and x_{i0} , we have

$$x_i(t) - x_r(t) \to 0 \text{ as } t \to \infty.$$

Due to the distributed nature of the control law (3.7) as imposed by the network topology, the distributed linear quadratic tracking problem is a non-convex optimization problem [74]. It is therefore difficult, if not impossible, to find a closed form solution for an optimal controller, or such optimal controller may not even exist. Therefore, in this chapter we will design distributed control laws which solve a *suboptimality* version of this problem.

To proceed, for the *i*th follower we introduce the following error state

$$e_i = x_i - x_r$$

for i = 1, 2, ..., N. Subsequently, the dynamics of e_i is given by

$$\dot{e}_i = Ae_i + Bu_i, \quad i = 1, 2, \dots, N.$$
 (3.9)

Denoting $\mathbf{x} = (x_1^{\top}, \dots, x_N^{\top})^{\top}$, $\mathbf{u} = (u_1^{\top}, \dots, u_N^{\top})^{\top}$, and $\mathbf{e} = (e_1^{\top}, \dots, e_N^{\top})^{\top}$, we can then rewrite the error system (3.9) in compact form as

$$\dot{\mathbf{e}} = (I_N \otimes A)\mathbf{e} + (I_N \otimes B)\mathbf{u}, \quad \mathbf{e}(0) = \mathbf{e}_0.$$
(3.10)

Note that

$$\mathbf{e} = \mathbf{x} - \mathbf{1}_N \otimes x_r$$

Correspondingly, by using the fact $(L \otimes K)(\mathbf{1} \otimes x_r) = 0$, the control law (3.7) can be given by

$$\mathbf{u}(t) = (\Gamma \otimes K)\mathbf{e} \tag{3.11}$$

where $\Gamma = L + G$ and $G = \text{diag}(g_1, g_2, \dots, g_N)$. Similarly, the cost functional (3.8) can be written in terms of e and u as

$$J(\mathbf{u}) = \int_0^\infty \mathbf{e}^\top (\Gamma \otimes Q) \mathbf{e} + \mathbf{u}^\top (I_N \otimes R) \mathbf{u} \, dt.$$
(3.12)

Now, by substituting the control law (3.11) into the error dynamics (3.10), we obtain the closed-loop error system

$$\dot{\mathbf{e}} = (I_N \otimes A + \Gamma \otimes BK)\mathbf{e}, \quad \mathbf{e}(0) = \mathbf{e}_0.$$
 (3.13)

and the associated cost is now given by

$$J(K) = \int_0^\infty \mathbf{e}^\top \left(\Gamma \otimes Q + \Gamma^2 \otimes K^\top R K \right) \mathbf{e} \, dt \tag{3.14}$$

Note that the controlled leader-follower system (3.5) and (3.6) reaches tracking synchronization, i.e., the states of all followers track the state of the leader asymptotically, if and only if the error dynamics (3.13) is stable.

Let

$$B(r) = \{ \mathbf{e}_0 \in \mathbb{R}^{nN} \mid \|\mathbf{e}_0\| \leqslant r \}$$
(3.15)

be the closed ball of radius r in the state space \mathbb{R}^{nN} of the error system (3.13). Then, for the leader-follower system (3.5) and (3.6) with initial states such that the error initial state is contained in a closed ball of a given radius, we want to design a distributed diffusive controller such that tracking synchronization is achieved and, for all initial states satisfying (3.15), the associated cost is smaller than an a priori given upper bound. Thus, the problem that we will address is the following:

Problem 3.1. Consider the leader-follower multi-agent system (3.5) and (3.6) and the associated cost functional (3.8). Let r > 0 be a given radius and let $\gamma > 0$ be an a priori given upper bound for the cost. The problem is to find a distributed diffusive control law of the form (3.7) such that the controlled leader-follower system reaches tracking synchronization and, for all initial conditions \mathbf{x}_0 and \mathbf{x}_{r0} such that $\mathbf{e}_0 = \mathbf{x}_0 - \mathbf{1}_N \otimes \mathbf{x}_{r0}$ satisfies (3.15), the associated cost (3.8) is smaller than the given upper bound, i.e., $J(K) < \gamma$.

3.4 Distributed suboptimal tracking control for leaderfollower multi-agent systems

In this section, we will address Problem 3.1 and provide a suitable control design method. As mentioned before, the distributed control law (3.7) achieves tracking synchronization and suboptimal performance for the leader-follower system (3.5) and (3.6) with respect to the given tolerance on the cost functional (3.8) if and only if the error dynamics (3.13) is stable and $J(K) < \gamma$.

Now, let $U \in \mathbb{R}^{N \times N}$ be an orthogonal matrix that diagonalizes $\Gamma = L + G$. Define

$$U^{\dagger}\Gamma U := \Lambda = \operatorname{diag}(\lambda_1, \lambda_2, \dots, \lambda_N).$$

It follows from Lemma 1.1 that $\lambda_i > 0$ for all i = 1, 2, ..., N. To simplify the problem formulated in the previous section, by applying the state transformation

 $\bar{\mathbf{e}} = (U^{\top} \otimes I_n) \mathbf{e}$, system (3.13) becomes

$$\dot{\bar{\mathbf{e}}} = (I_N \otimes A + \Lambda \otimes BK) \,\bar{\mathbf{e}}, \quad \bar{\mathbf{e}}(0) = \bar{\mathbf{e}}_0 \tag{3.16}$$

where $\bar{\mathbf{e}} = (\bar{e}_1^\top, \dots, \bar{e}_N^\top)^\top$. In terms of the transformed variable, the cost (3.14) is then given by

$$J(K) = \int_0^\infty \sum_{i=1}^N \bar{e}_i^\top (\lambda_i Q + \lambda_i^2 K^\top R K) \bar{e}_i \, dt.$$
(3.17)

Note that the transformed states \bar{e}_i , i = 1, 2, ..., N appearing in system (3.16) and cost (3.17) are decoupled from each other. Then we can write system (3.16) as

$$\dot{\bar{e}}_i = (A + \lambda_i B K) \bar{e}_i, \quad i = 1, 2, \dots, N.$$
 (3.18)

Also, the cost (3.17) equals

$$J(K) = \sum_{i=1}^{N} J_i(K)$$
(3.19)

with

$$J_{i}(K) = \int_{0}^{\infty} \bar{e}_{i}^{\top} (\lambda_{i}Q + \lambda_{i}^{2}K^{\top}RK) \bar{e}_{i} dt, \quad i = 1, 2, \dots, N.$$
(3.20)

Clearly, the controlled leader-follower system (3.5) and (3.6) reaches tracking synchronization with control law (3.7) if and only if, for i = 1, 2, ..., N, the systems (3.18) are stable. In addition, the control law (3.7) is suboptimal if $J(K) < \gamma$.

So far, we have transformed the problem of distributed suboptimal control for the leader-follower system (3.5) and (3.6) into the problem of finding one single static feedback gain $K \in \mathbb{R}^{m \times n}$ such that the systems (3.18) are stable for i = 1, 2, ..., N and $J(K) < \gamma$. Since the pair (A, B) is stabilizable, there exists such a feedback gain K [54, 122].

The following lemma then provides a necessary and sufficient condition for a given feedback gain K to stabilize all systems (3.18) and for given initial states guarantee that $J(K) < \gamma$.

Lemma 3.2. Let K be a feedback gain. Consider the systems (3.18) with given initial states $\bar{e}_{10}, \bar{e}_{20}, \ldots, \bar{e}_{N0}$ and associated cost functionals (3.19) and (3.20). Let $\gamma > 0$. Then all systems (3.18) are stable and $J(K) < \gamma$ if and only if there exist $P_i > 0$ satisfying

$$(A + \lambda_i BK)^{\top} P_i + P_i (A + \lambda_i BK) + \lambda_i Q + \lambda_i^2 K^{\top} RK < 0$$
(3.21)

and

$$\sum_{i=1}^{N} \bar{e}_{i0}^{\top} P_i \bar{e}_{i0} < \gamma, \tag{3.22}$$

for $i = 1, 2, \ldots, N$, respectively.

Proof. (if) Since (3.22) holds, there exist $\gamma_i := \bar{e}_{i0}^\top P_i \bar{e}_{i0} + \epsilon_i$ with sufficiently small $\epsilon_i > 0, i = 1, 2, ..., N$ such that $\sum_{i=1}^N \gamma_i < \gamma$. Because there exists $P_i > 0$ such that (3.21) and $\bar{e}_{i0}^\top P_i \bar{e}_{i0} < \gamma_i$ holds for all i = 1, 2, ..., N, by taking $\bar{A} = A + \lambda_i BK$ and $\bar{Q} = \lambda_i Q + \lambda_i^2 K^\top RK$, it follows from Theorem 3.1 that all systems (3.18) are stable and $J_i(K) < \gamma_i$ for i = 1, 2, ..., N. Since $J(K) = \sum_{i=1}^N J_i(K)$, this implies that $J(K) < \sum_{i=1}^N \gamma_i < \gamma$.

(only if) Since $J(K) < \gamma$ and $J(K) = \sum_{i=1}^{N} J_i(K)$, there exist $\gamma_i := J_i(K) + \epsilon_i$ with sufficiently small $\epsilon_i > 0$, i = 1, 2, ..., N such that $\sum_{i=1}^{N} \gamma_i < \gamma$. Because all systems (3.18) are stable and $J_i(K) < \gamma_i$ for i = 1, 2, ..., N, by taking $\bar{A} = A + \lambda_i BK$ and $\bar{Q} = \lambda_i Q + \lambda_i^2 K^\top RK$, it again follows from Theorem 3.1 that there exist $P_i > 0$ such that (3.21) and $\bar{x}_{i0}^\top P_i \bar{x}_{i0} < \gamma_i$ hold for all i = 1, 2, ..., N. Since $\sum_{i=1}^{N} \gamma_i < \gamma$, this implies that $\sum_{i=1}^{N} \bar{x}_{i0}^\top P_i \bar{x}_{i0} < \sum_{i=1}^{N} \gamma_i < \gamma$.

Lemma 3.2 establishes a necessary and sufficient condition for a given feedback gain *K* to stabilize all systems (3.18) and to satisfy, for given initial states of these systems, $J(K) < \gamma$. However, Lemma 3.2 does not yet provide a design method for computing such *K*. Therefore, in the following we will provide a method to find such *K*.

Lemma 3.3. Consider the leader-follower system (3.5) and (3.6) with associated cost functional (3.8). Let x_{r0} be the given initial state of the leader and x_{i0} , i = 1, 2..., N be the given initial states of the followers, respectively. Let $\gamma > 0$ be a given tolerance. Let c be any real number such that $0 < c < \frac{2}{\lambda_N}$. We distinguish two cases:

(*i*) *if*

$$\frac{2}{\lambda_1 + \lambda_N} \leqslant c < \frac{2}{\lambda_N},\tag{3.23}$$

then there exists P > 0 satisfying

$$A^{\top}P + PA + (c^2\lambda_N^2 - 2c\lambda_N)PBR^{-1}B^{\top}P + \lambda_NQ < 0.$$
(3.24)

(ii) if

$$0 < c < \frac{2}{\lambda_1 + \lambda_N},\tag{3.25}$$

then there exists P > 0 satisfying

$$A^{\top}P + PA + (c^{2}\lambda_{1}^{2} - 2c\lambda_{1})PBR^{-1}B^{\top}P + \lambda_{N}Q < 0.$$
(3.26)

In both cases, if in addition P satisfies

(*i*) *if*

$$\sum_{i=1}^{N} (x_{i0} - x_{r0})^{\top} P(x_{i0} - x_{r0}) < \gamma,$$
(3.27)

then the distributed control law (3.7) with $K := -cR^{-1}B^{\top}P$ achieves tracking synchronization for the controlled leader-follower system (3.5) and (3.6), and with the initial states x_{r0} and x_{i0} we have $J(K) < \gamma$.

Proof. We will only give the proof for case (i) above. Using the upper and lower bounds on *c* given by (3.23), it can be verified that $c^2 \lambda_i^2 - 2c\lambda_i \leq c^2 \lambda_N^2 - 2c\lambda_N < 0$ for i = 1, 2, ..., N. It is then easily seen that (3.24) has many positive definite solutions. Since also $\lambda_i \leq \lambda_N$, any such solution *P* is a solution to the N - 1 Riccati inequalities

$$A^{\top}P + PA + (c^{2}\lambda_{i}^{2} - 2c\lambda_{i})PBR^{-1}B^{\top}P + \lambda_{i}Q < 0, \quad i = 1, 2, \dots, N.$$
(3.28)

Equivalently, P also satisfies the Lyapunov inequalities

$$(A - c\lambda_i BR^{-1}B^{\top}P)^{\top}P + P(A - c\lambda_i BR^{-1}B^{\top}P) + \lambda_i Q + c^2 \lambda_i^2 PBR^{-1}B^{\top}P < 0, \quad i = 1, 2, \dots, N.$$
(3.29)

Next, by substituting $\bar{\mathbf{e}} = (U^{\top} \otimes I_n)\mathbf{e}$ into (3.22) we have $\sum_{i=1}^{N} e_{i0}^{\top} P e_{i0} < \gamma$, which is equal to (3.27).

Next, taking $P_i = P$ for i = 1, 2, ..., N and $K := -cR^{-1}B^{\top}P$ in (3.21) and (3.22) immediately gives us (3.29) and (3.27). Then it follows from Lemma 3.2 that all systems (3.18) are stable and $J(K) < \gamma$. Subsequently, the controlled leader-follower system reaches tracking synchronization and $J(K) < \gamma$.

We will now apply Lemma 3.3 to establish a solution to Problem 3.1. The next theorem provides a condition under which, for given radius r and upper bound γ , distributed diffusive suboptimal control laws exist, and explains how these can be computed.

Theorem 3.4. Consider the leader-follower system (3.5) and (3.6) with associated cost functional (3.8). Let r > 0 be a given radius and let $\gamma > 0$ be an a priori given upper bound for the cost. Let c be any real number such that $0 < c < \frac{2}{\lambda_N}$. We distinguish two cases:

$$\frac{2}{\lambda_1 + \lambda_N} \leqslant c < \frac{2}{\lambda_N},\tag{3.30}$$

then there exists P > 0 satisfying

$$A^{\top}P + PA + (c^2\lambda_N^2 - 2c\lambda_N)PBR^{-1}B^{\top}P + \lambda_N Q < 0.$$
(3.31)

(ii) if

$$0 < c < \frac{2}{\lambda_1 + \lambda_N},\tag{3.32}$$

then there exists P > 0 satisfying

$$A^{\top}P + PA + (c^{2}\lambda_{1}^{2} - 2c\lambda_{1})PBR^{-1}B^{\top}P + \lambda_{N}Q < 0.$$
(3.33)

In both cases, if in addition P satisfies

$$P < \frac{\gamma}{r^2} I, \tag{3.34}$$

then the distributed control law (3.7) with $K := -cR^{-1}B^{\top}P$ achieves tracking synchronization for the controlled leader-follower system (3.5) and (3.6) and $J(K) < \gamma$ for all initial states x_{r0} and \mathbf{x}_0 satisfying

$$\boldsymbol{x}_0 - \boldsymbol{1}_N \otimes \boldsymbol{x}_{r0} \in B(r). \tag{3.35}$$

Proof. Again, we only give proof for case (i). Let P > 0 satisfy (3.31) and (3.34). Next, we will show that if the initial states x_{r0} and \mathbf{x}_0 satisfy $\mathbf{x}_0 - \mathbf{1}_N \otimes x_{r0} \in B(r)$, then (3.27) holds. Indeed, if $\|\mathbf{x}_0 - \mathbf{1}_N \otimes x_{r0}\| \leq r$, then

$$\sum_{i=1}^{N} (x_{i0} - x_{r0})^{\top} P(x_{i0} - x_{r0})$$
$$= (\mathbf{x}_0 - \mathbf{1}_N \otimes x_{r0})^{\top} (I \otimes P) (\mathbf{x}_0 - \mathbf{1}_N \otimes x_{r0})$$
$$< \frac{\gamma}{r^2} \|\mathbf{x}_0 - \mathbf{1}_N \otimes x_{r0}\|^2 \leq \gamma.$$

It then follows from Lemma 3.3 that the controlled leader-follower system (3.5) and (3.6) reaches tracking synchronization with the given *K* and $J(K) < \gamma$ for all initial states x_{r0} and \mathbf{x}_0 satisfying (3.35).

Remark 3.5. Theorem 3.4 states that after choosing *c* satisfying the inequality (3.30) for case (i) and finding P > 0 satisfying (3.31) and (3.34), the distributed control law with local gain $K = -cR^{-1}B^{\top}P$ is γ -suboptimal for all initial states of the leader-follower system satisfying the condition (3.35). According to (3.34), the smaller the solution *P* of (3.31), the smaller the quotient $\frac{\gamma}{r^2}$ is allowed to be, leading to a smaller upper bound and a larger radius. The question then arises: how should we choose the parameter *c* in (3.30) so that the Riccati inequality (3.31)

allows a positive definite solution that is as small as possible. In fact, one can find a positive definite solution $P(c, \epsilon)$ to (3.31) by solving the Riccati equation

$$A^{\top}P + PA - PB\bar{R}(c)^{-1}B^{\top}P + \bar{Q}(\epsilon) = 0$$
(3.36)

with $\bar{R}(c) = \frac{1}{-c^2 \lambda_N^2 + 2c\lambda_N} R$ and $\bar{Q}(\epsilon) = \lambda_N Q + \epsilon I_n$ where c is chosen as in (3.30) and $\epsilon > 0$. If c_1 and c_2 as in (3.30) satisfy $c_1 \leq c_2$, then we have $\bar{R}(c_1) \leq \bar{R}(c_2)$, so, clearly, $P(c_1, \epsilon) \leq P(c_2, \epsilon)$. Similarly, if $0 < \epsilon_1 \leq \epsilon_2$, we immediately have $\bar{Q}(\epsilon_1) \leq \bar{Q}(\epsilon_2)$. Again, it follows that $P(c, \epsilon_1) \leq P(c, \epsilon_2)$. Therefore, if we choose $\epsilon > 0$ very close to 0 and $c = \frac{2}{\lambda_1 + \lambda_N}$, we find the 'best' solution to the Riccati inequality (3.31) in the sense explained above.

Likewise, if *c* satisfies (3.32) corresponding to case (ii), it can be shown that if we choose $\epsilon > 0$ very close to 0 and c > 0 very close to $\frac{2}{\lambda_1 + \lambda_N}$, we find the 'best' solution to the Riccati inequality (3.33) in the sense explained above.

3.5 Simulation example

In this section, we will use a numerical example borrowed from [76] to illustrate the design method for the distributed suboptimal control laws given in Theorem 3.4.

Consider a leader-follower multi-agent system, consisting of one leader and five followers. The dynamics of the leader is given by

$$\dot{x}_r(t) = Ax_r(t), \quad x_r(0) = x_{r0},$$

and the dynamics of the followers are identical and represented by

$$\dot{x}_i(t) = Ax_i(t) + Bu_i(t), \quad x_i(0) = x_{i0}, \quad i = 1, 2, \dots, 5$$

where

$$A = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix}, \quad B = \begin{pmatrix} 0 \\ 1 \end{pmatrix}.$$

The pair (A, B) is stabilizable. Assume the underlying graph representing the communication between the leader and the followers is given as in Figure 3.1. The graph representing the communication between the followers is then the



Figure 3.1: The underlying graph of the communication between the leader and the followers.

undirected cycle graph with the Laplacian matrix

$$L = \begin{pmatrix} 2 & -1 & 0 & 0 & -1 \\ -1 & 2 & -1 & 0 & 0 \\ 0 & -1 & 2 & -1 & 0 \\ 0 & 0 & -1 & 2 & -1 \\ -1 & 0 & 0 & -1 & 2 \end{pmatrix}.$$

Since the leader shares its state information only with follower 2, it follows from Lemma 1.1 that the associated diagonal matrix $G = \text{diag}(g_1, g_2, \dots, g_5) = \text{diag}(0, 1, 0, 0, 0)$. Furthermore, we consider the cost functional

$$J(u_1, u_2, \dots, u_5) = \int_0^\infty \frac{1}{2} \sum_{i=1}^5 \sum_{j=1}^5 a_{ij} (x_i - x_j)^\top Q(x_i - x_j) + \sum_{i=1}^5 g_i (x_i - x_r)^\top Q(x_i - x_r) + \sum_{i=1}^5 u_i^\top R u_i \, dt$$

with

$$Q = \begin{pmatrix} 2 & 0\\ 0 & 1 \end{pmatrix}, \quad R = 1.$$

Let the desired tolerance for the cost functional be $\gamma = 20$. Our aim is then to design a control law of the form

$$u_i(t) = K \sum_{j=1}^{5} a_{ij}(x_i(t) - x_j(t)) + K g_i(x_i(t) - x_r(t))$$
(3.37)

such that the controlled leader-follower system reaches tracking synchronization



Figure 3.2: Plots of the states x_{r1} and $\mathbf{x}^1 = (x_{1,1}, x_{2,1}, \dots, x_{5,1})$ (upper plot) and x_{r2} and $\mathbf{x}^2 = (x_{1,2}, x_{2,2}, \dots, x_{5,2})$ (lower plot) of the six decoupled local agents without control

and the associated cost satisfies J(K) < 20 for all initial states \mathbf{x}_0 and x_{r0} satisfying the condition $\|\mathbf{x}_0 - \mathbf{1}_5 \otimes x_{r0}\| \leq r$ with radius r to be specified later.

In this simulation example, we will use the design method of case (i) in Theorem 3.4. For $\Gamma = L + G$ the smallest and largest eigenvalues are $\lambda_1 = 0.1392$ and $\lambda_5 = 4.1149$, respectively. We first compute a solution P > 0 to (3.31) by solving

$$A^{\top}P + PA + (c^2\lambda_5^2 - 2c\lambda_5)PBR^{-1}B^{\top}P + \lambda_5Q + \epsilon I_2 = 0$$
(3.38)

with ϵ sufficiently small as mentioned in Remark 3.5. Here we choose $\epsilon = 0.01$. In addition, we choose $c = \frac{2}{\lambda_1 + \lambda_5} = 0.4701$, which is the 'best' choice as mentioned in Remark 3.5. Then by solving (3.38) using Matlab, we compute

$$P = \begin{pmatrix} 13.2553 & 3.3886\\ 3.3886 & 9.2760 \end{pmatrix}.$$

Correspondingly, the control gain is equal to $K = (1.5931 \ 4.3610)$. We now compute the radius r of a ball B(r) of initial states for which the distributed control law (3.37) is suboptimal, i.e. J(K) < 20. We compute that the largest eigenvalue of P is equal to 15.1952. Hence for every radius r such that $\frac{20}{r^2} > 15.1952$ the



Figure 3.3: Plots of the states x_{r1} and $\mathbf{x}^1 = (x_{1,1}, x_{2,1}, \dots, x_{5,1})$ (upper plot) and x_{r2} and $\mathbf{x}^2 = (x_{1,2}, x_{2,2}, \dots, x_{5,2})$ (lower plot) of the controlled leader-follower system

inequality (3.34) holds. Thus, the distributed controller with local gain *K* is suboptimal for all x_{r0} and \mathbf{x}_0 satisfying $\|\mathbf{x}_0 - \mathbf{1}_5 \otimes x_{r0}\| \leq r$ with r < 1.1473.

As an example, the following initial states of the agents satisfy this norm bound: $x_{r0}^{\top} = (0.3 - 0.5), x_{10}^{\top} = (0.7 - 0.2), x_{20}^{\top} = (0.3 - 0.6), x_{30}^{\top} = (0.2 - 0.3), x_{40}^{\top} = (-0.1 - 0.7), x_{50}^{\top} = (0.2 - 0.6).$ The plots of the state of the six local agents without control are shown in Figure 3.2. Figure 3.3 shows that the controlled leader-follower system reaches tracking synchronization.

3.6 Conclusions

In this chapter, we have studied the distributed linear quadratic tracking control problem for leader-follower multi-agent systems. We have considered leader-follower systems consisting of one autonomous leader and *N* followers, together with an associated global cost functional. We assume that the leader shares its state information with at least one of the followers and the underlying graph connecting the followers is a connected simple undirected graph. For this type of leader-follower systems, we have provided a design method to compute distributed suboptimal control laws such that the controlled network reaches tracking

synchronization and the associated cost is smaller than a given tolerance for all initial states bounded in norm by a given radius. The computation of the local gain involves the solution of a single Riccati inequality, whose dimension is equal to the dimension of the agent dynamics, and also involves the largest and smallest eigenvalue of a positive definite matrix capturing the underlying graph structure.

Distributed linear quadratic control: compute locally and act globally

In this chapter we consider the distributed linear quadratic control problem for networks of agents with single integrator dynamics. We first establish a general formulation of the distributed LQ problem and show that the optimal control gain depends on global information on the network. Thus, the optimal protocol can only be computed in a centralized fashion. In order to overcome this drawback, we propose the design of protocols that are computed in a decentralized way. We will write the global cost functional as a sum of *local* cost functionals, each associated with one of the agents. In order to achieve 'good' performance of the controlled network, each agent then computes its own local gain, using sampled information of its neighboring agents. This decentralized computation will only lead to suboptimal global network behavior. However, we will show that the resulting network will reach consensus.

4.1 Introduction

The distributed linear quadratic (LQ) optimal control problem is the problem of interconnecting a finite number of identical agents according to a given network graph to achieve consensus optimally. Each agent receives input only from its neighbors, in the form of a linear feedback of the relative states amplified by a certain constant gain. Such control law is called a *distributed diffusive control law*. The problem of minimizing a given quadratic cost functional over all distributed diffusive control laws that achieve consensus is then called the distributed LQ problem corresponding to this cost functional.

In the case that the agent dynamics is given by a general state space system, this optimal control problem is non-convex and difficult to solve, and it is unclear whether a solution exists in general, see [37]. In contrast, for the case of single integrator dynamics it is fairly easy to find an explicit expression for the distributed diffusive optimal control law, see, for example, [10]. Although a solution to the problem is available, it turns out however that *global information* on

the network is needed to compute this optimal control law. More specifically, the distributed diffusive optimal control law can be computed only by a (virtual) supervisor that knows the network graph and the initial states of all the agents. Thus, although the resulting optimal control law *operates* in a distributed fashion, its actual computation can only be performed in a *centralized* way.

Formulating the distributed LQ problem as a problem of minimizing a *global* cost functional is therefore not practical. Indeed, the centralized computation requires that the local optimal gains needs to be re-designed by the supervisor in case that changes in the network occur. For example, by adding or removing agents from the network, its graph will change, and new initial states will occur while existing ones will disappear.

In this chapter we will address this drawback and present a *decentralized* design method to compute a distributed controller: each agent will compute its own local control law. For this computation, the agent will not need knowledge of the network graph or the initial states of all other agents. This will then enable 'plug-and-play' operations on the network, since each agent will be able to automatically recompute its local gain whenever a new agent is added or removed.

In order to achieve this decentralized computation scheme we will write the original global cost functional as the sum of *local* LQ tracking cost functionals, each associated with one of the agents. The agents can not solve these optimal tracking problems explicitly because the reference signals depend on the future dynamics of the neighbors. However, using sampling, suboptimal local gains are obtained. This decentralized computation will not necessarily result in optimality of the global network behavior. We will however show that the resulting network will reach consensus.

The distributed LQ control problem has attracted much attention in the past, see e.g. [7, 10, 73, 74]. In [7], a distributed suboptimal controller for a global cost functional was developed to stabilize a network with general agent dynamics. A similar cost functional was also considered in [16] for designing distributed controllers with guaranteed performance. The distributed LQ control problem with general agent dynamics was also dealt with in [75] and [123] by adopting an inverse optimal control approach. In [98] a game theoretic approach was considered to obtain a suboptimal solution. Also, [37] considers a suboptimal version of this problem. In [76], a suboptimal consensus controller design was developed by employing a hierarchical LQ control approach for an appropriately chosen global performance index, and a similar idea for constructing a particular cost functional was employed in [77] to design a reduced order distributed controller. In [90] a distributed optimal control method was adopted to decouple a class of linear multi-agent systems with state coupled nonlinear uncertainties.

The common feature of all work referred to above is that the computation of the control gains needs global information on the network. This disadvantage can be avoided by adopting adaptive control methods [52] or by using reinforcement learning [108], [68]. In [70] and [109], it was shown that diffusive couplings are necessary for minimization of cost functionals of a particular form, involving the weighted squared synchronization error.

Below we list the contributions of this chapter.

- 1. We show that for agents with single integrator dynamics, in any distributed LQ cost functional, the state weighting matrix must be equal to a weighted square of the Laplacian of the network graph.
- 2. We give a solution to this general distributed LQ problem, and show that computation of the optimal protocol requires exact knowledge of the Laplacian as well as the initial state of the entire network.
- 3. We represent the global cost functional as a sum of local LQ tracking cost functionals, one for each agent. Using sampling, suboptimal local gains are obtained. Computation of these gains is completely decentralized.
- 4. We show that these gains lead to a protocol that achieves consensus of the network.

The outline of this chapter is as follows. In Section 4.2, we derive a general formulation of the distributed LQ problem. In Section 4.3, we show that computation of the optimal control laws requires complete knowledge of the network graph and the initial state of the entire network. In Section 4.4, we propose a decentralized method to compute suboptimal (local) control laws. In order to do this, we need to apply ideas from linear quadratic tracking, and these are reviewed in Section 4.5. Then, in Section 4.6, we compute these local control laws, and show that the network reaches consensus if all agents apply their own local gain. To illustrate the designed control protocol, a simulation example is provided in Section 4.7. Finally, in Section 4.8, we give some concluding remarks.

4.2 The general form of a distributed linear quadratic cost functional

In this section we will show that in any distributed LQ cost functional, the state weighting matrix must be a weighted square of the Laplacian of the network graph. We will also give two important examples of distributed LQ cost functionals.

We consider a network of agents described by scalar single integrator dynamics

$$\dot{x}_i(t) = u_i(t), \quad x_i(0) = x_{i0}, \quad i = 1, 2, \dots, N,$$
(4.1)

with $x_{i0} \in \mathbb{R}$ the initial state of agent *i*. By collecting the states and inputs of the individual agents into the vectors $\mathbf{x} = (x_1, x_2, \dots, x_N)^{\top}$ and $\mathbf{u} = (u_1, u_2, \dots, u_N)^{\top}$, (4.1) can be written as

$$\dot{\mathbf{x}}(t) = \mathbf{u}(t), \quad \mathbf{x}(0) = \mathbf{x}_0. \tag{4.2}$$

A general class of LQ cost functionals are those of the form

$$J(\mathbf{x}_0, \mathbf{u}) = \int_0^\infty \mathbf{x}^\top(t) Q \mathbf{x}(t) + \mathbf{u}^\top(t) R \mathbf{u}(t) dt, \qquad (4.3)$$

where $Q \in \mathbb{R}^{N \times N}$, $R \in \mathbb{R}^{N \times N}$ and $Q \ge 0$ and R > 0.

In the context of distributed LQ control we only allow distributed diffusive control laws that achieve consensus, i.e. the controlled trajectories converge to $im(\mathbf{1}_N)$, the span of the vector of ones. Thus the class of control laws over which we want to minimize (4.3) consists of those of the form $\mathbf{u} = -gL\mathbf{x}$, with $L \in \mathbb{R}^{N \times N}$ the Laplacian of the network graph and where g > 0, see e.g. [85].

We will now show that for a cost functional (4.3) to make sense in this context, the weighting matrix Q must be of the form Q = LWL for some positive semidefinite matrix W.

Lemma 4.1. $J(\mathbf{x}_0, \mathbf{u}) < \infty$ for all $\mathbf{x}_0 \in \mathbb{R}^N$ and control laws of the form $\mathbf{u} = -gL\mathbf{x}$ with g > 0 only if there exists a positive semi-definite $W \in \mathbb{R}^{N \times N}$ such that Q = LWL.

Proof. Write $Q = C^T C$ for some C. Now, let $\bar{\mathbf{x}}(t)$ denote any nonzero state trajectory generated by the control law $\mathbf{u} = -gL\mathbf{x}$ with g > 0 and let $\bar{\mathbf{u}}(t) = -gL\bar{\mathbf{x}}(t)$. It is well known that this control law achieves consensus (see [85]) so we have $\bar{\mathbf{x}}(t) \rightarrow c\mathbf{1}_N$ for some nonzero constant c. Now assume that the control law $\mathbf{u} = -gL\mathbf{x}$ gives finite cost, i.e. $J(\mathbf{x}_0, \bar{\mathbf{u}}) < \infty$. This implies

$$\int_0^\infty \bar{\mathbf{x}}^\top(t) C^\top C \bar{\mathbf{x}}(t) dt < \infty$$

and hence $C\bar{\mathbf{x}}(t) \to 0$. Thus we obtain $\mathbf{1}_N \in \ker(C)$, equivalently, $\ker(L) \subset \ker(C)$. We thus conclude that there exists a matrix V such that C = VL so the state weighting matrix Q must be of the form $Q = LV^{\top}VL$ for some matrix V. This proves our claim.

We have thus shown that, for a general LQ cost functional to make sense in the context of distributed diffusive control for multi-agent systems, it must necessarily be of the form

$$J(\mathbf{x}_0, \mathbf{u}) = \int_0^\infty \mathbf{x}^\top(t) L W L \mathbf{x}(t) + \mathbf{u}^\top(t) R \mathbf{u}(t) dt, \qquad (4.4)$$

for some $W \ge 0$ and R > 0. The corresponding distributed LQ problem is to

minimize, for the system (4.2) with initial state \mathbf{x}_0 , the cost functional (4.4) over all control laws of the form $\mathbf{u} = -gL\mathbf{x}$ with g > 0.

As an illustration, we will now provide two important special cases of LQ cost functionals. The first one was studied before in [37] and [10]:

$$J(\mathbf{x}_0, \mathbf{u}) = \sum_{i=1}^N \int_0^\infty \sum_{j \in \mathcal{N}_i} q(x_i(t) - x_j(t))^2 + r u_i^2(t) dt,$$
(4.5)

where q and r are positive real numbers. Clearly, (4.5) is equal to

$$J(\mathbf{x}_0, \mathbf{u}) = \int_0^\infty \mathbf{x}^\top(t) 2q L \mathbf{x}(t) + r \mathbf{u}^\top(t) \mathbf{u}(t) dt.$$

Note that $2qL = L(2qL^{\dagger})L$ with L^{\dagger} the Moore-Penrose inverse of L (which is indeed positive semi-definite). Thus this cost functional is of the form (4.4) with $W = 2qL^{\dagger}$ and R = rI.

As a second example, consider

$$J(\mathbf{x}_0, \mathbf{u}) = \sum_{i=1}^{N} \int_0^\infty q \left(x_i(t) - a_i(t) \right)^2 + r u_i^2(t) dt,$$
(4.6)

with

$$a_i(t) := \frac{1}{d_i + 1} \left(x_i(t) + \sum_{j \in \mathcal{N}_i} x_j(t) \right).$$
(4.7)

Here, q and r are positive weights, d_i denotes the node degree of agent i and \mathcal{N}_i its set of neighbors. The idea of the cost functional (4.6) is to minimize the sum of the deviations between the state $x_i(t)$ and the average $a_i(t)$ of the states of its neighbors (including itself) and the control energy. In order to put this in the form (4.4), define

$$G := (D + I_N)^{-1} (A + I_N) \in \mathbb{R}^{N \times N},$$
(4.8)

where $D \in \mathbb{R}^{N \times N}$ is the degree matrix and $A \in \mathbb{R}^{N \times N}$ the adjacency matrix. Then clearly

$$\mathbf{a}(t) = G\mathbf{x}(t),$$

where $\mathbf{x} = (x_1, x_2, \dots, x_N)^{\top}$ and $\mathbf{a} = (a_1, a_2, \dots, a_N)^{\top}$. It is then easily seen that

$$J(\mathbf{x}_0, \mathbf{u}) = \int_0^\infty q \mathbf{x}^\top (t) (I_N - G)^\top (I_N - G) \mathbf{x}(t) + r \mathbf{u}^\top (t) \mathbf{u}(t) dt.$$

Since $(I_N - G)^{\top}(I_N - G) = L(D + I_N)^{-2}L$, we conclude that (4.6) is a special case of (4.4) with $W = q(D + I_N)^{-2}$ and $R = rI_N$.

4.3 Centralized computation of the optimal control gain

In this section we will briefly give a solution to the general distributed LQ problem with cost functional (4.4) as introduced in Section 4.2, thus generalizing the result from [10] to general distributed LQ cost functionals. We will show that, indeed, computation of the optimal protocol requires global information on the network graph and the initial state of the entire network.

Consider the cost functional (4.4) together with the dynamics (4.2) with given initial state \mathbf{x}_0 . Since the admissible control laws are given by $\mathbf{u} = -gL\mathbf{x}$, the associated state trajectory is $\mathbf{x}(t) = e^{-gLt}\mathbf{x}_0$ and $\mathbf{u}(t) = -gL\mathbf{x}(t)$. Substituting this into the cost functional yields

$$J(g) := \mathbf{x}_0^\top \left(\int_0^\infty e^{-gLt} \left(LWL + g^2 LRL \right) e^{-gLt} dt \right) \mathbf{x}_0$$
(4.9)

Clearly, we need to minimize J(g) over g > 0. Substituting $gt = \tau$, we find

$$J(g) := \mathbf{x}_0^\top \int_0^\infty e^{-\tau L} \left(\frac{1}{g} LWL + gLRL\right) e^{-\tau L} d\tau \, \mathbf{x}_0.$$

Define

$$X_0 := \int_0^\infty e^{-\tau L} L W L e^{-\tau L} d\tau$$

and

$$Y_0 := \int_0^\infty e^{-\tau L} LR L e^{-\tau L} d\tau.$$

It turns out that both integrals indeed exist, and can be computed as particular solutions of the Lyapunov equations

$$-LX - XL + LWL = 0, (4.10a)$$

$$-LY - YL + LRL = 0. \tag{4.10b}$$

Indeed, although *L* is not Hurwitz, these equations do have positive semi-definite solutions *X* and *Y* and, in fact, *X*₀ is the unique positive semi-definite solution *X* to (4.10a) with the property that $im(\mathbf{1}_N) \subset ker(X)$. Likewise *Y*₀ is the unique positive semi-definite solution *Y* of (4.10b) with the property that $im(\mathbf{1}_N) \subset ker(Y)$ (see Proposition 1 in [44]). It follows from (4.10b) that, in fact, $ker(Y_0) = im(\mathbf{1}_N)$. Thus we see that

$$J(g) = \frac{1}{g} \mathbf{x}_0^\top X_0 \mathbf{x}_0 + g \mathbf{x}_0^\top Y_0 \mathbf{x}_0.$$

In order to minimize J(g) we distinguish three cases. (i) If $\mathbf{x}_0 \in \ker(Y_0) = \operatorname{im}(\mathbf{1}_N)$ then we must have $\mathbf{x}_0 \in \ker(X_0)$ as well, so J(g) = 0 for all θ and every g > 0 is optimal. (ii) If $\mathbf{x}_0^{\top} Y_0 \mathbf{x}_0 > 0$ and $\mathbf{x}_0^{\top} X_0 \mathbf{x}_0 = 0$ then no optimal g > 0 exists. (iii) If $\mathbf{x}_0^{\top} Y_0 \mathbf{x}_0 > 0$ and $\mathbf{x}_0^{\top} X_0 \mathbf{x}_0 > 0$ then an optimal g > 0 exists and can be shown to be equal to

$$g^* = \left(\frac{\mathbf{x}_0^\top X_0 \mathbf{x}_0}{\mathbf{x}_0^\top Y_0 \mathbf{x}_0}\right)^{\frac{1}{2}}$$

It is clear that the computation of the optimal gain g requires exact knowledge of the network graph in the form of the Laplacian L. Also, the optimal gain clearly depends on the global initial state of the network.

4.4 Towards decentralized computation

In this section we will propose a new approach to compute 'good' local gains that can be computed in a decentralized way. Instead of doing this for the general LQ cost functional (4.4), we will zoom in on the particular case given by (4.6) - (4.7).

In order to decentralize the computation, instead of minimizing the global cost functional (4.6) for the multi-agent system (4.2), we write it as a sum of *local* cost functionals, one for each agent in the network.

More specifically, the associated local cost functional for agent *i* is given by

$$J_i(u_i) = \int_0^\infty q \left(x_i(t) - a_i(t) \right)^2 + r u_i^2(t) \, dt, \tag{4.11}$$

where $a_i(t)$ is defined in (4.7), for i = 1, 2, ..., N. This local cost functional penalizes the squared difference between the state of the *i*th agent and the average of the states of its neighboring agents (including itself), and the local control energy. By minimizing (4.11), agent *i* would make the difference between its own state and the average of the states of its neighbors (including itself) small. Note, however, that it is impossible for agent *i* to minimize this local cost functional since the trajectory $a_i(t)$ for $t \in [0, \infty)$ associated with the neighboring agents is *not known*, so also not available to the *i*th agent. Thus, because direct minimization of (4.11) is impossible, as an alternative we will replace each of these local optimal control problems by *a sequence of linear quadratic tracking problems* that do turn out to be tractable.

More specifically, we choose a sampling period T > 0, and introduce the following sampling procedure. For each nonnegative integer k, at time t = kT the *i*th agent receives the sampled state value $x_j(kT)$ of its neighboring agents and

takes the average of these, which is given by

$$a_i(kT) = \frac{1}{d_i + 1} \left(x_i(kT) + \sum_{j \in \mathcal{N}_i} x_j(kT) \right).$$
(4.12)

Then, the *i*th agent minimizes the cost functional

$$J_{i,k}(u) = \int_0^\infty e^{-2\alpha t} \left(q \left(x_i(t) - a_i(kT) \right)^2 + r u_i^2(t) \right) dt.$$
(4.13)

In fact, this is a discounted linear quadratic tracking problem with constant reference signal $a_i(kT)$ and discount factor $\alpha > 0$. By solving this linear quadratic tracking problem, agent *i* obtains an optimal control law over an infinite time interval. However, agent *i* applies this control law only on the time interval [kT, (k + 1)T).

Then, at time t = (k+1)T the above procedure is repeated, i.e. agent *i* receives the updated average $a_i((k+1)T)$, and subsequently solves the discounted tracking problem with cost functional $J_{i,k+1}(u)$ which involves the constant updated reference signal $a_i((k+1)T)$. By performing this control design procedure sequentially at each sampling time kT, we then obtain a single control law for agent *i* over the entire interval $[0, \infty)$.

Based on this control design procedure for the individual agents, we will obtain a distributed control protocol for the entire multi-agent system, simply by letting all agents compute their own control law. In the sequel we will analyze this protocol and show that it achieves consensus for the network:

Definition 4.1. A distributed control protocol is said to achieve consensus for the network if $x_i(t) - x_j(t) \rightarrow 0$ as $t \rightarrow \infty$ for all initial states of agents *i* and *j*, for all i, j = 1, 2, ..., N.

In order to obtain an explicit expression for the control protocol proposed above, we will study the linear quadratic tracking problem for a single linear system. This will be done in the next section.

4.5 The discounted linear quadratic tracking problem

In this section, we will deal with the discounted linear quadratic tracking problem for a given linear system. The linear quadratic tracking problem has been studied before, see e.g. [67]. Here, however, we will solve it by transforming it into a standard linear quadratic control problem.

Consider the linear time-invariant system

$$\dot{x}(t) = Ax(t) + Bu(t), \quad x(0) = x_0,$$
(4.14)

with $A \in \mathbb{R}^{n \times n}$ and $B \in \mathbb{R}^{n \times m}$, and where $x(t) \in \mathbb{R}^n$, $u(t) \in \mathbb{R}^m$ denote the state and the input, respectively. We assume that the pair (A, B) is stabilizable. Given is also a constant reference signal $r_{ref}(t) = r$ with $r \in \mathbb{R}^n$. Next, we introduce a discounted quadratic cost functional given by

$$J(u) = \int_0^\infty e^{-2\alpha t} [(x(t) - r)^\top Q (x(t) - r) + u^\top (t) R u(t)] dt$$
(4.15)

where $Q \in \mathbb{R}^{n \times n}$, $R \in \mathbb{R}^{m \times m}$ and Q > 0 and R > 0 are given weight matrices and $\alpha > 0$ is a discount factor [67]. The linear quadratic tracking problem is to determine for every initial state x_0 a piecewise continuous input function u(t) that minimizes the cost functional (4.15).

To solve this problem, we introduce the variables

$$z(t) = e^{-\alpha t} x(t), \quad z_r(t) = e^{-\alpha t} r, \quad v(t) = e^{-\alpha t} u(t),$$
 (4.16)

and denote $\xi(t) = (z^{\top}(t), z_r^{\top}(t))^{\top}$. Then we obtain an auxiliary system in terms of ξ and v, given by

$$\dot{\xi}(t) = A_e \xi(t) + B_e v(t), \quad \xi_0 = (x_0^{\top}, r^{\top})^{\top},$$

where $\xi_0 \in \mathbb{R}^{2n}$ is the initial state and

$$A_e = \begin{pmatrix} A - \alpha I_n & 0\\ 0 & -\alpha I_n \end{pmatrix}, \quad B_e = \begin{pmatrix} B\\ 0 \end{pmatrix}.$$

In terms of the new variables ξ and v, the cost functional (4.15) can be written as

$$J(v) = \int_0^\infty \xi^\top(t) Q_e \xi(t) + v^\top(t) R v(t) dt,$$

where

$$Q_e = \begin{pmatrix} Q & -Q \\ -Q & Q \end{pmatrix} \in \mathbb{R}^{2n \times 2n}.$$

The problem is now to find, for every initial state ξ_0 , a piecewise continuous input function v(t) that minimizes this cost functional. This is a so-called a *free endpoint* standard LQ control problem, see [102, pp. 218]. Since the pair (A, B) is stabilizable, the pair (A_e, B_e) is also stabilizable and hence the input function v(t) that minimizes the cost functional J(v) is generated by the feedback law

$$v(t) = -R^{-1}B_e^{+}P_e^{-}\xi(t), \qquad (4.17)$$

where $P_e^- \in \mathbb{R}^{2n \times 2n}$ is the smallest positive semi-definite solution of the Riccati

equation

$$A_e^{\top} P_e^{-} + P_e^{-} A_e - P_e^{-} B_e R^{-1} B_e^{\top} P_e^{-} + Q_e = 0.$$
(4.18)

Now, partition

$$P_e^- := \begin{pmatrix} P_1 & P_{12} \\ P_{12}^\top & P_2 \end{pmatrix},$$

where all blocks have dimension $n \times n$. Recalling (4.16) and (4.17), we then immediately find an expression for the input function u(t) that minimizes the cost functional (4.15) for the system (4.14) and reference signal $r_{ref}(t) = r$.

Theorem 4.2. The input function u(t) that minimizes the cost functional (4.15) is generated by the control law

$$u(t) = K_1 x(t) + K_2 r, (4.19)$$

where $K_1 = -R^{-1}B^{\top}P_1$ and $K_2 = -R^{-1}B^{\top}P_{12}$.

The proof follows immediately from the above considerations. See also [67].

Remark 4.3. Let e(t) := x(t) - r denote the tracking error. Because Q > 0, the control law (4.19) only guarantees that $\bar{e}(t) := e^{-\alpha t}e(t)$ tends to zero as t goes to infinity. Thus, the feedback law that minimizes the LQ tracking cost functional (4.15) only guarantees the actual tracking error e(t) to be exponentially bounded with growth rate $\alpha > 0$. Note that $\alpha > 0$ can be taken arbitrarily small.

It will be shown however that, for the multi-agent system case, the control design method established in this section will, nevertheless, lead to a protocol that achieves consensus.

4.6 Consensus analysis

In this section, we will show that, by adopting the control design method for the multi-agent system (4.2) as proposed in Section 4.4, the resulting distributed control protocol achieves consensus for the entire network.

As already explained in Section 4.4, we choose a sampling period T > 0and introduce a sampling procedure. For each nonnegative integer k, at time t = kT the *i*th agent receives the sampled state value of its neighboring agents (including itself) and minimizes the cost functional (4.13), which is a discounted linear quadratic tracking problem with constant reference signal $r_{ref}(t) = a_i(kT)$ and discount factor $\alpha > 0$.

According to the theory on the discounted LQ tracking problem described in Section 4.5, the local optimal control law for agent *i* at time t = kT over the whole

time horizon $[0,\infty)$ is therefore of the form

$$u_{i,k}(t) = g_{i,k}x_i(t) + g'_{i,k}a_i(kT),$$
(4.20)

in which the control gains $g_{i,k}$ and $g'_{i,k}$ can be computed explicitly by solving the Riccati equation (4.18) associated with the LQ tracking problem for agent *i*.

Lemma 4.4. Consider, at time t = kT, the *i*th agent of the multi-agent system (4.1) with associated local cost functional (4.13). Denote

$$\bar{A} = \begin{pmatrix} -\alpha & 0\\ 0 & -\alpha \end{pmatrix}, \quad \bar{B} = \begin{pmatrix} 1\\ 0 \end{pmatrix}, \quad \bar{Q} = \begin{pmatrix} q & -q\\ -q & q \end{pmatrix}.$$

Let $\bar{P} := \begin{pmatrix} p_1 & p_{12} \\ p_{12} & p_2 \end{pmatrix}$ be the smallest positive semi-definite solution of the Riccati equation

$$\bar{A}^{\top}\bar{P} + \bar{P}\bar{A} - r^{-1}\bar{P}\bar{B}\bar{B}^{\top}\bar{P} + \bar{Q} = 0.$$
(4.21)

Then the local control law (4.20) with $g_{i,k} := -r^{-1}p_1$ and $g'_{i,k} := -r^{-1}p_{12}$ minimizes the cost (4.13) for agent *i*.

Proof. This follows immediately from Theorem 4.2.

Next, agent *i* applies the control law (4.20) only on the time interval [kT, (k + 1)T). Then, at time t = (k + 1)T the above procedure is repeated.

Since, for all i = 1, 2, ..., N and k = 0, 1, ..., the matrices \overline{A} , \overline{B} and \overline{Q} are independent of i and k, the same holds for the gains $g_{i,k}$ and $g'_{i,k}$. In the sequel, we will therefore drop the subscripts in the control gains $g_{i,k}$ and $g'_{i,k}$ and denote them by g and g', respectively. Moreover, using (4.21), we compute $g = r^{-1}(\alpha - \sqrt{\alpha^2 + rq})$ and g' = -g.

By performing this procedure sequentially at each sampling time kT, we then obtain a single control law for agent *i* over the entire interval $[0, \infty)$ as

$$u_{i,k}(t) = gx_i(t) - ga_i(kT), \quad t \in [kT, (k+1)T),$$
(4.22)

where $g = r^{-1}(\alpha - \sqrt{\alpha^2 + rq}) < 0.$

Recall that $\mathbf{a}(t) = G\mathbf{x}(t)$, with *G* given by (4.8), and that $\mathbf{a}(kT) = G\mathbf{x}(kT)$. Therefore, the local control laws for the individual agents lead to a distributed control protocol

$$\mathbf{u}_k(t) = g\mathbf{x}(t) - gG\mathbf{x}(kT), \quad t \in [kT, (k+1)T).$$

$$(4.23)$$

Now, by applying the protocol (4.23) to the multi-agent system (4.1), we find that

the controlled network is represented by

$$\dot{\mathbf{x}}(t) = g\mathbf{x}(t) - gG\mathbf{x}(kT), \quad t \in [kT, (k+1)T).$$
(4.24)

In the remainder of this section, we will analyze this representation, and show that consensus is achieved, i.e. for each initial state $\mathbf{x}(0) = \mathbf{x}_0$ we have $x_i(t) - x_j(t) \rightarrow 0$ as t tends to infinity.

In order to do this, note that the solution of (4.24) with initial state $\mathbf{x}(0) = \mathbf{x}_0$ is given by

$$\mathbf{x}(t) = e^{g(t-kT)}\mathbf{x}(kT) - \int_{kT}^{t} e^{g(t-\tau)}gG \,\mathbf{x}(kT) \,d\tau,$$
(4.25)

for $t \in [kT, (k + 1)T)$, k = 0, 1, 2, ... Obviously, for each initial state \mathbf{x}_0 , the corresponding solution $\mathbf{x}(t)$ is continuous. From (4.25) we see that the sequence of network states $\mathbf{x}(kT)$ evaluated at the discrete time instances kT, k = 0, 1, ... satisfies the difference equation

$$\mathbf{x}((k+1)T) = \Gamma \mathbf{x}(kT), \tag{4.26}$$

 $\Gamma = e^{gT} I_N - (e^{gT} - 1)G \in \mathbb{R}^{N \times N}.$

Clearly, the network reaches consensus if and only if for each \mathbf{x}_0 , $x_i(kT) - x_j(kT) \rightarrow 0$ as *t* tends to infinity.

We proceed with analyzing the eigenvalues of *G*.

Lemma 4.5. The matrix G has an eigenvalue 1 with algebraic multiplicity equal to one and associated eigenvector $\mathbf{1}_N$. The remaining eigenvalues of G are all real and have absolute value strictly less than 1.

Proof. Since L = D - A, we have $G = I_N - (D + I_N)^{-1}L$. Hence we have $\tilde{D}^{\frac{1}{2}}G\tilde{D}^{-\frac{1}{2}} = I_N - \tilde{D}^{-\frac{1}{2}}L\tilde{D}^{-\frac{1}{2}}$ where $\tilde{D} = D + I_N$. Note that the right hand side is symmetric and hence has only real eigenvalues. Thus, by matrix similarity, G also has only real eigenvalues.

Next, we show that *G* has a simple eigenvalue 1 with associated eigenvector $\mathbf{1}_N$. First note that

$$G\mathbf{1}_N = (I_N - (D + I_N)^{-1}L)\mathbf{1}_N = \mathbf{1}_N.$$
(4.27)

Hence, indeed, 1 is an eigenvalue of *G* with eigenvector $\mathbf{1}_N$. Since *G* is similar to a symmetric matrix, it is diagonalizable, so the algebraic multiplicity of its eigenvalue 1 must be equal to its geometric multiplicity. Suppose now that 1 is not a simple eigenvalue. Then there must exist a second eigenvector, say *v*, which is linearly independent of $\mathbf{1}_N$. This implies Gv = v. Then Lv = 0, so *v* must be a multiple of $\mathbf{1}_N$. This is a contradiction. We conclude that the eigenvalue 1 is indeed simple.

Finally, it follows from Gershgorin's Theorem [30] that every eigenvalue λ of G satisfies $-1 < \lambda \leq 1$.

Before we give the main result of this chapter, we first review the following proposition.

Proposition 4.6. Consider the discrete-time system

$$x(k+1) = Ax(k), \quad x(0) = x(0), \quad y(k) = Cx(k)$$

with $A \in \mathbb{R}^{n \times n}$ and $C \in \mathbb{R}^{p \times n}$, where $x(k) \in \mathbb{R}^n$ is the state, x_0 is the initial state and $y(k) \in \mathbb{R}^p$ is the output. Then, $y(k) \to 0$ as $k \to \infty$ for all initial states x_0 if and only if $X_+(A) \subset \ker(C)$. Here, $X_+(A)$ is the unstable subspace, i.e., the sum of the generalized eigenspaces of A associated with its eigenvalues in $\{\lambda \in \mathbb{C} \mid |\lambda| \ge 1\}$.

Proof. A proof can be given by generalizing the results [102, pp. 99] to the discrete time case. \Box

We are now ready to present the main result of this chapter.

Theorem 4.7. Consider the multi-agent system (4.1). Let T > 0 be a sampling period, $\alpha > 0$ a discount factor, and let q, r > 0 be given weights. Let \overline{P} be the smallest positive semi-definite solution of the Riccati equation (4.21) and partition $\overline{P} := \begin{pmatrix} p_1 & p_{12} \\ p_{12} & p_2 \end{pmatrix}$. Then the distributed control protocol (4.23) with $g = -r^{-1}p_1$ and $g' = -r^{-1}p_{12}$ achieves consensus for the controlled network (4.24).

Proof. The network reaches consensus if and only if $L\mathbf{x}(kT) \to 0$ as $k \to \infty$. Since $\ker(L) = \operatorname{im}(\mathbf{1}_N)$, it then follows from Proposition 4.6 that consensus is achieved if and only if $X_+(\Gamma) \subset \ker(L)$, equivalently, the sum of the generalized eigenspaces of Γ corresponding to the eigenvalues λ with $|\lambda| \ge 1$ is equal to $\operatorname{im}(\mathbf{1}_N)$.

Indeed, we will show that all eigenvalues λ of Γ are real and satisfy $-1 < \lambda \leq 1$, and $\lambda = 1$ is a simple eigenvalue with associated eigenvector $\mathbf{1}_N$.

Recall that $\Gamma = e^{gT}I_N - (e^{gT} - 1)G$. Hence, μ is an eigenvalue of Γ if and only if $\mu = e^{gT} - \lambda(e^{gT} - 1)$ where λ is an eigenvalue of G. It was shown in Lemma 4.5 that all eigenvalues λ of G are real and satisfy $-1 < \lambda \leq 1$ and, moreover, $\lambda = 1$ is a simple eigenvalue. Using the fact that g < 0 we thus obtain that the eigenvalues μ of Γ satisfy $-1 < \mu \leq 1$ and $\mu = 1$ is a simple eigenvalue of Γ .

Finally, we will show $\mu = 1$ has eigenvector $\mathbf{1}_N$. Indeed, this follows from $\Gamma \mathbf{1}_N = (e^{gT}I_N - (e^{gT} - 1)G)\mathbf{1}_N = \mathbf{1}_N$. This completes the proof.

Remark 4.8. By analyzing the eigenvalues μ of Γ satisfying $-1 < \mu < 1$, it can be seen that, for given α , the convergence rate of the difference equation (4.26) increases with increasing sampling period *T*. The total time it takes to
reach a disagreement smaller than a given tolerance is then the product of the number of iterations in (4.26) and this sampling period. It might therefore be more advantageous to use a smaller sampling period with a larger number of required iterations, but yet leading to a smaller total time. In other words, the choice of sampling period is a trade-off between the total time required to obtain an acceptable disagreement, and the number of iterations in (4.26).

4.7 Simulation example

In this section, we provide a simple simulation example to illustrate the proposed control design method.

Consider a network of six agents with single integrator dynamics

$$\dot{x}_i(t) = u_i(t), \quad i = 1, 2, \dots, 6,$$

where the initial states are $x_{10} = 1$, $x_{20} = 2$, $x_{30} = -1$, $x_{40} = -2$, $x_{50} = 1$ and $x_{60} = 3$. We assume that the communication among these agents is represented by an undirected cycle graph with six nodes. The associated Laplacian matrix is given by

$$L = \begin{pmatrix} 2 & -1 & 0 & 0 & 0 & -1 \\ -1 & 2 & -1 & 0 & 0 & 0 \\ 0 & -1 & 2 & -1 & 0 & 0 \\ 0 & 0 & -1 & 2 & -1 & 0 \\ 0 & 0 & 0 & -1 & 2 & -1 \\ -1 & 0 & 0 & 0 & -1 & 2 \end{pmatrix}.$$
 (4.28)

First, we take the sampling period to be equal T = 10. On the time interval $t \in [kT, (k + 1)T)$, k = 0, 1, ..., we consider the local cost functional (4.13) for agent *i*. We choose the weights to be q = 2, r = 1 and the discount factor $\alpha = 0.01$. We adopt the control design proposed in Theorem 4.7 and compute the smallest positive semi-definite of the Riccati equation

$$A^{\top}P + PA - r^{-1}PBB^{\top}P + Q = 0$$

with

$$A = \begin{pmatrix} -0.01 & 0\\ 0 & -0.01 \end{pmatrix}, \quad B = \begin{pmatrix} 1\\ 0 \end{pmatrix}, \quad Q = \begin{pmatrix} 2 & -2\\ -2 & 2 \end{pmatrix}.$$

This Riccati equation has a unique positive semi-definite solution which is given by

$$P = \begin{pmatrix} 1.4042 & -1.4042 \\ -1.4042 & 1.4042 \end{pmatrix}.$$



Figure 4.1: Plot of the states of the controlled network with T = 10



Figure 4.2: Plot of the states of the controlled network with T = 0.1

Thus we find the control gains g = -1.4042 and g' = 1.4042. Subsequently, the local control law for agent *i* is given by $u_{i,k}(t) = -1.4042x_i(t) + 1.4042a_i(kT)$ and the dynamics of the controlled agents is given by

$$\dot{x}_i(t) = -1.4042x_i(t) + 1.4042a_i(kT)$$

for $t \in [kT, (k+1)T)$ and i = 1, 2, 3 and k = 0, 1, ... In Figure 4.1 we have plotted the controlled trajectories of the individual agents. It can be seen that the protocol resulting from the local control laws indeed achieves consensus. Note that the states of these agents achieves consensus at around T = 150. The results of a second simulation, this time with sampling period T = 0.1, are plotted in Figure 4.2. By comparing Figure 4.1 and 4.2, it can be seen that the network reaches consensus faster by taking a smaller sampling period.

4.8 Conclusions

We have studied the distributed linear quadratic control problem for a network of agents with single integrator dynamics. We have shown that the computation of control gains that minimize global cost functionals needs global information, in particular knowledge of the initial states of all agents and the Laplacian matrix. We have also shown that this drawback can be overcome by transforming the global cost functional into discounted local cost functionals and assigning each of these to an associated agent. In such a way, each agent computes its own control gain, using sampled information of its neighboring agents. Finally, we have shown that the resulting control protocol achieves consensus for the network.

A suboptimality approach to distributed \mathcal{H}_2 optimal control by state feedback

This chapter deals with the distributed \mathcal{H}_2 optimal control problem for linear multi-agent systems by static relative state feedback. In particular, we consider a *suboptimality* version of this problem. Given a linear multi-agent system with identical agent dynamics, an associated \mathcal{H}_2 cost functional, and a desired upper bound for the cost, our aim is to design a distributed diffusive static protocol such that the protocol achieves state synchronization while the associated cost is smaller than the given upper bound. To that end, we first analyze the \mathcal{H}_2 performance of linear systems and then apply the results to linear multi-agent systems. Based on these results, we provide a design method for computing such a distributed suboptimal protocol. The expression for the local control gain involves a solution to a single Riccati inequality of dimension equal to the dimension of the individual agent dynamics, and the smallest nonzero and the largest eigenvalue of the Laplacian matrix.

5.1 Introduction

The design of distributed protocols for multi-agent systems has received extensive attention in the past decade [85]. This increase in attention is partly due to the broad range of applications of multi-agent systems, e.g., formation control [79], intelligent transportation systems [5], and smart grids [17]. One of the challenging problems in the context of multi-agent systems is to develop distributed diffusive protocols that minimize given cost criteria, while the agents of the network reach a common goal, e.g., state synchronization. The difficulties of designing such distributed diffusive optimal protocols are due to the structural constraints on the communication among these agents, that is, each agent can only receive information from certain other agents. Therefore, in general, distributed optimal control problems are non-convex and difficult to solve.

To overcome this problem, much effort has been devoted to the design of

distributed suboptimal protocols for multi-agent systems. In [7], the authors established a design method to compute distributed suboptimal stabilizing controllers with respect to a global linear quadratic cost functional, which contains terms that penalize the states and inputs of each agent and also the relative states between each agent and its neighboring agents. Later on, an inverse optimal control problem was addressed in [75]. In that paper, the authors showed that there exists a global optimal synchronizing controller if the weighting matrices of the linear quadratic cost functional are chosen to be of a certain special form. For other papers related to distributed linear quadratic optimal control, see also [37, 74, 77].

On the other hand, there has been some work on the design of structured stabilizing controllers for large-scale systems. In [91], the aim was to design decentralized optimal controllers, subject to some constraints on the controller structure, to minimize the closed-loop norm of a feedback system. The authors showed that if the constraints on the controller structure have the property of quadratic invariance, the solution of such problems can be computed efficiently via convex programming. In more recent work, [19] studied the distributed optimal problem for linear discrete-time systems. The authors showed that the problem can be relaxed to a semidefinite program, and a globally optimal distributed controller can be obtained if the semidefinite program relaxation has a rank one solution. In [18], the authors derived a condition under which, given a centralized optimal controller, there exists a distributed suboptimal controller whose state and input trajectories are close to those of the closed-loop system by using this centralized controller.

All the existing work mentioned above deals with finding distributed suboptimal protocols whose performance is very close to being optimal, or to find distributed optimal protocols for certain special cost functionals. In this chapter, however, we want to find a distributed diffusive suboptimal static protocol for linear multi-agent networks such that the associated cost functional is smaller than an, a priori given, desired tolerance (upper bound). We consider a group of identical agents whose dynamics is represented by a finite dimensional linear system. In addition, a connected, simple, undirected weighted graph is given, representing the communication between these agents. Furthermore, we introduce an \mathcal{H}_2 cost functional. Our aim is to design a distributed diffusive static protocol that achieves state synchronization and guarantees the associated cost to be smaller than a given upper bound.

The outline of this chapter is as follows. In Section 5.2, we formulate the distributed \mathcal{H}_2 suboptimal control problem for linear multi-agent systems. We then present the analysis and design of \mathcal{H}_2 suboptimal control laws for general linear systems in Section 5.3, providing necessary results for treating the distributed \mathcal{H}_2 suboptimal control problem. In Section 5.4, we deal with the distributed \mathcal{H}_2 suboptimal control problem. Finally, Section 5.5 concludes this chapter.

5.2 **Problem formulation**

In this chapter, we consider a multi-agent system consisting of N agents with identical dynamics. The interconnection topology among the agents is assumed to be represented by a connected, simple undirected weighted graph with associated Laplacian matrix L. The dynamics of agent i is represented by the following continuous-time linear-time-invariant (LTI) system

$$\dot{x}_i(t) = Ax_i(t) + Bu_i(t) + Ed_i(t), z_i(t) = Cx_i(t) + Du_i(t), \qquad i = 1, 2, \dots, N,$$
(5.1)

where $x_i \in \mathbb{R}^n$, $u_i \in \mathbb{R}^m$, $z_i \in \mathbb{R}^p$ and $d_i \in \mathbb{R}^q$ are the state, the coupling input, the output and the external disturbance input of the *i*th agent, respectively. The matrices A, B, C, D and E have suitable dimensions. We assume that the pair (A, B) is stabilizable. In this chapter, we consider the case that the agents (5.1) are interconnected by means of a distributed diffusive static protocol of the form

$$u_i = K \sum_{j=1}^{N} a_{ij}(x_j - x_i), \quad i = 1, 2, \dots, N,$$
 (5.2)

where $K \in \mathbb{R}^{m \times n}$ is a feedback gain to be designed. The coefficients a_{ij} are the entries of the adjacency matrix \mathcal{A} of the underlying graph.

Denote the aggregate vectors as

$$\begin{split} \mathbf{x} &= (x_1^{\top}, x_2^{\top}, \dots, x_N^{\top})^{\top} \in \mathbb{R}^{nN}, \quad \mathbf{u} = (u_1^{\top}, u_2^{\top}, \dots, u_N^{\top})^{\top} \in \mathbb{R}^{mN}, \\ \mathbf{z} &= (z_1^{\top}, z_2^{\top}, \dots, z_N^{\top})^{\top} \in \mathbb{R}^{pN}, \quad \mathbf{d} = (d_1^{\top}, d_2^{\top}, \dots, d_N^{\top})^{\top} \in \mathbb{R}^{qN}. \end{split}$$

We can then write system (5.1) in compact form as

$$\dot{\mathbf{x}} = (I_N \otimes A)\mathbf{x} + (I_N \otimes B)\mathbf{u} + (I_N \otimes E)\mathbf{d}, \mathbf{z} = (I_N \otimes C)\mathbf{x} + (I_N \otimes D)\mathbf{u},$$
(5.3)

the protocol (5.2) is now of the form

$$\mathbf{u} = (L \otimes K)\mathbf{x}.\tag{5.4}$$

Foremost, we want our protocol to achieve state synchronization for the overall network. This is defined as follows.

Definition 5.1. The protocol (5.4) is said to achieve state synchronization if, whenever the disturbance input is equal to zero, i.e. $\mathbf{d} = 0$, then for all i, j = 1, 2, ..., N we have $x_i(t) - x_j(t) \to 0$ as $t \to \infty$.

The distributed \mathcal{H}_2 optimal control problem is to minimize a given \mathcal{H}_2 cost functional for multi-agent system (5.3) over all protocols (5.4) that achieve state synchronization. Note that in the context of distributed control for multi-agent systems, we are interested in the differences of the state and output values of the agents in the controlled network. Observe also that the differences of the state and output values of communicating agents are captured by the incidence matrix *R* of the underlying graph. Therefore, we define a new output values as

$$\boldsymbol{\zeta} = (W^{\frac{1}{2}}R^{\top} \otimes I_p)\mathbf{z}$$

with $\boldsymbol{\zeta} = (\zeta_1^{\top}, \zeta_2^{\top}, \dots, \zeta_M^{\top})^{\top} \in \mathbb{R}^{pM}$, where *W* is the weight matrix given by (1.2). Thus, the output $\boldsymbol{\zeta}$ reflects the weighted disagreement between the outputs of the agents in accordance with the weights of the edges connecting these agents. Subsequently, we have the following input/state/output model

$$\dot{\mathbf{x}} = (I_N \otimes A)\mathbf{x} + (I_N \otimes B)\mathbf{u} + (I_N \otimes E)\mathbf{d},$$

$$\boldsymbol{\zeta} = (W^{\frac{1}{2}}R^\top \otimes C)\mathbf{x} + (W^{\frac{1}{2}}R^\top \otimes D)\mathbf{u}.$$
(5.5)

Next, by substituting protocol (5.4) into equations (5.5), we obtain the following equations for the controlled network

$$\dot{\mathbf{x}} = (I_N \otimes A + L \otimes BK)\mathbf{x} + (I_N \otimes E)\mathbf{d},$$

$$\boldsymbol{\zeta} = (W^{\frac{1}{2}}R^{\top} \otimes C + W^{\frac{1}{2}}R^{\top}L \otimes DK)\mathbf{x}.$$

Denote $\tilde{A} := I_N \otimes A + L \otimes BK$, $\tilde{E} := I_N \otimes E$, $\tilde{C} := W^{\frac{1}{2}}R^{\top} \otimes C + W^{\frac{1}{2}}R^{\top}L \otimes DK$. The impulse response from the disturbance **d** to the output $\boldsymbol{\zeta}$ is then given by

$$T_K(t) = \tilde{C}e^{\tilde{A}t}\tilde{E}$$

Subsequently, we define the associated H_2 cost functional as

$$J(K) := \int_0^\infty \operatorname{tr}\left[T_K^\top(t)T_K(t)\right] dt.$$
(5.6)

Since the protocol (5.4) has a special form which contains the Kronecker product of the to be designed feedback gain K with the given Laplacian matrix L, the distributed \mathcal{H}_2 optimal control problem is non-convex, and therefore difficult to solve in general. Therefore, instead of trying to solve the distributed \mathcal{H}_2 optimal control problem itself, we will address a suboptimality version of this problem, i.e., we want to design a state synchronizing, distributed diffusive, static protocol such that the associated cost is smaller than an a priori given upper bound. More concretely, the problem we want to address is the following: **Problem 5.1.** Consider multi-agent system (5.3), with interconnection topology among the agents represented by a connected, simple undirected weighted graph with associated Laplacian matrix L, together with cost functional J(K) given by (5.6). Let $\gamma > 0$ be a given tolerance. Our aim is to design a matrix $K \in \mathbb{R}^{n \times m}$ such that the distributed diffusive static protocol $\mathbf{u} = (L \otimes K)\mathbf{x}$ achieves state synchronization and $J(K) < \gamma$.

Before we address Problem 5.1, we will first briefly discuss the H_2 suboptimal control problem for general linear systems, in that way collecting the required preliminary results to treat the actual distributed H_2 suboptimal control problem for multi-agent systems. This will be the subject of the next section.

5.3 *H*₂ suboptimal control for linear systems by static state feedback

In this section, we consider the \mathcal{H}_2 suboptimal control problem for linear systems. We will first analyze the \mathcal{H}_2 performance of a given system with disturbance inputs. Subsequently, we will discuss how to design suboptimal protocols for a linear system with control inputs and disturbance inputs.

5.3.1 H_2 performance analysis for linear systems with disturbance inputs

In this subsection, we will analyze the H_2 performance for linear systems with disturbance inputs. More specifically, we consider the following linear system

$$\dot{x}(t) = \bar{A}x(t) + \bar{E}d(t),$$

$$z(t) = \bar{C}x(t),$$
(5.7)

where $x \in \mathbb{R}^n$ represents the state, $d \in \mathbb{R}^q$ the disturbance input and $z \in \mathbb{R}^p$ the output. The matrices \overline{A} , \overline{C} and \overline{E} have suitable dimensions. The impulse response matrix of system (5.7) from the disturbance d to the output z is $T(t) = \overline{C}e^{\overline{A}t}\overline{E}$. The associated \mathcal{H}_2 performance is given by

$$J = \int_0^\infty \operatorname{tr}\left[T^\top(t)T(t)\right]dt,\tag{5.8}$$

which measures the performance of system (5.7) as the square of the \mathcal{L}_2 -norm of its impulse response matrix. Note that performance (5.8) is finite if the system is internally stable, i.e., \overline{A} is Hurwitz. Our aim is to find conditions such that the performance (5.8) is smaller than a given upper bound. For this, we have the following lemma. See also [124] or [93].

Lemma 5.1. Consider system (5.7) with associated performance (5.8). The performance is finite if \overline{A} is Hurwitz. In that case, we have

$$J = \operatorname{tr}\left(\bar{E}^{\top}Y\bar{E}\right),\tag{5.9}$$

where Y is the unique positive semidefinite solution of

$$\bar{A}^{\top}Y + Y\bar{A} + \bar{C}^{\top}\bar{C} = 0.$$
 (5.10)

Alternatively,

$$J = \inf\{\operatorname{tr}\left(\bar{E}^{\top}P\bar{E}\right) \mid P > 0 \text{ and } \bar{A}^{\top}P + P\bar{A} + \bar{C}^{\top}\bar{C} < 0\}.$$
(5.11)

For a proof of Lemma 5.1, see the proof of Theorem 4.6.2 in [100].

The following lemma now establishes a *necessary* and *sufficient* condition [33], such that the system (5.7) is stable and, for a given upper bound $\gamma > 0$, the performance (5.8) satisfies $J < \gamma$.

Lemma 5.2. Consider system (5.7) with associated performance (5.8). Given $\gamma > 0$. Then \overline{A} is Hurwitz and $J < \gamma$ if and only if there exists a positive definite matrix P satisfying

$$\bar{A}^{\top}P + P\bar{A} + \bar{C}^{\top}\bar{C} < 0, \qquad (5.12)$$

$$\operatorname{tr}\left(\bar{E}^{\top}P\bar{E}\right) < \gamma. \tag{5.13}$$

Proof. (if) Let P > 0 satisfy (5.12). Then $\bar{A}^{\top}P + P\bar{A} < 0$. Note also that P > 0, which implies that \bar{A} is Hurwitz. If P > 0 also satisfies (5.13), then it follows from Lemma 5.1 that $J \leq \operatorname{tr}(\bar{E}^{\top}P\bar{E}) < \gamma$.

(only if) If \bar{A} is Hurwitz and $J < \gamma$, it follows again from Lemma 5.1 that there exists a positive definite solution P to (5.12) and (5.13) such that $J \leq \text{tr} \left(\bar{E}^{\top} P \bar{E} \right) < \gamma$.

5.3.2 \mathcal{H}_2 suboptimal control for linear systems with control and disturbance inputs

In this subsection, we will discuss the H_2 suboptimal control problem for linear systems with control inputs and disturbance inputs. More specifically, we consider the linear system

$$\dot{x}(t) = Ax(t) + Bu(t) + Ed(t),$$

 $z(t) = Cx(t) + Du(t),$
(5.14)

where $x \in \mathbb{R}^n$ represents the state, $u \in \mathbb{R}^m$ the control input, $z \in \mathbb{R}^p$ the output, and $d \in \mathbb{R}^q$ the external disturbance input. The matrices A, B, C, D and E have suitable dimensions. We assume that the pair (A, B) is stabilizable. Using the static state feedback

$$u = Kx \tag{5.15}$$

yields the closed-loop system

$$\dot{x} = (A + BK)x + Ed,$$

$$z = (C + DK)x.$$
(5.16)

We measure the performance of system (5.16) by considering the square of the \mathcal{L}_2 -norm of its impulse response matrix. Therefore, we define the associated \mathcal{H}_2 cost functional as

$$J(K) = \int_0^\infty \operatorname{tr}\left[T_K^{\mathsf{T}}(t)T_K(t)\right]dt,\tag{5.17}$$

where $T_K(t) = (C + DK)e^{(A+BK)t}E$ is the closed-loop impulse response matrix of system (5.16) from the disturbance input *d* to the output *z*. Let $\gamma > 0$ be a given upper bound for the cost J(K). We are interested in finding a static state feedback of the form (5.15) such that A + BK is Hurwitz and the associated cost is smaller than the given upper bound γ , i.e. $J(K) < \gamma$.

The following lemma yields a sufficient condition for the existence of such a static state feedback and how to compute one.

Lemma 5.3. Consider system (5.14) with associated cost functional (5.17). Let $\gamma > 0$. Assume that the pair (A, B) is stabilizable. Assume that $D^{\top}C = 0$ and $D^{\top}D = I_m$. Suppose that there exists a positive definite matrix P satisfying

$$A^{\top}P + PA - PBB^{\top}P + C^{\top}C < 0, \tag{5.18}$$

$$\operatorname{tr}\left(E^{\top}PE\right) < \gamma. \tag{5.19}$$

Let $K = -B^{\top}P$. Then A + BK is Hurwitz and $J(K) < \gamma$.

Proof. Substituting $K = -B^{\top}P$ into system (5.16) gives us

$$\dot{x} = (A - BB^{\top}P)x + Ed,$$

$$z = (C - DB^{\top}P)x.$$

Since $D^{\top}C = 0$ and $D^{\top}D = I_m$, inequality (5.18) is equivalent to

$$(A - BB^{\top}P)^{\top}P + P(A - BB^{\top}P) + (C - DB^{\top}P)^{\top}(C - DB^{\top}P) < 0.$$
(5.20)

Since P > 0 is a solution of (5.18), it also satisfies (5.20), which implies that $A - BB^{\top}P$ is Hurwitz. Since (5.19) also holds, by taking $\bar{A} = A - BB^{\top}P$, $\bar{C} = C - DB^{\top}P$ and $\bar{E} = E$, it immediately follows from Theorem 5.2 that $J(K) < \gamma$.

5.4 Distributed H_2 suboptimal protocols by static state feedback

In the previous section, we have dealt with the \mathcal{H}_2 suboptimal control problem for linear systems, collecting the necessary results for treating the distributed \mathcal{H}_2 suboptimal control problem. In the present section, we will deal with the distributed \mathcal{H}_2 suboptimal control problem for multi-agent networks with identical linear agent dynamics.

As has already been shown in Section 5.2, the dynamics of the multi-agent network we consider is given by

$$\dot{\mathbf{x}} = (I_N \otimes A + L \otimes BK)\mathbf{x} + (I_N \otimes E)\mathbf{d}, \boldsymbol{\zeta} = (W^{\frac{1}{2}}R^\top \otimes C + W^{\frac{1}{2}}R^\top L \otimes DK)\mathbf{x}.$$
(5.21)

For convenience, we also repeat here the associated \mathcal{H}_2 cost functional

$$J(K) = \int_0^\infty \operatorname{tr} \left[T_K^\top(t) T_K(t) \right] dt, \qquad (5.22)$$

where $T_K(t) = \tilde{C}e^{\tilde{A}t}\tilde{E}$ is the impulse response matrix from the disturbance input **d** to the output $\boldsymbol{\zeta}$ with $\tilde{A} := I_N \otimes A + L \otimes BK$, $\tilde{E} := I_N \otimes E$ and $\tilde{C} := W^{\frac{1}{2}}R^{\top} \otimes C + W^{\frac{1}{2}}R^{\top}L \otimes DK$.

The distributed \mathcal{H}_2 suboptimal control problem is to find a distributed diffusive static protocol (5.4) with gain matrix K that achieves state synchronization and such that the associated cost (5.22) is smaller than a given upper bound $\gamma > 0$, i.e. $J(K) < \gamma$. We further assume that $D^{\top}C = 0$ and $D^{\top}D = I_m$, i.e. we assume that the distributed \mathcal{H}_2 suboptimal control problem is in *standard form*.

Next, we will first apply the state transformation

$$\bar{\mathbf{x}} = (U^{\top} \otimes I_n)\mathbf{x}$$

where the orthogonal matrix U is defined in (1.1). After this state transformation, the equations of the controlled network become

$$\dot{\bar{\mathbf{x}}} = (I_N \otimes A + \Lambda \otimes BK)\bar{\mathbf{x}} + (U^\top \otimes E)\mathbf{d},$$
$$\boldsymbol{\zeta} = (W^{\frac{1}{2}}R^\top U \otimes C + W^{\frac{1}{2}}R^\top LU \otimes DK)\bar{\mathbf{x}},$$

and our cost functional is equal to

$$J(K) = \int_0^\infty \operatorname{tr}\left[T_K^\top(t)T_K(t)\right]dt,\tag{5.23}$$

where

$$T_K(t) = C_o e^{A_o t} E_o \tag{5.24}$$

is the impulse response matrix from the disturbance input **d** to the output ζ with $A_o := I_N \otimes A + \Lambda \otimes BK$, $C_o := W^{\frac{1}{2}}R^{\top}U \otimes C + W^{\frac{1}{2}}R^{\top}LU \otimes DK$ and $E_o := U^{\top} \otimes E$. Note that, by applying the state transformation, only the system model has changed while the impulse response and the associated cost remain the same.

In order to proceed, we introduce the following N - 1 auxiliary linear systems

$$\begin{aligned} \xi_i &= A\xi_i + \lambda_i Bv_i + E\delta_i, \\ \eta_i &= \sqrt{\lambda_i} C\xi_i + \lambda_i \sqrt{\lambda_i} Dv_i, \end{aligned}$$
 (5.25)

where λ_i , i = 2, 3, ..., N are the nonzero eigenvalues of the graph Laplacian *L*. Using in all systems (5.25) the identical static state feedback

$$v_i = K\xi_i, \quad i = 2, 3, \dots, N$$
 (5.26)

yields the closed-loop systems

$$\dot{\xi}_i = (A + \lambda_i BK)\xi_i + E\delta_i,
\eta_i = (\sqrt{\lambda_i}C + \lambda_i\sqrt{\lambda_i}DK)\xi_i, \qquad i = 2, 3, \dots, N.$$
(5.27)

We further introduce the associated \mathcal{H}_2 cost functionals

$$J_i(K) = \int_0^\infty \operatorname{tr} \left[T_{i,K}^\top(t) T_{i,K}(t) \right] dt, \quad i = 2, 3, \dots, N,$$
(5.28)

where

$$T_{i,K} = (\sqrt{\lambda_i}C + \lambda_i\sqrt{\lambda_i}DK)e^{(A+\lambda_iBK)t}E,$$
(5.29)

for i = 2, 3, ..., N, are the closed-loop impulse response matrices from the disturbance δ_i to the output η_i , respectively.

It turns out that our original cost functional can be expressed as the sum of the cost functionals associated with the auxiliary systems (5.25). In fact, the following theorem holds.

Theorem 5.4. Consider the network (5.21) with associated cost (5.22) and the systems (5.27) with associated costs (5.28) for i = 2, 3, ..., N, respectively. Then the protocol (5.4) achieves state synchronization for the network (5.21) if and only if the static state feedback

(5.26) internally stabilizes all systems (5.25). Moreover,

$$J(K) = \sum_{i=2}^{N} J_i(K).$$
 (5.30)

Proof. It is a standard result that the protocol (5.4) achieves state synchronization for the network (5.21) if and only if the static state feedback (5.26) internally stabilizes all systems (5.25). See e.g. [54] or [103].

We now prove (5.30). Let *K* be such that synchronization is achieved. Then we have

$$J(K) = \int_0^\infty \operatorname{tr}\left(\bar{E}_e^\top e^{\bar{A}_e^\top t} \bar{C}_e^\top \bar{C}_e e^{\bar{A}_e t} \bar{E}_e\right) dt$$

with

$$\bar{A}_e = I_N \otimes A + \Lambda \otimes BK, \quad \bar{C}_e = W^{\frac{1}{2}} R^\top U \otimes C + W^{\frac{1}{2}} R^\top L U \otimes DK, \quad \bar{E}_e = U^\top \otimes E.$$

Since $U^{\top}LU = \Lambda$, $L = RWR^{\top}$ and $D^{\top}C = 0$, we have $\bar{C}_e^{\top}\bar{C}_e = \tilde{C}_e^{\top}\tilde{C}_e$ with

$$\tilde{C}_e := \Lambda^{\frac{1}{2}} \otimes C + \Lambda^{\frac{3}{2}} \otimes DK.$$

We also have $\overline{E}_e \overline{E}_e^{\top} = \widetilde{E}_e \widetilde{E}_e^{\top}$ with $\widetilde{E}_e := I_N \otimes E$. Thus we find that

$$\operatorname{tr}\left(\bar{E}_{e}^{\top}e^{\bar{A}_{e}^{\top}t}\bar{C}_{e}^{\top}\bar{C}_{e}e^{\bar{A}_{e}t}\bar{E}_{e}\right) = \operatorname{tr}\left(\tilde{E}_{e}^{\top}e^{\bar{A}_{e}^{\top}t}\tilde{C}_{e}^{\top}\tilde{C}_{e}e^{\bar{A}_{e}t}\tilde{E}_{e}\right).$$
(5.31)

We now analyze the matrix function $\tilde{C}_e e^{\bar{A}_e t} \tilde{E}_e$ appearing in (5.31). It is straightforward to show that

$$\tilde{C}_e e^{\bar{A}_e t} \tilde{E}_e = \text{blockdiag} \left(0, C_2 e^{A_2 t} E, \dots, C_N e^{A_N t} E \right),$$

where $A_i = A + \lambda_i BK$ and $C_i = \sqrt{\lambda_i}C + \lambda_i \sqrt{\lambda_i}DK$. Thus we find that

$$J(K) = \int_0^\infty \sum_{i=2}^N \operatorname{tr} \left[T_{i,K}^\top(t) T_{i,K}(t) \right] dt.$$

The claim (5.30) then follows immediately.

Based on Theorem 5.4, we have transformed the problem of distributed \mathcal{H}_2 suboptimal control for the network (5.21) into \mathcal{H}_2 suboptimal control problems for N - 1 linear systems (5.27) with the same feedback gain K. Next, we want to establish conditions under which all N - 1 systems (5.27) are internally stable and the state feedback (5.26) is suboptimal.

The following lemma yields a necessary and sufficient condition for a given

gain matrix $K \in \mathbb{R}^{m \times n}$ such that all systems (5.27) are internally stable and $\sum_{i=2}^{N} J_i(K) < \gamma$.

Lemma 5.5. Consider the closed-loop systems (5.27) with the associated cost functionals (5.28). Let $\gamma > 0$ be a given tolerance. The static state feedback (5.26) internally stabilizes all systems and $\sum_{i=2}^{N} J_i(K) < \gamma$ if and only if there exist positive definite matrices $P_i, i = 2, 3, ..., N$ satisfying

$$(A + \lambda_i BK)^{\top} P_i + P_i (A + \lambda_i BK) + (\sqrt{\lambda_i} C + \lambda_i \sqrt{\lambda_i} DK)^{\top} (\sqrt{\lambda_i} C + \lambda_i \sqrt{\lambda_i} DK) < 0,$$
(5.32)

$$\sum_{i=2}^{N} \operatorname{tr}\left(E^{\top} P_{i} E\right) < \gamma.$$
 (5.33)

Proof. (if) Since (5.33) holds, there exist sufficiently small $\epsilon_i > 0, i = 2, ..., N$ such that $\sum_{i=2}^{N} \gamma_i < \gamma$ where $\gamma_i := \text{tr}(E^{\top}P_iE) + \epsilon_i$. Because there exists P_i such that (5.32) and tr $(E^{\top}P_iE) < \gamma_i$ hold for all i = 2, ..., N, by taking $\overline{A} = A + \lambda_i BK$ and $\overline{C} = \sqrt{\lambda_i}C + \lambda_i\sqrt{\lambda_i}DK$, it follows from Theorem 5.2 that all systems (5.27) are internally stable and $J_i(K) < \gamma_i$ for i = 2, ..., N. Therefore, $\sum_{i=2}^{N} J_i(K) < \gamma$.

(only if) Since $\sum_{i=2}^{N} J_i(K) < \gamma$, there exist sufficiently small $\epsilon_i > 0, i = 2, ..., N$ such that $\sum_{i=2}^{N} \gamma_i < \gamma$ where $\gamma_i := J_i(K) + \epsilon_i$. Because all systems (5.27) are internally stable and $J_i(K) < \gamma_i$ for i = 2, ..., N, by taking $\overline{A} = A + \lambda_i BK$ and $\overline{C} = \sqrt{\lambda_i C} + \lambda_i \sqrt{\lambda_i} DK$, it follows from Theorem 5.2 that there exist positive semidefinite matrices P_i such that (5.32) and tr $(E^{\top} P_i E) < \gamma_i$ hold for all i = 2, ..., N. Since $\sum_{i=2}^{N} \gamma_i < \gamma$, this implies that $\sum_{i=2}^{N} \operatorname{tr} (E^{\top} P_i E) < \gamma$.

Lemma 5.5 establishes a necessary and sufficient condition for a given gain matrix K to internally stabilize all closed-loop systems (5.27) and to achieve $\sum_{i=2}^{N} J_i(K) < \gamma$. However, it does yet not provide a method to compute such gain matrix K. To this end, in the following theorem, we will provide a design method for computing such a gain matrix K and, correspondingly, a distributed suboptimal protocol for multi-agent system (5.1) together with cost functional (5.22).

Theorem 5.6. Consider multi-agent system (5.1) with the associated cost functional (5.22). Let $\gamma > 0$ be a given tolerance. Let c be any real number such that $0 < c < \frac{2}{\lambda_N^2}$. We distinguish two cases:

(*i*) *if*

$$0 < c \leqslant \frac{2}{\lambda_2^2 + \lambda_2 \lambda_N + \lambda_N^2},\tag{5.34}$$

then there exists a positive definite matrix P satisfying

$$A^{\top}P + PA + (c^2\lambda_2^3 - 2c\lambda_2)PBB^{\top}P + \lambda_N C^{\top}C < 0.$$
(5.35)

(ii) if

$$\frac{2}{\lambda_2^2 + \lambda_2 \lambda_N + \lambda_N^2} < c < \frac{2}{\lambda_N^2},\tag{5.36}$$

then there exists a positive definite matrix P satisfying

$$A^{\top}P + PA + (c^2\lambda_N^3 - 2c\lambda_N)PBB^{\top}P + \lambda_N C^{\top}C < 0.$$
(5.37)

In both cases, if in addition P satisfies

$$\operatorname{tr}\left(\boldsymbol{E}^{\top}\boldsymbol{P}\boldsymbol{E}\right) < \frac{\gamma}{N-1}.$$
(5.38)

Then protocol (5.4) with $K := -cB^{\top}P$ achieves synchronization, and it is suboptimal, *i.e.* $J(K) < \gamma$.

Proof. We will only provide the proof for case (i) above. Using the upper and lower bound on *c* given by (5.34), it can be verified that $c^2 \lambda_i^3 - 2c\lambda_i \leq c^2 \lambda_2^3 - 2c\lambda_2 < 0$ for i = 2, 3, ..., N. Since also $\lambda_i \leq \lambda_N$, one can see that the positive definite solution *P* of (5.35) also satisfies the N - 1 Riccati inequalities

$$A^{\top}P + PA + (c^2\lambda_i^3 - 2c\lambda_i)PBB^{\top}P + \lambda_i C^{\top}C < 0$$
(5.39)

for i = 2, ..., N. Equivalently, P also satisfies the Lyapunov inequalities

$$(A - c\lambda_i BB^{\top}P)^{\top}P + P(A - c\lambda_i BB^{\top}P) + c^2\lambda_i^3 PBB^{\top}P + \lambda_i C^{\top}C < 0,$$

for i = 2, ..., N. Taking $P_i = P$ for i = 2, 3, ..., N and $K := -cB^{\top}P$ in inequalities (5.32) and (5.33) immediately gives us inequalities (5.39) and

$$\sum_{i=2}^{N} \operatorname{tr}\left(E^{\top} P E \right) < \gamma.$$

Then it follows from Lemma 5.5 that all systems (5.27) are internally stable and $\sum_{i=2}^{N} J_i(K) < \gamma$. Furthermore, it follows from Theorem 5.4 that the protocol (5.4) achieves state synchronization for the network (5.21) and $J(K) < \gamma$.

Remark 5.7. Theorem 5.6 states that by choosing *c* satisfying (5.34) for case (i) and P > 0 satisfying (5.35), the distributed static protocol with local gain $K = -cB^{\top}P$ is suboptimal if *P* also satisfies (5.38). Then the question arises: how should we

choose c and P such that the local gain of the suboptimal protocol is 'best' in the sense that we have tr $(E^{\top}PE)$ and, consequently, J(K) as small as possible? Since smaller P leads to smaller tr $(E^{\top}PE)$ and, consequently, smaller J(K), we should therefore try to find P as small as possible. In fact, one can find a positive definite solution $P = P(c, \epsilon)$ to (5.35) by solving

$$A^{\top}P + PA - PBR(c)^{-1}B^{\top}P + Q(\epsilon) = 0$$

with $R(c) = \frac{1}{-c^2\lambda_2^3 + 2c\lambda_2}I_n$ and $Q(\epsilon) = \lambda_N C^\top C + \epsilon I_n$ where c is chosen as in (5.34) and $\epsilon > 0$ arbitrary. If c_1 and c_2 as in (5.34) satisfy $c_1 \leq c_2$, then we have $R(c_1) \leq R(c_2)$, so, clearly, $P(c_1,\epsilon) \leq P(c_2,\epsilon)$. Similarly, if $0 < \epsilon_1 \leq \epsilon_2$, we immediately have $Q(\epsilon_1) \leq Q(\epsilon_2)$. Again, it follows that $P(c,\epsilon_1) \leq P(c,\epsilon_2)$. Therefore, if we choose $\epsilon > 0$ very close to 0 and $c = \frac{2}{\lambda_2^2 + \lambda_2 \lambda_N + \lambda_N^2}$, we find the 'best' solution to the Riccati inequality (5.37) in the sense explained above.

Likewise, if we choose *c* satisfying (5.36) corresponding to case (ii), it can be shown that if we choose $\epsilon > 0$ very close to 0 and c > 0 very close to $\frac{2}{\lambda_2^2 + \lambda_2 \lambda_N + \lambda_N^2}$, we find the 'best' solution to the Riccati inequality (5.37).

5.5 Conclusions

In this chapter, we have studied a distributed \mathcal{H}_2 suboptimal control problem for linear multi-agent systems with connected, simple undirected weighted graph. Given a multi-agent system with identical agent dynamics, an associated global \mathcal{H}_2 cost functional and also a desired upper bound for the cost, we have provided a design method for computing a distributed suboptimal protocol such that the protocol achieves state synchronization and the associated cost is smaller than the given upper bound. The expression for the local control gain is provided in terms of solutions of a single Riccati inequality, whose dimension is equal to the dimension of the individual agent dynamics, and also involves the largest and the smallest nonzero eigenvalue of the Laplacian matrix.

6 Distributed \mathcal{H}_2 suboptimal control by dynamic output feedback

This chapter deals with distributed \mathcal{H}_2 suboptimal control by dynamic relative output feedback for homogeneous linear multi-agent systems. Given a linear multi-agent system, together with an associated \mathcal{H}_2 cost functional, the objective is to design dynamic output feedback protocols that guarantee the associated cost to be smaller than an a priori given upper bound while synchronizing the controlled network. A design method is provided to compute such protocols. The computation of the two local gains in these protocols involves two Riccati inequalities, each of dimension equal to the dimension of the state space of the agents. The largest and smallest nonzero eigenvalue of the Laplacian matrix of the network graph are also used in the computation of one of the two local gains. A simulation example is provided to illustrate the performance of the proposed protocols.

6.1 Introduction

The design of distributed protocols for networked multi-agent systems has been one of the most active research topics in the field of systems and control over the last two decades, see e.g. [11, 85]. This is partly due to the broad range of applications of multi-agent systems, e.g. smart grids [17], formation control [79, 118], and intelligent transportation systems [4]. One of the challenging problems in the context of linear multi-agent systems is the problem of developing distributed protocols to minimize given quadratic cost criteria while the agents reach a common goal, e.g., synchronization. Due to the structural constraints that are imposed on the control laws by the communication topology, such optimal control problems are difficult to solve. These structural constraints make distributed optimal control problems non-convex, and it is unclear under what conditions solutions exist in general.

In the existing literature, many efforts have been devoted to addressing distributed linear quadratic optimal control problems. In [7], distributed suboptimal stabilizing controllers were computed to stabilize multi-agent networks with identical agent dynamics subject to a global linear quadratic cost functional. For a network of agents with single integrator dynamics, an explicit expression for the optimal gain was given in [10], see also [38]. In [75] and [123], a distributed linear quadratic control problem was dealt with using an inverse optimality approach. This approach was further employed in [77] to design reduced order controllers. Recently, also in [37], the distributed LQ suboptimal control problem was considered. In parallel to the above, much work has been put into the problem of distributed \mathcal{H}_2 optimal control. Given a particular global \mathcal{H}_2 cost functional, [55] and [53] proposed distributed suboptimal stabilizing protocols involving static state feedback for multi-agent systems with undirected graphs. Later on, in [112] these results were generalized to directed graphs. For a given \mathcal{H}_2 cost criterion that penalizes the weighted differences between the outputs of the communicating agents, in [36] a distributed suboptimal synchronizing protocol based on static relative state feedback was established.

In the past, also the design of structured controllers for large-scale systems has attracted much attention. In [91], the notion of quadratic invariance was adopted to develop decentralized controllers that minimize the performance of the feedback system with constraints on the controller structure. In [57], the so called alternating direction method of multipliers was adopted to design sparse feedback gains that minimize an \mathcal{H}_2 performance. In [18], conditions were provided under which, for a given optimal centralized controller, a distributed suboptimal controller exists so that the resulting closed loop state and input trajectories are close in a certain sense.

The distributed \mathcal{H}_2 optimal control problem for multi-agent systems by dynamic relative output feedback is to find an optimal distributed dynamic protocol that achieves synchronization for the controlled network and that minimizes the \mathcal{H}_2 cost functional. This problem, however, is a non-convex optimization problem, and therefore it is unclear whether such optimal protocol exists, or whether a closed form solution can be given. Therefore, in this chapter, we look at an alternative version of this problem that requires only *suboptimality*. More precisely, we extend our preliminary results in Chapter 5 on static relative state feedback to the general case of dynamic protocols using relative measurement outputs. The main contributions of this chapter are the following.

- 1. We solve the open problem of finding, for a single continuous-time linear system, a separation principle based \mathcal{H}_2 suboptimal dynamic output feedback controller. This result extends the recent result in [25] on the separation principle in \mathcal{H}_2 suboptimal control for discrete-time systems.
- 2. Based on the above result, we provide a method for computing distributed \mathcal{H}_2 suboptimal dynamic output feedback protocols for linear multi-agent

systems.

The outline of this chapter is as follows. In Section 6.2, we formulate the distributed \mathcal{H}_2 suboptimal control problem by dynamic relative output feedback for linear multi-agent systems. In order to solve this problem, in Section 6.3, we first study \mathcal{H}_2 suboptimal control by dynamic output feedback for a single linear system. In Section 6.4 we then treat the problem introduced in Section 6.2. To illustrate our method, a simulation example is provided in Section 6.5. Finally, Section 6.6 concludes this chapter.

6.2 **Problem formulation**

In this chapter, we consider a homogeneous multi-agent system consisting of N identical agents, where the underlying network graph is a connected, simple undirected weighted graph with associated adjacency matrix A and Laplacian matrix L. The dynamics of the *i*th agent is represented by a finite-dimensional linear time-invariant system

$$\dot{x}_{i} = Ax_{i} + Bu_{i} + Ed_{i},
y_{i} = C_{1}x_{i} + D_{1}d_{i}, \qquad i = 1, 2, \dots, N,
z_{i} = C_{2}x_{i} + D_{2}u_{i},$$
(6.1)

where $x_i \in \mathbb{R}^n$ is the state, $u_i \in \mathbb{R}^m$ is the coupling input, $d_i \in \mathbb{R}^q$ is an unknown external disturbance, $y_i \in \mathbb{R}^r$ is the measured output and $z_i \in \mathbb{R}^p$ is the output to be controlled. The matrices A, B, C_1 , D_1 , C_2 , D_2 and E are of compatible dimensions. Throughout this chapter we assume that the pair (A, B) is stabilizable and the pair (C_1, A) is detectable. The agents (6.1) are to be interconnected by means of a dynamic output feedback protocol. Following [103] and [121], we consider observer based dynamic protocols of the form

$$\dot{w}_{i} = Aw_{i} + B\sum_{j=1}^{N} a_{ij}(u_{i} - u_{j}) + G\left(\sum_{j=1}^{N} a_{ij}(y_{i} - y_{j}) - C_{1}w_{i}\right), \qquad (6.2)$$
$$u_{i} = Fw_{i}, \quad i = 1, 2, \dots, N,$$

where $G \in \mathbb{R}^{n \times r}$ and $F \in \mathbb{R}^{m \times n}$ are local gains to be designed. The coefficients a_{ij} are the entries of the adjacency matrix \mathcal{A} of the underlying network graph. We briefly explain the structure of this protocol. Each local controller of the protocol (6.2) observes the weighted sum of the relative input signals $\sum_{j=1}^{N} a_{ij}(u_i - u_j)$ and the weighted sum of the disagreements between the measured output signals $\sum_{j=1}^{N} a_{ij}(y_i - y_j)$. The first equation in (6.2) in fact represents an asymptotic

observer for the weighted sum of the relative states of agent *i*, and the state of this observer is an estimate of this value. Note that, for the error $e_i := w_i - \sum_{j=1}^{N} a_{ij}(x_i - x_j)$, the error dynamics is $\dot{e}_i = (A - GC_1)e_i + \sum_{j=1}^{N} a_{ij}(GD_1 - E)(d_i - d_j)$. If the disturbance inputs to all the agents are zero, i.e. $d_i = 0$ for i = 1, 2, ..., N, then the error dynamics is asymptotically stable if and only if $A - GC_1$ is Hurwitz. An estimate of the weighted sum of the relative states of each agent is then fed back to this agent using a static gain.

Denote by $\mathbf{x} = (x_1^{\top}, x_2^{\top}, \dots, x_N^{\top})^{\top}$ the aggregate state vector and likewise define **u**, **y**, **z**, **d** and **w**. The multi-agent system (6.1) can then be written in compact form as

$$\mathbf{x} = (I_N \otimes A)\mathbf{x} + (I_N \otimes B)\mathbf{u} + (I_N \otimes E)\mathbf{d},$$

$$\mathbf{y} = (I_N \otimes C_1)\mathbf{x} + (I_N \otimes D_1)\mathbf{d},$$

$$\mathbf{z} = (I_N \otimes C_2)\mathbf{x} + (I_N \otimes D_2)\mathbf{u},$$

(6.3)

and the dynamic protocol (6.2) is represented by

$$\dot{\mathbf{w}} = (I_N \otimes (A - GC_1) + L \otimes BF) \, \mathbf{w} + (L \otimes G) \mathbf{y}, \mathbf{u} = (I_N \otimes F) \mathbf{w}.$$
(6.4)

By interconnecting the network (6.3) using the dynamic protocol (6.4), we obtain the controlled network

$$\begin{pmatrix} \dot{\mathbf{x}} \\ \dot{\mathbf{w}} \end{pmatrix} = \begin{pmatrix} I_N \otimes A & I_N \otimes BF \\ L \otimes GC_1 & I_N \otimes (A - GC_1) + L \otimes BF \end{pmatrix} \begin{pmatrix} \mathbf{x} \\ \mathbf{w} \end{pmatrix} + \begin{pmatrix} I_N \otimes E \\ L \otimes GD_1 \end{pmatrix} \mathbf{d},$$

$$\mathbf{z} = \begin{pmatrix} I_N \otimes C_2 & I_N \otimes D_2F \end{pmatrix} \begin{pmatrix} \mathbf{x} \\ \mathbf{w} \end{pmatrix}.$$

$$(6.5)$$

Foremost, we want the dynamic protocol (6.4) to achieve synchronization for the network.

Definition 6.1. The protocol (6.4) is said to synchronize the network if, whenever the external disturbances of all agents are equal to zero, i.e. $\mathbf{d} = 0$, we have $x_i(t) - x_j(t) \to 0$ and $w_i(t) - w_j(t) \to 0$ as $t \to \infty$, for all i, j = 1, 2, ..., N.

The distributed \mathcal{H}_2 optimal control problem by dynamic output feedback is to minimize a given global \mathcal{H}_2 cost functional over all dynamic protocols of the form (6.4) that achieve synchronization for the controlled network. In the context of distributed control for multi-agent systems, we are interested in the differences of the state and output values of the agents in the controlled network, see e.g. [45] or [69]. Note that these differences are captured by the incidence matrix R of the underlying graph. Therefore, we introduce a new output variable as

$$\boldsymbol{\zeta} = (W^{\frac{1}{2}}R^{\top} \otimes I_p)\mathbf{z} \tag{6.6}$$

with $\boldsymbol{\zeta} = (\zeta_1^{\top}, \zeta_2^{\top}, \dots, \zeta_M^{\top})^{\top} \in \mathbb{R}^{pM}$, where *W* is the weight matrix of the underlying graph, as defined in (1.2). Thus, the output $\boldsymbol{\zeta}$ is the vector of weighted disagreements between the outputs of the agents, in which the weights are given by the square roots of the edge weights connecting these agents. Subsequently, we consider the network (6.5) with this new output:

$$\begin{pmatrix} \dot{\mathbf{x}} \\ \dot{\mathbf{w}} \end{pmatrix} = \begin{pmatrix} I_N \otimes A & I_N \otimes BF \\ L \otimes GC_1 & I_N \otimes (A - GC_1) + L \otimes BF \end{pmatrix} \begin{pmatrix} \mathbf{x} \\ \mathbf{w} \end{pmatrix} + \begin{pmatrix} I_N \otimes E \\ L \otimes GD_1 \end{pmatrix} \mathbf{d},$$

$$\boldsymbol{\zeta} = \begin{pmatrix} W^{\frac{1}{2}} R^\top \otimes C_2 & W^{\frac{1}{2}} R^\top \otimes D_2F \end{pmatrix} \begin{pmatrix} \mathbf{x} \\ \mathbf{w} \end{pmatrix}.$$
(6.7)

Denote

$$A_e = \begin{pmatrix} I_N \otimes A & I_N \otimes BF \\ L \otimes GC_1 & I_N \otimes (A - GC_1) + L \otimes BF \end{pmatrix},$$

$$C_e = \begin{pmatrix} W^{\frac{1}{2}}R^{\top} \otimes C_2 & W^{\frac{1}{2}}R^{\top} \otimes D_2F \end{pmatrix},$$

$$E_e = \begin{pmatrix} I_N \otimes E \\ L \otimes GD_1 \end{pmatrix}.$$

The impulse response matrix from the external disturbance **d** to the output ζ is then equal to

$$T_{F,G}(t) = C_e e^{A_e t} E_e. (6.8)$$

Next, the associated global \mathcal{H}_2 cost functional is defined to be the squared \mathcal{L}_2 -norm of the closed loop impulse response, and is given by

$$J(F,G) := \int_0^\infty \operatorname{tr} \left[T_{F,G}^\top(t) T_{F,G}(t) \right] dt.$$
(6.9)

The distributed \mathcal{H}_2 optimal control problem by dynamic output feedback is the problem of minimizing (6.9) over all dynamic protocols of the form (6.4) that achieve synchronization for the network. Unfortunately, due to the particular form of the protocol (6.4), this optimization problem is, in general, non-convex and difficult to solve, and a closed form solution has not been provided in the literature up to now. Therefore, instead of trying to find an optimal solution, in this chapter we will address a suboptimality version of the problem. More specifically, we will design synchronizing dynamic protocols (6.4) that guarantee the associated cost (6.9) to be smaller than an a priori given upper bound. More concretely, the problem that we will address is the following:

Problem 6.1. Let $\gamma > 0$ be a given tolerance. Design local gains $F \in \mathbb{R}^{m \times n}$ and $G \in \mathbb{R}^{n \times r}$ such that the dynamic protocol (6.4) achieves $J(F, G) < \gamma$ and synchronizes the network.

Before we address Problem 6.1, we will first study the \mathcal{H}_2 suboptimal control problem by dynamic output feedback for a single linear system. In that way, we will collect the required preliminary results to treat the actual distributed \mathcal{H}_2 suboptimal control problem for multi-agent systems.

6.3 *H*₂ suboptimal control for linear systems by dynamic output feedback

In this section, we will discuss the \mathcal{H}_2 suboptimal control problem by dynamic output feedback for a single linear system. This problem has been dealt with before, see e.g. [95], [96], [100] or [25]. In particular, in [25], the *separation principle* for \mathcal{H}_2 suboptimal control for discrete-time linear systems was established. Here, we will establish the analogue of that result for the continuous-time case.

Consider the linear system

$$\begin{aligned} \dot{x} &= \bar{A}x + \bar{B}u + \bar{E}d, \\ y &= \bar{C}_1 x + \bar{D}_1 d, \\ z &= \bar{C}_2 x + \bar{D}_2 u, \end{aligned} \tag{6.10}$$

where $x \in \mathbb{R}^n$ is the state, $u \in \mathbb{R}^m$ the control input, $d \in \mathbb{R}^q$ an unknown external disturbance, $y \in \mathbb{R}^r$ the measured output, and $z \in \mathbb{R}^p$ the output to be controlled. The matrices \overline{A} , \overline{B} , \overline{C}_1 , \overline{D}_1 , \overline{C}_2 , \overline{D}_2 and \overline{E} have compatible dimensions. In this section, we assume that the pair $(\overline{A}, \overline{B})$ is stabilizable and that the pair $(\overline{C}_1, \overline{A})$ is detectable. Moreover, we consider dynamic output feedback controllers of the form

$$\dot{w} = \bar{A}w + \bar{B}u + G\left(y - \bar{C}_1w\right),$$

$$u = Fw.$$
(6.11)

where $w \in \mathbb{R}^n$ is the state of the controller, and $F \in \mathbb{R}^{m \times n}$ and $G \in \mathbb{R}^{n \times r}$ are gain matrices to be designed. By interconnecting the controller (6.11) and the system (6.10), we obtain the controlled system

$$\begin{pmatrix} \dot{x} \\ \dot{w} \end{pmatrix} = \begin{pmatrix} \bar{A} & \bar{B}F \\ G\bar{C}_1 & \bar{A} + \bar{B}F - G\bar{C}_1 \end{pmatrix} \begin{pmatrix} x \\ w \end{pmatrix} + \begin{pmatrix} \bar{E} \\ G\bar{D}_1 \end{pmatrix} d,$$

$$z = (\bar{C}_2 \quad \bar{D}_2F) \begin{pmatrix} x \\ w \end{pmatrix}.$$

$$(6.12)$$

Denote

$$A_a = \begin{pmatrix} \bar{A} & \bar{B}F \\ G\bar{C}_1 & \bar{A} + \bar{B}F - G\bar{C}_1 \end{pmatrix}, \quad C_a = \begin{pmatrix} \bar{C}_2 & \bar{D}_2F \end{pmatrix}, \quad E_a = \begin{pmatrix} \bar{E} \\ G\bar{D}_1 \end{pmatrix}.$$

Then the impulse response matrix from the disturbance d to the output z is given by $T_{F,G}(t) = C_a e^{A_a t} E_a$. Next, we introduce the associated \mathcal{H}_2 cost functional, given by

$$J(F,G) := \int_0^\infty \operatorname{tr} \left[T_{F,G}^\top(t) T_{F,G}(t) \right] dt.$$
(6.13)

We are interested in the problem of finding a controller of the form (6.11) such that the controlled system (6.12) is internally stable and the associated cost (6.13) is smaller than an a priori given upper bound.

Before we proceed, we will first review a well-known result that provides *necessary and sufficient* conditions such that a closed loop system is H_2 suboptimal, see e.g. [95, Proposition 3.13].

Proposition 6.1. Let $\gamma > 0$. Then the following statements are equivalent:

- (i) the system (6.12) is internally stable and $J(F,G) < \gamma$.
- (ii) there exists $X_a > 0$ such that

$$A_a X_a + X_a A_a^\top + E_a E_a^\top < 0,$$

$$\operatorname{tr} \left(C_a X_a C_a^\top \right) < \gamma.$$
(6.14)

(iii) there exists $Y_a > 0$ such that

$$A_a^{\top} Y_a + Y_a A_a + C_a^{\top} C_a < 0,$$

$$\operatorname{tr} \left(E_a^{\top} Y_a E_a \right) < \gamma.$$
(6.15)

The following lemma is an extension of Theorem 6 in [25]. It provides conditions under which the controller (6.11) with gain matrices F and $G = Q\bar{C}_1^{\top}$ is suboptimal for the continuous-time system (6.10), where Q is a particular real symmetric solution of a given Riccati inequality. The result shows that the separation principle is also applicable in the context of \mathcal{H}_2 suboptimal control for continuous-time systems.

Lemma 6.2. Let $\gamma > 0$ be a given tolerance. Assume that $\overline{D}_1 \overline{E}^\top = 0$, $\overline{D}_2^\top \overline{C}_2 = 0$, $\overline{D}_1 \overline{D}_1^\top = I_r$ and $\overline{D}_2^\top \overline{D}_2 > 0$. Let $F \in \mathbb{R}^{m \times n}$. Suppose that there exists P > 0 satisfying

$$(\bar{A} + \bar{B}F)^{\top}P + P(\bar{A} + \bar{B}F) + (\bar{C}_2 + \bar{D}_2F)^{\top}(\bar{C}_2 + \bar{D}_2F) < 0.$$
(6.16)

Let Q > 0 be a solution of the Riccati inequality

$$\bar{A}Q + Q\bar{A}^{\top} - Q\bar{C}_{1}^{\top}\bar{C}_{1}Q + \bar{E}\bar{E}^{\top} < 0.$$
 (6.17)

If, moreover, the inequality

$$\operatorname{tr}\left(\bar{C}_{1}QPQ\bar{C}_{1}^{\top}\right) + \operatorname{tr}\left(\bar{C}_{2}Q\bar{C}_{2}^{\top}\right) < \gamma \tag{6.18}$$

holds, then the controller (6.11) with the gains F and $G = Q\bar{C}_1^{\top}$ yields an internally stable closed loop system (6.12), and it is suboptimal, i.e. $J(F,G) < \gamma$.

Proof. Let Q > 0 satisfy (6.17) and gain matrix F be given. Note that (6.18) is equivalent to

$$\operatorname{tr}\left(\bar{C}_{1}QPQ\bar{C}_{1}^{\top}\right) < \gamma - \operatorname{tr}\left(\bar{C}_{2}Q\bar{C}_{2}^{\top}\right).$$
(6.19)

According to cases (ii) and (iii) in Proposition 6.1, there exists P > 0 satisfying (6.16) and (6.19) if and only if there exists $\Delta > 0$ satisfying

$$\operatorname{tr}\left((\bar{C}_2 + \bar{D}_2 F)\Delta(\bar{C}_2 + \bar{D}_2 F)^{\top}\right) < \gamma - \operatorname{tr}\left(\bar{C}_2 Q \bar{C}_2^{\top}\right)$$
(6.20)

and

$$(\bar{A} + \bar{B}F)\Delta + \Delta(\bar{A} + \bar{B}F)^{\top} + Q\bar{C}_1^{\top}\bar{C}_1Q < 0.$$
(6.21)

On the other hand, by applying the state transformation

$$\begin{pmatrix} w \\ e \end{pmatrix} = \begin{pmatrix} 0 & I_n \\ -I_n & I_n \end{pmatrix} \begin{pmatrix} x \\ w \end{pmatrix}.$$

The system (6.12) then becomes

$$\begin{pmatrix} \dot{w} \\ \dot{e} \end{pmatrix} = \begin{pmatrix} \bar{A} + \bar{B}F & -G\bar{C}_1 \\ 0 & \bar{A} - G\bar{C}_1 \end{pmatrix} \begin{pmatrix} w \\ e \end{pmatrix} + \begin{pmatrix} G\bar{D}_1 \\ G\bar{D}_1 - \bar{E} \end{pmatrix} d,$$

$$z = (\bar{C}_2 + \bar{D}_2F & -\bar{C}_2) \begin{pmatrix} w \\ e \end{pmatrix}.$$

$$(6.22)$$

Clearly, the system (6.12) is internally stable if and only if $\overline{A} + \overline{B}F$ and $\overline{A} - G\overline{C}_1$ are Hurwitz. Thus, what remains to show is that the controller (6.11) with the gains F and $G = Q\overline{C}_1^{\top}$ internally stabilizes the system (6.10) and that $J(F,G) < \gamma$.

Note that (6.17) is equivalent to

$$(\bar{A} - Q\bar{C}_1^{\top}\bar{C}_1)Q + Q(\bar{A} - Q\bar{C}_1^{\top}\bar{C}_1)^{\top} + (\bar{E} + Q\bar{C}_1^{\top}\bar{D}_1)(\bar{E} + Q\bar{C}_1^{\top}\bar{D}_1)^{\top} < 0,$$
(6.23)

where we use the fact that $\bar{D}_1 \bar{E}^{\top} = 0$ and $\bar{D}_1 \bar{D}_1^{\top} = I_r$. Since $G = Q \bar{C}_1^{\top}$, it then follows that $\bar{A} - G \bar{C}_1$ is Hurwitz. Similarly, it follows from (6.16) that $\bar{A} + \bar{B}F$ is

Hurwitz. Consequently, the system (6.12) is internally stable.

Next, we will show that $J(F,G) < \gamma$. Again consider (6.22) and denote

$$\bar{A}_a = \begin{pmatrix} \bar{A} + \bar{B}F & -G\bar{C}_1 \\ 0 & \bar{A} - G\bar{C}_1 \end{pmatrix}, \quad \bar{C}_a = \begin{pmatrix} \bar{C}_2 + \bar{D}_2F & -\bar{C}_2 \end{pmatrix}, \quad \bar{E}_a = \begin{pmatrix} G\bar{D}_1 \\ G\bar{D}_1 - \bar{E} \end{pmatrix}.$$

According to Proposition 6.1, in particular the inequalities in (6.14), we have $J(F,G) < \gamma$ if and only if there exists $P_a > 0$ satisfying

$$\bar{A}_a P_a + P_a \bar{A}_a^\top + \bar{E}_a \bar{E}_a^\top < 0,
\operatorname{tr}(\bar{C}_a P_a \bar{C}_a^\top) < \gamma.$$
(6.24)

We will show that the existence of solutions Q > 0 and $\Delta > 0$ to the inequalities (6.17), (6.20) and (6.21) implies that (6.24) has a solution $P_a > 0$. Let

$$P_a = \begin{pmatrix} \Delta & 0\\ 0 & Q \end{pmatrix}.$$

Clearly, $P_a > 0$. By substituting P_a , \bar{A}_a , \bar{E}_a and \bar{C}_a into (6.24), we obtain

$$\begin{pmatrix} R_1 & R_{12} \\ R_{12}^\top & R_2 \end{pmatrix} < 0, \tag{6.25}$$

where

$$R_{1} = (\bar{A} + \bar{B}F)\Delta + \Delta(\bar{A} - \bar{B}F)^{\top} + GG^{\top},$$

$$R_{12} = -G\bar{C}_{1}Q + GG^{\top},$$

$$R_{2} = (\bar{A} - G\bar{C}_{1})Q + Q(\bar{A} - G\bar{C}_{1})^{\top} + (G\bar{D}_{1} - \bar{E})(G\bar{D}_{1} - \bar{E})^{\top},$$

and

$$\operatorname{tr}\left((\bar{C}_2 + \bar{D}_2 F)\Delta(\bar{C}_2 + \bar{D}_2 F)^{\top}\right) + \operatorname{tr}\left(\bar{C}_2 Q \bar{C}_2^{\top}\right) < \gamma.$$
(6.26)

It then follows from $G = Q\bar{C}_1^{\top}$, (6.17) and (6.21) that $R_1 < 0$, $R_{12} = 0$ and $R_2 < 0$. Subsequently, R < 0. Also, it follows from (6.20) that (6.26) holds. Hence, $J(F,G) < \gamma$. This completes the proof.

Theorem 6.3. Let $\gamma > 0$. Assume that $\bar{D}_1 \bar{E}^{\top} = 0$, $\bar{D}_2^{\top} \bar{C}_2 = 0$ and $\bar{D}_1 \bar{D}_1^{\top} = I_r$, $\bar{D}_2^{\top} \bar{D}_2 > 0$. Suppose that there exist P > 0 and Q > 0 satisfying

$$\bar{A}^{\top}P + P\bar{A} - P\bar{B}(\bar{D}_{2}^{\top}\bar{D}_{2})^{-1}\bar{B}^{\top}P + \bar{C}_{2}^{\top}\bar{C}_{2} < 0,$$
(6.27)

$$\bar{A}Q + Q\bar{A}^{\top} - Q\bar{C}_{1}^{\top}\bar{C}_{1}Q + \bar{E}\bar{E}^{\top} < 0,$$
 (6.28)

$$\operatorname{tr}\left(\bar{C}_{1}QPQ\bar{C}_{1}^{\top}\right) + \operatorname{tr}\left(\bar{C}_{2}Q\bar{C}_{2}^{\top}\right) < \gamma.$$
(6.29)

Let $G = Q\bar{C}_1^{\top}$ and $F = -(\bar{D}_2^{\top}\bar{D}_2)^{-1}\bar{B}^{\top}P$. Then the controller (6.11) internally stabilizes the system (6.10), and it is suboptimal, i.e. $J(F,G) < \gamma$.

Proof. Substituting $F = -(\bar{D}_2^{\top}\bar{D}_2)^{-1}\bar{B}^{\top}P$ into (6.16) gives us the inequality (6.27). The rest follows from Lemma 6.2.

We are now ready to deal with the distributed H_2 suboptimal control problem by dynamic output feedback for multi-agent systems.

6.4 Distributed H_2 suboptimal protocols by dynamic output feedback

In this section, we will address Problem 6.1. For the multi-agent system (6.1), we will establish a design method for local gains F and G such that the protocol (6.2) achieves $J(F, G) < \gamma$ and synchronizes the network (6.7).

Let *U* be an orthogonal matrix such that $U^{\top}LU = \Lambda = \text{diag}(0, \lambda_2, ..., \lambda_N)$ with $0 = \lambda_1 < \lambda_2 \leq \cdots \leq \lambda_N$ the eigenvalues of the Laplacian matrix. We apply the state transformation

$$\begin{pmatrix} \bar{\mathbf{x}} \\ \bar{\mathbf{w}} \end{pmatrix} = \begin{pmatrix} U^{\top} \otimes I_n & 0 \\ 0 & U^{\top} \otimes I_n \end{pmatrix} \begin{pmatrix} \mathbf{x} \\ \mathbf{w} \end{pmatrix}.$$
 (6.30)

Then the controlled network (6.7) is also represented by

$$\begin{pmatrix} \dot{\mathbf{x}} \\ \dot{\mathbf{w}} \end{pmatrix} = \begin{pmatrix} I_N \otimes A & I_N \otimes BF \\ \Lambda \otimes GC_1 & I_N \otimes (A - GC_1) + \Lambda \otimes BF \end{pmatrix} \begin{pmatrix} \bar{\mathbf{x}} \\ \bar{\mathbf{w}} \end{pmatrix} + \begin{pmatrix} U^\top \otimes E \\ U^\top L \otimes GD_1 \end{pmatrix} \mathbf{d},$$
$$\boldsymbol{\zeta} = \begin{pmatrix} W^{\frac{1}{2}} R^\top U \otimes C_2 & W^{\frac{1}{2}} R^\top U \otimes D_2F \end{pmatrix} \begin{pmatrix} \bar{\mathbf{x}} \\ \bar{\mathbf{w}} \end{pmatrix}.$$
(6.31)

Denote

$$\bar{A}_e = \begin{pmatrix} I_N \otimes A & I_N \otimes BF \\ \Lambda \otimes GC_1 & I_N \otimes (A - GC_1) + \Lambda \otimes BF \end{pmatrix},$$
$$\bar{C}_e = \begin{pmatrix} W^{\frac{1}{2}}R^{\top}U \otimes C_2 & W^{\frac{1}{2}}R^{\top}U \otimes D_2F \end{pmatrix},$$
$$\bar{E}_e = \begin{pmatrix} U^{\top} \otimes E \\ U^{\top}L \otimes GD_1 \end{pmatrix}.$$

Obviously, the impulse response matrix $T_{F,G}(t)$ given by (6.8) is then equal to $\bar{C}_e e^{\bar{A}_e t} \bar{E}_e$.

In order to proceed, we now introduce the N - 1 auxiliary linear systems

$$\begin{aligned} \xi_i &= A\xi_i + \lambda_i B v_i + E\delta_i, \\ \vartheta_i &= C_1 \xi_i + D_1 \delta_i, \\ \eta_i &= \sqrt{\lambda_i} C_2 \xi_i + \lambda_i \sqrt{\lambda_i} D_2 v_i, \end{aligned}$$
(6.32)

and associated dynamic output feedback controllers

$$\dot{\omega}_i = A\omega_i + \lambda_i Bv_i + G(\vartheta_i - C_1\omega_i), v_i = F\omega_i, \quad i = 2, 3, \dots, N$$
(6.33)

with gain matrices F and G. By interconnecting (6.33) and (6.32), we obtain the N - 1 closed loop systems

$$\begin{pmatrix} \dot{\xi}_i \\ \dot{\omega}_i \end{pmatrix} = \begin{pmatrix} A & \lambda_i BF \\ GC_1 & A - GC_1 + \lambda_i BF \end{pmatrix} \begin{pmatrix} \xi_i \\ \omega_i \end{pmatrix} + \begin{pmatrix} E \\ GD_1 \end{pmatrix} \delta_i,$$

$$\eta_i = \left(\sqrt{\lambda_i} C_2 & \lambda_i \sqrt{\lambda_i} D_2 F\right) \begin{pmatrix} \xi_i \\ \omega_i \end{pmatrix},$$
 (6.34)

for i = 2, 3, ..., N. The impulse response matrix of (6.34) from the disturbance δ_i to the output η_i is equal to

$$T_{i,F,G}(t) = \bar{C}_i e^{A_i t} \bar{E}_i \tag{6.35}$$

with

$$\bar{A}_i = \begin{pmatrix} A & \lambda_i BF \\ GC_1 & A - GC_1 + \lambda_i BF \end{pmatrix}, \ \bar{E}_i = \begin{pmatrix} E \\ GD_1 \end{pmatrix}, \ \bar{C}_i = \begin{pmatrix} \sqrt{\lambda_i} C_2 & \lambda_i \sqrt{\lambda_i} D_2F \end{pmatrix}.$$

Furthermore, for each system (6.32) the associated H_2 cost functional is given by

$$J_i(F,G) := \int_0^\infty \operatorname{tr} \left[T_{i,F,G}^\top(t) T_{i,F,G}(t) \right] dt, \quad i = 2, 3, \dots, N.$$
 (6.36)

Then we have the following lemma:

Lemma 6.4. Let $F \in \mathbb{R}^{m \times n}$ and $G \in \mathbb{R}^{n \times r}$. Then the dynamic protocol (6.2) with gain matrices F and G achieves synchronization for the network (6.7) if and only if for each i = 2, 3, ..., N the controller (6.33) with gain matrices F and G internally stabilizes the system (6.32). Moreover, we have

$$J(F,G) = \sum_{i=2}^{N} J_i(F,G).$$
(6.37)

Proof. It follows immediately from [103, Lemmas 3.2 and 3.3] that the dynamic protocol (6.2) achieves synchronization for the network (6.7) if and only if for i = 2, 3, ..., N the system (6.32) is internally stabilized by the controller (6.33).

Next, we prove (6.37). Let F and G be such that synchronization is achieved. Then we have

$$J(F,G) = \int_0^\infty \operatorname{tr}\left(\bar{E}_e^\top e^{\bar{A}_e^\top t} \bar{C}_e^\top \bar{C}_e e^{\bar{A}_e t} \bar{E}_e\right) dt.$$

Since $U^{\top}LU = \Lambda$, $L = RWR^{\top}$, we have $\bar{C}_e^{\top}\bar{C}_e = \tilde{C}_e^{\top}\tilde{C}_e$ with

$$\tilde{C}_e := \left(\Lambda^{\frac{1}{2}} \otimes C_2 \ \Lambda^{\frac{1}{2}} \otimes D_2 F\right).$$

We also have $\bar{E}_e \bar{E}_e^{\top} = \tilde{E}_e \tilde{E}_e^{\top}$ with

$$\tilde{E}_e := \begin{pmatrix} I_N \otimes E \\ \Lambda \otimes GD_1 \end{pmatrix}.$$

Thus we find that

$$\operatorname{tr}\left(\bar{E}_{e}^{\top}e^{\bar{A}_{e}^{\top}t}\bar{C}_{e}^{\top}\bar{C}_{e}e^{\bar{A}_{e}t}\bar{E}_{e}\right) = \operatorname{tr}\left(\tilde{E}_{e}^{\top}e^{\bar{A}_{e}^{\top}t}\tilde{C}_{e}^{\top}\tilde{C}_{e}e^{\bar{A}_{e}t}\tilde{E}_{e}\right).$$
(6.38)

We now analyze the matrix function $\tilde{C}_e e^{\bar{A}_e t} \tilde{E}_e$ appearing in (6.38). By applying suitable permutations of the blocks appearing in the matrices \tilde{C}_e , \tilde{E}_e and \bar{A}_e , it is straightforward to show that

$$\tilde{C}_e e^{\bar{A}_e t} \tilde{E}_e = \text{blockdiag} \left(0, C_2 e^{A_2 t} E_2, \dots, C_N e^{A_N t} E_N \right),$$

where

$$A_i := \begin{pmatrix} A & BF \\ \lambda_i GC_1 & A - GC_1 + \lambda_i BF \end{pmatrix}, \ C_i := \begin{pmatrix} \sqrt{\lambda_i} C_2 & \sqrt{\lambda_i} D_2 F \end{pmatrix}, \ E_i := \begin{pmatrix} E \\ \lambda_i GD_1 \end{pmatrix}.$$

It is easily seen that for i = 2, 3, ..., N the systems (A_i, E_i, C_i) and $(\bar{A}_i, \bar{E}_i, \bar{C}_i)$ are isomorphic. Hence they have the same impulse response $T_{i,F,G}(t)$, which is given by (6.35), see e.g., [102, Theorem 3.10]. As a consequence we obtain that

$$\tilde{C}_e e^{A_e t} \tilde{E}_e = \text{blockdiag}(0, T_{2,F,G}(t), \dots, T_{N,F,G}(t))$$

Thus we find that

$$J(F,G) = \int_0^\infty \sum_{i=2}^N \operatorname{tr}\left[T_{i,F,G}^\top(t)T_{i,F,G}(t)\right] dt$$

The claim (6.37) then follows immediately.

By applying Lemma 6.4, we have transformed the distributed \mathcal{H}_2 suboptimal control problem by dynamic output feedback for the multi-agent network (6.7) into \mathcal{H}_2 suboptimal control problems for the N-1 linear systems (6.32) using controllers (6.33) with the same gain matrices F and G. Next, we establish conditions under which the N-1 systems (6.32) are internally stabilized by their corresponding controllers (6.33) for i = 2, 3, ..., N, while achieving $\sum_{i=2}^{N} J_i(F, G) < \gamma$.

Lemma 6.5. Let $\gamma > 0$ be a given tolerance. Assume that $D_1 E^{\top} = 0$, $D_2^{\top} C_2 = 0$, $D_1 D_1^{\top} = I_r$ and $D_2^{\top} D_2 = I_m$. For i = 2, 3, ..., N, let F, $P_i > 0$, and Q > 0 be such that the inequalities

$$(A + \lambda_i BF)^\top P_i + P_i (A + \lambda_i BF) + (\sqrt{\lambda_i} C_2 + \lambda_i \sqrt{\lambda_i} D_2 F)^\top (\sqrt{\lambda_i} C_2 + \lambda_i \sqrt{\lambda_i} D_2 F) < 0,$$
(6.39)
$$AQ + QA^\top - QC_1^\top C_1 Q + EE^\top < 0,$$
(6.40)

$$Q + QA^{\top} - QC_1^{\top}C_1Q + EE^{\top} < 0, \tag{6.40}$$

$$\sum_{i=2}^{N} \left[\operatorname{tr} \left(C_1 Q P_i Q C_1^{\top} \right) + \lambda_i \operatorname{tr} \left(C_2 Q C_2^{\top} \right) \right] < \gamma$$
(6.41)

hold. Then for each i = 2, 3, ..., N, the controller (6.33) with gain matrices F and $G = QC_1^{\top}$ internally stabilizes the system (6.32), and, moreover, $\sum_{i=2}^{N} J_i(F,G) < \gamma$.

Proof. By (6.41), for $\epsilon_i > 0$ sufficiently small, we have $\sum_{i=2}^N \gamma_i < \gamma$, where $\gamma_i :=$ $\operatorname{tr}\left(C_{1}QP_{i}QC_{1}^{\top}\right)+\lambda_{i}\operatorname{tr}\left(C_{2}QC_{2}^{\top}\right)+\epsilon_{i}. \text{ Since } \operatorname{tr}\left(C_{1}QP_{i}QC_{1}^{\top}\right)+\lambda_{i}\operatorname{tr}\left(C_{2}QC_{2}^{\top}\right)<\gamma_{i},$ by taking $\overline{A} = A$, $\overline{B} = \lambda_i B$, $\overline{C}_1 = C_1$, $\overline{D}_1 = D_1$, $\overline{C}_2 = \sqrt{\lambda_i} C_2$, $\overline{D}_2 = \lambda_i \sqrt{\lambda_i} D_2$, $\bar{C}_1 = C_1$ and $\bar{E} = E$ in Lemma 6.2, it follows that the controller (6.33) internally stabilizes the system (6.32) and $J_i(F,G) < \gamma_i$. Thus, from $\sum_{i=2}^N \gamma_i < \gamma$ it follows that $\sum_{i=2}^{N} J_i(F, G) < \gamma$.

Again, we note that the four conditions $D_1 E^{\top} = 0$, $D_2^{\top} C_2 = 0$, $D_1 D_1^{\top} = I_r$ and $D_2^{\top} D_2 = I_m$ are made here to simplify notation, and can be replaced by the regularity conditions $D_1 D_1^{\top} > 0$ and $D_2^{\top} D_2 > 0$ alone.

By combining Lemma 6.4 and Lemma 6.5 we have established sufficient conditions for given gain matrices F and G to synchronize the network (6.7) and to be suboptimal, i.e. $J(F,G) < \gamma$. In fact, G is taken to be equal to QC_1^{\top} , with Q > 0a solution to the Riccati inequality (6.40). However, no design method has yet been provided to compute a suitable matrix F. In the following theorem, we will establish a design method for computing such gain matrix F. Together with Ggiven above, this will lead to a distributed suboptimal protocol for multi-agent system (6.1) with associated cost functional (6.9).

Theorem 6.6. Let $\gamma > 0$ be a given tolerance. Assume that $D_1 E^{\top} = 0$, $D_2^{\top} C_2 = 0$, $D_1 D_1^{\top} = I_r$ and $D_2^{\top} D_2 = I_m$. Let Q > 0 satisfy

$$AQ + QA^{\top} - QC_1^{\top}C_1Q + E^{\top}E < 0.$$
(6.42)

Let c be any real number such that $0 < c < \frac{2}{\lambda_N^2}$. We distinguish two cases:

(*i*) *if*

$$\frac{2}{\lambda_2^2 + \lambda_2 \lambda_N + \lambda_N^2} \leqslant c < \frac{2}{\lambda_N^2},\tag{6.43}$$

then there exists P > 0 satisfying

$$A^{\top}P + PA + (c^2\lambda_N^3 - 2c\lambda_N)PBB^{\top}P + \lambda_N C_2^{\top}C_2 < 0.$$
(6.44)

(ii) if

$$0 < c < \frac{2}{\lambda_2^2 + \lambda_2 \lambda_N + \lambda_N^2},\tag{6.45}$$

then there exists P > 0 satisfying

$$A^{\top}P + PA + (c^{2}\lambda_{2}^{3} - 2c\lambda_{2})PBB^{\top}P + \lambda_{N}C_{2}^{\top}C_{2} < 0.$$
(6.46)

In both cases, if in addition P and Q satisfy

$$\operatorname{tr}\left(C_{1}QPQC_{1}^{\top}\right) + \lambda_{N}\operatorname{tr}\left(C_{2}QC_{2}^{\top}\right) < \frac{\gamma}{N-1}.$$
(6.47)

Then the protocol (6.2) with $F := -cB^{\top}P$ and $G := QC_1^{\top}$ synchronizes the network (6.7) and it is suboptimal, i.e. $J(F, G) < \gamma$.

Proof. We will only provide the proof for case (i) above. Using the upper and lower bound on *c* given by (6.43), it can be verified that $c^2 \lambda_N^3 - 2c\lambda_N < 0$. Thus the Riccati inequality (6.44) has positive definite solutions. Since $c^2 \lambda_i^3 - 2c\lambda_i \leq c^2 \lambda_N^3 - 2c\lambda_N < 0$ and $\lambda_i \leq \lambda_N$ for i = 2, 3, ..., N, any positive definite solution *P* of (6.44) also satisfies the N - 1 Riccati inequalities

$$A^{\top}P + PA + (c^2\lambda_i^3 - 2c\lambda_i)PBB^{\top}P + \lambda_i C_2^{\top}C_2 < 0,$$
(6.48)

equivalently,

$$(A - c\lambda_i BB^{\top}P)^{\top}P + P(A - c\lambda_i BB^{\top}P) + c^2\lambda_i^3 PBB^{\top}P + \lambda_i C_2^{\top}C_2 < 0,$$
(6.49)

for i = 2, ..., N. Using the conditions $D_2^{\top}C_2 = 0$ and $D_2^{\top}D_2 = I_m$ this yields

$$(A - c\lambda_i BB^{\top}P)^{\top}P + P(A - c\lambda_i BB^{\top}P) + (\sqrt{\lambda_i}C_2 + \lambda_i\sqrt{\lambda_i}D_2B^{\top}P)^{\top}(\sqrt{\lambda_i}C_2 + \lambda_i\sqrt{\lambda_i}D_2B^{\top}P) < 0,$$
(6.50)

for i = 2, ..., N. Taking $P_i = P$ for i = 2, 3, ..., N and $F = -cB^{\top}P$ in (6.50) immediately yields (6.39). Next, it follows from (6.47) that also (6.41) holds. By Lemma 6.5 then, all systems (6.32) are internally stabilized and $\sum_{i=2}^{N} J_i(F, G) < \gamma$. Subsequently, it follows from Lemma 6.4 that the protocol (6.2) achieves synchronization for the network (6.7) and $J(F, G) < \gamma$.

Remark 6.7. In Theorem 6.6, in order to select γ , the following should be done:

(i) First compute a solution Q > 0 of the Riccati inequality (6.42) and a solution P > 0 of the Riccati inequality (6.44) (or (6.46), depending on the choice of parameter *c*). Note that these solutions exist.

(ii) Let
$$S(P,Q) := \operatorname{tr}(C_1 Q P Q C_1^{\top}) + \lambda_N \operatorname{tr}(C_2 Q C_2^{\top}).$$

(iii) Then choose $\gamma > 0$ such that $(N-1)S(P,Q) < \gamma$.

Obviously, the smaller S(P, Q), the smaller the feasible upper bound γ . It can be shown that, unfortunately, the problem of minimizing S(P, Q) over all P, Q > 0 that satisfy (6.42) and (6.44) is a nonconvex optimization problem. However, since smaller Q leads to smaller tr $(C_2 Q C_2^{\top})$ and smaller P and Q leads to smaller tr $(C_1 Q P Q C_1^{\top})$ and, consequently, smaller feasible γ , we could therefore try to find P and Q as small as possible. In fact, one can find $Q = Q(\epsilon) > 0$ to (6.42) by solving

$$AQ + QA^{\top} - QC_1^{\top}C_1Q + E^{\top}E + \epsilon I_n = 0.$$
(6.51)

with $\epsilon > 0$ arbitrary. By using a standard argument, it can be shown that $Q(\epsilon)$ decreases as ϵ decreases, so ϵ should be taken close to 0 in order to get small Q. Similarly, one can find $P = P(c, \sigma) > 0$ satisfying (6.44) by solving

$$A^{\top}P + PA - PBR(c)^{-1}B^{\top}P + \lambda_N C_2^{\top}C_2 + \sigma I_n = 0$$
(6.52)

with $R(c) = \frac{1}{-c^2 \lambda_N^3 + 2c\lambda_N} I_n$, where *c* is chosen as in (6.43) and $\sigma > 0$ arbitrary. Again, it can be shown that $P(c, \sigma)$ decreases with decreasing σ and *c*. Therefore, small *P* is obtained by choosing $\sigma > 0$ close to 0 and $c = \frac{2}{\lambda_2^2 + \lambda_2 \lambda_N + \lambda_N^2}$.

Similarly, if *c* satisfies (6.45) corresponding to case (ii), it can be shown that if we choose $\epsilon > 0$ and $\sigma > 0$ very close to 0 and c > 0 very close to $\frac{2}{\lambda_2^2 + \lambda_2 \lambda_N + \lambda_N^2}$, we find small solutions to the Riccati inequalities (6.42) and (6.46) in the sense as explained above for case (i).

Remark 6.8. In Theorem 6.6, exact knowledge of the largest and the smallest nonzero eigenvalue of the Laplacian matrix is used to compute the local control gains *F* and *G*. We want to remark that our results can be extended to the case that only lower and upper bounds for these eigenvalues are known. In the literature, algorithms are given to estimate λ_2 in a distributed way, yielding lower and upper bounds, see e.g. [2]. Also, an upper bound for λ_N can be obtained in terms of the maximal node degree of the graph, see e.g. [1]. Using these lower and upper bounds on the largest and the smallest nonzero eigenvalue of the Laplacian matrix, results similar to Theorem 6.6 can be formulated, see e.g., [37] or [27].

6.5 Simulation example

In this section, we will give a simulation example to illustrate our design method. Consider a network of N = 6 identical agents of the form

$$\dot{x}_i = Ax_i + Bu_i + Ed_i,$$

 $y_i = C_1 x_i + D_1 d_i,$ $i = 1, 2, \dots, 6$
 $z_i = C_2 x_i + D_2 u_i,$

where $A = \begin{pmatrix} -2 & 2 \\ -1 & 1 \end{pmatrix}$, $B = \begin{pmatrix} 0 \\ 1 \end{pmatrix}$, $E = \begin{pmatrix} 0 & 0 \\ 0.5 & 0 \end{pmatrix}$, $C_1 = \begin{pmatrix} 1 & 0 \end{pmatrix}$, $D_1 = \begin{pmatrix} 0 & 1 \end{pmatrix}$, $C_2 = \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix}$, $D_2 = \begin{pmatrix} 0 \\ 1 \end{pmatrix}$. The pair (A, B) is stabilizable and the pair (C_1, A) is detectable. We also have $D_1E^{\top} = \begin{pmatrix} 0 & 0 \end{pmatrix}$, $D_2^{\top}C_2 = \begin{pmatrix} 0 & 0 \end{pmatrix}$ and $D_1D_1^{\top} = 1$, $D_2^{\top}D_2 = 1$. We assume that the communication among the six agents is represented by the undirected cycle graph. For this graph, the smallest non-zero and largest eigenvalue of the Laplacian are $\lambda_2 = 1$ and $\lambda_6 = 4$. Our goal is to design a distributed dynamic output feedback protocol of the form (6.2) that synchronizes the controlled network and guarantees the associated cost (6.9) to satisfy $J(F, G) < \gamma$. Let the desired upper bound for the cost be $\gamma = 17$.

We adopt the design method given in case (i) of Theorem 6.6. First we compute a positive definite solution P to (6.44) by solving the Riccati equation

$$A^{\top}P + PA + (c^2\lambda_6^3 - 2c\lambda_6)PBB^{\top}P + \lambda_6C_2^{\top}C_2 + \sigma I_2 = 0$$
 (6.53)

with $\sigma = 0.001$. Moreover, we choose $c = \frac{2}{\lambda_2^2 + \lambda_2 \lambda_6 + \lambda_6^2} = 0.0952$, which is in fact the 'best' choice for *c* as explained in Remark 6.7. Then, by solving (6.53) in Matlab,



Figure 6.1: Plots of the state vector $\mathbf{x}^1 = (x_{1,1}, x_{2,1}, \dots, x_{6,1})^{\top}$ and $\mathbf{x}^2 = (x_{1,2}, x_{2,2}, \dots, x_{6,2})^{\top}$ of the controlled network

we compute a positive definite solution

$$P = \begin{pmatrix} 0.9048 & -2.2810\\ -2.2810 & 6.9779 \end{pmatrix}.$$

Next, by solving the Riccati equation

$$AQ + QA^{\top} - QC_1^{\top}C_1Q + E^{\top}E + \epsilon I_2 = 0$$

with $\epsilon = 0.001$ in Matlab, we compute a positive definite solution

$$Q = \begin{pmatrix} 0.5000 & 0.5000\\ 0.5000 & 0.6250 \end{pmatrix}.$$

Accordingly, we compute the associated gain matrices

$$F = (0.2172 \quad -0.6646), \quad G = (0.5000 \quad 0.5000)^{+}$$

As an example, we take the initial states of the agents to be $x_{10} = (1 - 2)^{\top}$, $x_{20} = (2 -5)^{\top}$, $x_{30} = (3 -1)^{\top}$, $x_{40} = (4 - 2)^{\top}$, $x_{50} = (-1 - 2)^{\top}$ and $x_{60} = (-3 - 1)^{\top}$, and we take the initial states of the protocol to be zero. In Figure 6.1, we have plotted the controlled state trajectories of the agents. It can be seen that the designed protocol indeed synchronizes the network. The plots of the protocol



Figure 6.2: Plots of the state vector $\mathbf{w}^1 = (w_{1,1}, w_{2,1}, \dots, w_{6,1})^\top$ and $\mathbf{w}^2 = (w_{1,2}, w_{2,2}, \dots, w_{6,2})^\top$ of the dynamic protocol

states are shown in Figure 6.2. For each i, the state w_i of the local controller is an estimate of the weighted sum of the relative states of agent i, it is seen that the protocol states converge to zero. Moreover, we compute

$$5\left(\operatorname{tr}\left(C_{1}QPQC_{1}^{\top}\right)+\lambda_{6}\operatorname{tr}\left(C_{2}QC_{2}^{\top}\right)\right)=16.6509,$$

which is indeed smaller than the desired tolerance $\gamma = 17$.

6.6 Conclusions

In this chapter, we have studied the distributed \mathcal{H}_2 suboptimal control problem by dynamic output feedback for linear multi-agent systems. The interconnection structure between the agents is given by a connected undirected graph. Given a linear multi-agent system with identical agent dynamics and an associated global \mathcal{H}_2 cost functional, we have provided a design method for computing distributed protocols that guarantee the associated cost to be smaller than a given tolerance while synchronizing the controlled network. The local gains are given in terms of solutions of two Riccati inequalities, each of dimension equal to that of the agent dynamics. One these Riccati inequalities involves the largest and smallest nonzero eigenvalue of the Laplacian matrix of the network graph.

$\begin{array}{c} \mathcal{H}_2 \text{ suboptimal output synchronization} \\ \text{ of heterogeneous multi-agent systems} \end{array} \end{array}$

This chapter deals with the \mathcal{H}_2 suboptimal output synchronization problem for heterogeneous linear multi-agent systems. Given a multi-agent system with possibly distinct agents and an associated \mathcal{H}_2 cost functional, the aim is to design output feedback based protocols that guarantee the associated cost to be smaller than a given upper bound while the controlled network achieves output synchronization. A design method is provided to compute such protocols. For each agent, the computation of its two local control gains involves two Riccati inequalities, each of dimension equal to the state space dimension of the agent. We also consider the special case that full relative state information is available for each agent. A simulation example is provided to illustrate the performance of the proposed protocols.

7.1 Introduction

Over the last two decades, the problems of designing protocols that achieve consensus or synchronization in multi-agent systems have attracted much attention in the field of systems and control, see e.g. [85], [7], [54] and [11]. The essential feature of these problems is that, while each agent makes use of only local state or output information to implement its own local controller, the resulting global protocol will achieve consensus or synchronization for the global controlled multi-agent network [94], [103]. One of the challenging problems in this context is the problem of designing protocols that minimize given quadratic cost criteria while achieving consensus or synchronization, see e.g. [37], [39], [10], [75] and [76]. Due to the structural constraints imposed on the protocols, such optimal control problems are non-convex and very difficult to solve. It is also unclear whether in general closed form solutions exist.

In the past, many efforts have been devoted to designing distributed protocols for *homogeneous* multi-agent systems that guarantee suboptimal or optimal performance and achieve *state* synchronization or consensus. In [10], this was
done for distributed linear quadratic control of multi-agent systems with *single integrator* agent dynamics, see also [38]. In [76] and [37], multi-agent systems with general agent dynamics and a global linear quadratic cost functional were considered. In [75] and [123], an inverse optimal approach was adopted to address the distributed linear quadratic control problem, see also [77]. For \mathcal{H}_2 cost functionals of a particular form, [55] and [53] proposed distributed suboptimal protocols that stabilize the controlled multi-agent network. In [36], a distributed \mathcal{H}_2 suboptimal control problem was addressed using static state feedback. The results in [36] were then generalized in [39] to the case of dynamic output feedback.

More recently, *output* synchronization problems for *heterogeneous* multi-agent systems have also attracted much attention. In [115], it was shown that solvability of certain regulator equations is a necessary condition for output synchronization of heterogeneous multi-agent systems, and suitable protocols were proposed, see also [22]. In [46], by embedding an internal model in the local controller of each agent, dynamic output feedback based protocols were proposed for a class of heterogeneous uncertain multi-agent systems. In [59], it was shown that the outputs of the agents can be synchronized by a networked protocol if and only if these agents have certain dynamics in common. Later on, in [72] a linear quadratic control method was adopted for computing output synchronizing protocols. In [42], an \mathcal{L}_2 -gain output synchronization problem was addressed by casting this problem into a number of \mathcal{L}_2 -gain stabilization problems for certain linear systems, where the state space dimensions of these systems are equal to that of the agents. For related work, we also mention [51], [120] and [99], to name a few.

Up to now, little attention has been paid in the literature to problems of designing output synchronizing protocols for heterogeneous multi-agent systems that guarantee a certain performance. In the present chapter, we will deal with the problem of \mathcal{H}_2 optimal output synchronization for heterogeneous linear multiagent systems, i.e. the problem of minimizing a given \mathcal{H}_2 cost functional over all protocols that achieve output synchronization. Instead of addressing this *optimal* control problem, we will address a version of this problem that requires *suboptimality*. More specifically, we will extend previous results in [39] for homogeneous multi-agent systems to the case of heterogeneous multi-agent systems.

The outline of this chapter is as follows. In Section 7.2, we formulate the \mathcal{H}_2 suboptimal output synchronization problem. In order to solve this problem, in Section 7.3 we review some basic material on \mathcal{H}_2 suboptimal control by dynamic output feedback for linear systems, and some relevant results on output synchronization of heterogeneous multi-agent systems. In Section 7.4, we solve the problem introduced in Section 7.2 and provide a design method for obtaining \mathcal{H}_2 suboptimal protocols. In Section 7.5, we solve the special case of the \mathcal{H}_2 suboptimal output synchronization problem in which full state information of the agents is available. To illustrate the performance of our proposed protocols, a simulation

example is provided in Section 7.6. Finally, Section 7.7 concludes this chapter.

7.2 **Problem formulation**

In this chapter, we consider a heterogeneous linear multi-agent system consisting of *N* possibly distinct agents. The dynamics of the *i*th agent is represented by the linear time-invariant system

$$\dot{x}_{i} = A_{i}x_{i} + B_{i}u_{i} + E_{i}d_{i},
y_{i} = C_{1i}x_{i} + D_{1i}d_{i}, \qquad i = 1, 2, \dots, N,
z_{i} = C_{2i}x_{i} + D_{2i}u_{i},$$
(7.1)

where $x_i \in \mathbb{R}^{n_i}$ is the state, $u_i \in \mathbb{R}^{m_i}$ is the coupling input, $d_i \in \mathbb{R}^{q_i}$ is an unknown external disturbance input, $y_i \in \mathbb{R}^{r_i}$ is the measured output and $z_i \in \mathbb{R}^p$ is the output to be synchronized. The matrices A_i , B_i , C_{1i} , D_{1i} , C_{2i} , D_{2i} and E_i are of suitable dimensions. Throughout this chapter we assume that the pairs (A_i, B_i) are stabilizable and the pairs (C_{1i}, A_i) are detectable. Since in (7.1) the agents may have non-identical dynamics, in particular the state space dimensions of the agents may differ. Therefore, one can not expect to achieve *state* synchronization for the network. Instead, in the context of heterogeneous networks it is natural to consider *output* synchronization, see e.g. [115], [22] and [59].

It was shown in [115] that solvability of certain *regulator equations* is necessary for output synchronization of heterogeneous linear multi-agent systems, see also [22], [42], [99] and [3]. Following up on this, throughout this chapter we make the standard standing assumption that there exists a positive integer r such that the regulator equations

$$A_i \Pi_i + B_i \Gamma_i = \Pi_i S,$$

$$C_{2i} \Pi_i + D_{2i} \Gamma_i = R, \quad i = 1, 2, \dots, N$$
(7.2)

have solutions $\Pi_i \in \mathbb{R}^{n_i \times r}$, $\Gamma_i \in \mathbb{R}^{m_i \times r}$, $R \in \mathbb{R}^{p \times r}$ and $S \in \mathbb{R}^{r \times r}$, where the eigenvalues of *S* lie on the imaginary axis and the pair (R, S) is observable.

Following [115], we assume that the agents (7.1) should be interconnected by a protocol of the form

$$\dot{w}_{i} = A_{i}w_{i} + B_{i}u_{i} + G_{i}(y_{i} - C_{1i}w_{i}),$$

$$\dot{v}_{i} = Sv_{i} + \sum_{i=1}^{N} a_{ij}(v_{j} - v_{i}),$$

$$u_{i} = F_{i}(w_{i} - \Pi_{i}v_{i}) + \Gamma_{i}v_{i}, \quad i = 1, 2, \dots, N,$$

(7.3)

where $v_i \in \mathbb{R}^r$ and $w_i \in \mathbb{R}^{n_i}$ are the states of the *i*th local controller, the matrices *S*, Π_i and Γ_i are solutions of (7.2), and the matrices $F_i \in \mathbb{R}^{m_i \times n_i}$ and $G_i \in \mathbb{R}^{n_i \times r_i}$ are control gains to be designed. The coefficients a_{ij} are the entries of the adjacency matrix \mathcal{A} of the communication graph. Throughout this chapter it will be a standing assumption that the communication between the agents of the network is represented by a connected, simple undirected weighted graph. We briefly explain the structure of this protocol. The first equation in (7.3) has the structure of an asymptotic observer for the state of the *i*th agent. The second equation represents an auxiliary system associated with the *i*th agent. Each auxiliary system receives the relative state values with respect to its neighboring auxiliary systems. In this way, the network of auxiliary systems will reach state synchronization. The third equation in (7.3) is a static gain, it feeds back the value $w_i - \prod_i v_i$ and the state v_i of the associated auxiliary system to the *i*th agent. The idea of the protocol (7.3)is that, as time goes to infinity, the state x_i of the *i*th agent and its estimate w_i converge to $\Pi_i v_i$ due the first equation in (7.2). Subsequently, as a consequence of the second equation in (7.2), the outputs z_i of the agents will reach synchronization.

Denote by $\mathbf{x} = (x_1^{\top}, x_2^{\top}, \dots, x_N^{\top})^{\top}$ the aggregate state vector and likewise define **u**, **v**, **w**, **y**, **z** and **d**. Denote by *A* the block diagonal matrix

$$A = \text{blockdiag}(A_1, A_2, \dots, A_N) \tag{7.4}$$

and likewise define B, C_1 , C_2 , D_1 , D_2 and E. The multi-agent system (7.1) can then be written in compact form as

$$\dot{\mathbf{x}} = A\mathbf{x} + B\mathbf{u} + E\mathbf{d},$$

$$\mathbf{y} = C_1 \mathbf{x} + D_1 \mathbf{d},$$

$$\mathbf{z} = C_2 \mathbf{x} + D_2 \mathbf{u}.$$
(7.5)

Similarly, denote

98

 $F = blockdiag(F_1, F_2, \ldots, F_N)$

and likewise define G, Γ and Π . The protocol (7.3) can be written in compact form as

$$\dot{\mathbf{w}} = A\mathbf{w} + B\mathbf{u} + G(\mathbf{y} - C_1\mathbf{w}),$$

$$\dot{\mathbf{v}} = (I_N \otimes S - L \otimes I_r)\mathbf{v},$$

$$\mathbf{u} = F\mathbf{w} + (\Gamma - F\Pi)\mathbf{v}.$$
(7.6)

Next, denote

$$\mathbf{x}_o = (\mathbf{x}^\top, \mathbf{w}^\top, \mathbf{v}^\top)^\top.$$

By interconnecting the system (7.5) and the protocol (7.6), the controlled network

is then represented in compact form by

$$\begin{aligned} \dot{\mathbf{x}}_o &= A_o \mathbf{x}_o + E_o \mathbf{d}, \\ \mathbf{z} &= C_o \mathbf{x}_o, \end{aligned} \tag{7.7}$$

where

$$A_{o} = \begin{pmatrix} A & BF & B\Gamma - BF\Pi \\ GC_{1} & A + BF - GC_{1} & B\Gamma - BF\Pi \\ 0 & 0 & I_{N} \otimes S - L \otimes I_{r} \end{pmatrix},$$
$$C_{o} = \begin{pmatrix} C_{2} & D_{2}F & D_{2}\Gamma - D_{2}F\Pi \end{pmatrix}, \quad E_{o} = \begin{pmatrix} E \\ GD_{1} \\ 0 \end{pmatrix}.$$

Foremost, we want the protocol (7.3) to achieve output synchronization for the overall network:

Definition 7.1. The protocol (7.3) is said to achieve *z*-output synchronization for the network (7.7) if, for all i, j = 1, 2, ..., N, we have $z_i(t) - z_j(t) \rightarrow 0$, $v_i(t) - v_j(t) \rightarrow 0$ and $w_i(t) - w_j(t) \rightarrow 0$ as $t \rightarrow \infty$.

In the context of output synchronization, we are interested in the differences of the output values of the agents in the controlled network. Since the differences of the output values of communicating agents are captured by the incidence matrix R of the communication graph [63], we define a performance output variable as

$$\boldsymbol{\zeta} = (W^{\frac{1}{2}}R^{\top} \otimes I_p)\mathbf{z}_p$$

where *W* is the weight matrix defined in (1.2). The output ζ reflects the weighted disagreement between the outputs of the agents in accordance with the weights of the edges connecting these agents. Subsequently, we have the following equations for the controlled network

$$\begin{aligned} \dot{\mathbf{x}}_o &= A_o \mathbf{x}_o + E_o \mathbf{d}, \\ \mathbf{z} &= C_o \mathbf{x}_o, \\ \boldsymbol{\zeta} &= C_p \mathbf{x}_o, \end{aligned} \tag{7.8}$$

where

$$C_p = (W^{\frac{1}{2}}R^{\top} \otimes I_p)C_o$$

The impulse response matrix of the disturbance **d** to the performance output ζ is given by

$$T_d(t) = C_p e^{A_o t} E_o. aga{7.9}$$

The performance of the network is now quantified by the H_2 -norm of this impulse

response. Thus we define the associated \mathcal{H}_2 cost functional as

$$J := \int_0^\infty \operatorname{tr} \left[T_d^\top(t) T_d(t) \right] dt.$$
(7.10)

Note that the cost functional (7.10) is a function of the gain matrices F_1, F_2, \ldots, F_N and G_1, G_2, \ldots, G_N .

The \mathcal{H}_2 optimal output synchronization problem is now defined as the problem of minimizing the cost functional (7.10) over all protocols (7.3) that achieve output synchronization. Since the protocol (7.3) has a particular structure imposed by the communication topology, the \mathcal{H}_2 optimal output synchronization problem is a nonconvex optimization problem, and it is unclear whether a closed form solution exists in general. Therefore, in this chapter we will address a version of this problem that only requires *suboptimality*. The aim of this chapter is then to design a protocol of the form (7.3) that guarantees the associated cost (7.10) to be smaller than an a priori given upper bound while achieving **z**-output synchronization for the network. More concretely, the problem we will address is the following:

Problem 7.1. Let $\gamma > 0$ be a given tolerance. Design gain matrices F_1, F_2, \ldots, F_N and G_1, G_2, \ldots, G_N such that the resulting protocol (7.3) achieves *z*-output synchronization and its associated cost (7.10) satisfies $J < \gamma$.

To solve Problem 7.1, in the next section we will first review some preliminary results on \mathcal{H}_2 suboptimal control for linear systems and on output synchronization of heterogeneous linear multi-agent systems. It will become clear later on that these preliminary results are necessary ingredients to address Problem 7.1.

7.3 Preliminary results

7.3.1 \mathcal{H}_2 suboptimal control for linear systems by dynamic output feedback

In this subsection, we will review the \mathcal{H}_2 suboptimal control problem by dynamic output feedback for linear systems, see e.g. [95], [96], [100], [25] and [39]. In particular, we will review the results from [39] on separation principle based \mathcal{H}_2 suboptimal control for continuous-time linear systems.

Consider the system

$$\begin{aligned} \dot{x} &= \bar{A}x + \bar{B}u + \bar{E}d, \\ y &= \bar{C}_1 x + \bar{D}_1 d, \\ z &= \bar{C}_2 x + \bar{D}_2 u, \end{aligned} \tag{7.11}$$

where $x \in \mathbb{R}^n$ is the state, $u \in \mathbb{R}^m$ is the control input, $d \in \mathbb{R}^q$ is an unknown

external disturbance input, $y \in \mathbb{R}^r$ is the measured output, and $z \in \mathbb{R}^p$ is the output to be controlled. The matrices \overline{A} , \overline{B} , \overline{C}_1 , \overline{C}_2 , \overline{D}_1 , \overline{D}_2 and \overline{E} are of suitable dimensions. We assume that the pair $(\overline{A}, \overline{B})$ is stabilizable and the pair $(\overline{C}_1, \overline{A})$ is detectable. We consider dynamic output feedback controllers of the form

$$\dot{w} = \bar{A}w + \bar{B}u + G\left(y - \bar{C}_1w\right),$$

$$u = Fw,$$
(7.12)

where $w \in \mathbb{R}^n$ is the state of the controller, $F \in \mathbb{R}^{m \times n}$ and $G \in \mathbb{R}^{n \times r}$ are gain matrices to be designed. By interconnecting the controller (7.12) and the system (7.11), we obtain the controlled system

$$\begin{pmatrix} \dot{x} \\ \dot{w} \end{pmatrix} = \begin{pmatrix} \bar{A} & \bar{B}F \\ G\bar{C}_1 & \bar{A} + \bar{B}F - G\bar{C}_1 \end{pmatrix} \begin{pmatrix} x \\ w \end{pmatrix} + \begin{pmatrix} \bar{E} \\ G\bar{D}_1 \end{pmatrix} d,$$

$$z = (\bar{C}_2 \quad \bar{D}_2F) \begin{pmatrix} x \\ w \end{pmatrix}.$$

$$(7.13)$$

Denote

$$A_e = \begin{pmatrix} \bar{A} & \bar{B}F \\ G\bar{C}_1 & \bar{A} + \bar{B}F - G\bar{C}_1 \end{pmatrix}, \quad C_e = \begin{pmatrix} \bar{C}_2 & \bar{D}_2F \end{pmatrix}, \quad E_e = \begin{pmatrix} \bar{E} \\ G\bar{D}_1 \end{pmatrix}.$$

The impulse response matrix of the disturbance *d* to the output *z* is given by $T_{F,G}(t) = C_e e^{A_e t} E_e$. We define the \mathcal{H}_2 cost functional as

$$J(F,G) := \int_0^\infty \operatorname{tr}\left[T_{F,G}^{\top}(t)T_{F,G}(t)\right] dt.$$
(7.14)

The \mathcal{H}_2 suboptimal control problem by dynamic output feedback is the problem of finding a controller of the form (7.12) such that the associated cost (7.14) is smaller than an a priori given upper bound and the controlled system (7.13) is internally stable. The following lemma provides a design method for computing such a controller, see also [39, Theorem 4].

Lemma 7.1. Let $\gamma > 0$ be a given tolerance. Assume that $\overline{D}_1 \overline{E}^\top = 0$, $\overline{D}_2^\top \overline{C}_2 = 0$ and $\overline{D}_1 \overline{D}_1^\top = I_r$, $\overline{D}_2^\top \overline{D}_2 = I_m$. Let P > 0 and Q > 0 satisfy the Riccati inequalities

$$\bar{A}^{\top}P + P\bar{A} - P\bar{B}\bar{B}^{\top}P + \bar{C}_{2}^{\top}\bar{C}_{2} < 0,$$

$$\bar{A}Q + Q\bar{A}^{\top} - Q\bar{C}_{1}^{\top}\bar{C}_{1}Q + \bar{E}\bar{E}^{\top} < 0.$$

If, in addition, such P and Q satisfy

$$\operatorname{tr}\left(\bar{C}_{1}QPQ\bar{C}_{1}^{\top}\right)+\operatorname{tr}\left(\bar{C}_{2}Q\bar{C}_{2}^{\top}\right)<\gamma,$$

then the controller (7.12) with $F = -\overline{B}^{\top}P$ and $G = Q\overline{C}_1^{\top}$ internally stabilizes the system (7.11) and is suboptimal, i.e. $J(F,G) < \gamma$.

For a proof of Lemma 7.1, we refer to [39, Theorem 4].

7.3.2 Output synchronization of heterogeneous linear multi-agent systems

In this subsection, we will review some relevant results on output synchronization of heterogeneous linear multi-agent systems, see also [115], [22], [46] and [59].

Consider a heterogeneous linear multi-agent system consisting of N possibly distinct agents. The dynamics of the *i*th agent is represented by the linear time-invariant system

$$\dot{x}_{i} = A_{i}x_{i} + B_{i}u_{i},
y_{i} = C_{1i}x_{i}, \qquad i = 1, 2, \dots, N.
z_{i} = C_{2i}x_{i} + D_{2i}u_{i},$$
(7.15)

The agents (7.15) will be interconnected by a protocol of the form (7.3), where the matrices S, Γ_i and Π_i are assumed to satisfy the regulator equations (7.2). The multi-agent system (7.15) can be written in compact form as

$$\dot{\mathbf{x}} = A\mathbf{x} + B\mathbf{u},$$

$$\mathbf{y} = C_1 \mathbf{x},$$

$$\mathbf{z} = C_2 \mathbf{x} + D_2 \mathbf{u},$$

$$(7.16)$$

and the protocol (7.3) can be written as (7.6). By interconnecting the system (7.16) and the protocol (7.6), the controlled network is then given by

$$\begin{aligned} \dot{\mathbf{x}}_o &= A_o \mathbf{x}_o, \\ \mathbf{z} &= C_o \mathbf{x}_o. \end{aligned} \tag{7.17}$$

The following lemma yields conditions under which the controlled network (7.17) achieves **z**-output synchronization.

Lemma 7.2. Consider the multi-agent system (7.15) and the protocol (7.3). Let gain matrices F_i and G_i be such that the matrices $A_i + B_iF_i$ and $A_i - G_iC_{1i}$ are Hurwitz. Then the associated protocol (7.3) achieves *z*-output synchronization for the network.

A proof of Lemma 7.2 can be given along the lines of the proof of [115, Theorem 5].

We are now ready to deal with the H_2 suboptimal output synchronization problem formulated in Problem 7.1.

7.4 H_2 suboptimal output synchronization protocols by dynamic output feedback

In this section, we will resolve Problem 7.1. More specifically, we will establish a design method for computing gain matrices F_1, F_2, \ldots, F_N and G_1, G_2, \ldots, G_N such that the associated protocol (7.3) achieves **z**-output synchronization and guarantees $J < \gamma$.

In the sequel, we will first show that this problem can be simplified by transforming it into \mathcal{H}_2 suboptimal control problems for N auxiliary systems. The suboptimal gains F_i and G_i for these N separate problems will turn out to also yield a suboptimal protocol for the heterogeneous network.

To this end, we introduce the following N auxiliary systems

$$\dot{\xi}_{i} = A_{i}\xi_{i} + B_{i}\nu_{i} + E_{i}\delta_{i},
\vartheta_{i} = C_{1i}\xi_{i} + D_{1i}\delta_{i},
\eta_{i} = C_{2i}\xi_{i} + D_{2i}\nu_{i}, \quad i = 1, 2, ..., N,$$
(7.18)

where $\xi_i \in \mathbb{R}^{n_i}$ is the state, $\nu_i \in \mathbb{R}^{m_i}$ is the coupling input, $\delta_i \in \mathbb{R}^{q_i}$ is an unknown external disturbance input, $\vartheta_i \in \mathbb{R}^{r_i}$ is the measured output and $\eta_i \in \mathbb{R}^p$ is the output to be controlled. For given gain matrices F_i and G_i , consider the dynamic output feedback controllers

$$\dot{\omega}_i = A_i \omega_i + B_i \nu_i + G_i (\vartheta_i - C_{1i} \omega_i),$$

$$\nu_i = F_i \omega_i, \quad i = 1, 2, \dots, N,$$
(7.19)

where $\omega_i \in \mathbb{R}^n$ is the state of the *i*th controller.

By interconnecting the systems (7.18) and the controllers (7.19), we obtain the N controlled auxiliary systems

$$\begin{pmatrix} \dot{\xi}_i \\ \dot{\omega}_i \end{pmatrix} = \begin{pmatrix} A_i & B_i F_i \\ G_i C_{1i} & A_i + B_i F_i - G_i C_{1i} \end{pmatrix} \begin{pmatrix} \xi_i \\ \omega_i \end{pmatrix} + \begin{pmatrix} E_i \\ G_i D_{1i} \end{pmatrix} \delta_i,$$

$$\eta_i = \begin{pmatrix} C_{2i} & D_{2i} F_i \end{pmatrix} \begin{pmatrix} \xi_i \\ \omega_i \end{pmatrix}, \quad i = 1, 2, \dots, N.$$

$$(7.20)$$

For $i = 1, 2, \ldots, N$, denote

$$\bar{A}_i = \begin{pmatrix} A_i & B_i F_i \\ G_i C_{1i} & A_i + B_i F_i - G_i C_{1i} \end{pmatrix}, \quad \bar{C}_i = \begin{pmatrix} C_{2i} & D_{2i} F_i \end{pmatrix}, \quad \bar{E}_i = \begin{pmatrix} E_i \\ G_i D_{1i} \end{pmatrix}.$$

The impulse response matrix of the disturbance δ_i to the output η_i is equal to

$$T_{\delta i}(t) = \bar{C}_i e^{A_i t} \bar{E}_i,$$

and an associated \mathcal{H}_2 cost functional is defined as

$$J_i = \int_0^\infty \operatorname{tr}[T_{\delta i}^\top(t)T_{\delta i}(t)]dt.$$
(7.21)

The following lemma holds.

Lemma 7.3. Let $\gamma > 0$ be a given tolerance. Assume, for i = 1, 2, ..., N, the systems (7.20) are internally stable and the costs (7.21) satisfy

$$\sum_{i=1}^{N} J_i < \frac{\gamma}{\lambda_N},\tag{7.22}$$

where λ_N is the largest eigenvalue of the Laplacian matrix L. Then the protocol (7.3) achieves *z*-output synchronization for the network (7.8) and the associated cost (7.10) satisfies $J < \gamma$.

Proof. First, note that the systems (7.20) are internally stable if and only if the matrices $A_i + B_iF_i$ and $A_i - G_iC_{1i}$ are Hurwitz, see e.g. [102, Section 3.12]. Hence, by Lemma 7.2, if the systems (7.20) are internally stable, then the network controlled using the protocol (7.3) reaches **z**-output synchronization.

Next, we will show that if (7.22) holds, then $J < \gamma$. Note that (7.22) is equivalent to

$$\lambda_N \sum_{i=1}^N \int_0^\infty \operatorname{tr}[T_{\delta i}^\top(t) T_{\delta i}(t)] dt < \gamma.$$
(7.23)

In turn, the inequality (7.23) holds if and only if

$$\lambda_N \int_0^\infty \operatorname{tr}[\bar{T}_d^\top(t)\bar{T}_d(t)]dt < \gamma \tag{7.24}$$

holds, where

$$\bar{T}_d = \bar{C}_o e^{\bar{A}_o t} \bar{E}_o$$

with

$$\bar{A}_o = \begin{pmatrix} A & BF \\ GC_1 & A + BF - GC_1 \end{pmatrix}, \quad \bar{C}_o = \begin{pmatrix} C_2 & D_2F \end{pmatrix}, \quad \bar{E}_o = \begin{pmatrix} E \\ GD_1 \end{pmatrix}.$$

Recall that the matrix *A* is the block diagonal matrix defined in (7.4), similarly for the matrices *B*, *C*₁, *C*₂, *D*₁, *D*₂, *E*, *F* and *G*. Using the fact that $\lambda_N I_{pN} - L \otimes I_p \ge 0$,

it can be shown that (7.24) implies

$$\int_0^\infty \operatorname{tr}[\bar{T}_d^\top(t)(L\otimes I_p)\bar{T}_d(t)]dt < \gamma.$$
(7.25)

On the other hand,

$$\int_0^\infty \operatorname{tr}[\bar{T}_d^\top(t)(L \otimes I_p)\bar{T}_d(t)]dt = \int_0^\infty \operatorname{tr}\left[T_d^\top(t)T_d(t)\right]dt$$
(7.26)

with $T_d(t)$ given by (7.9). Note that the right hand side of (7.26) is exactly the cost J given by (7.10) associated with the network (7.8). It follows that $J < \gamma$. This completes the proof.

By the previous, if the gain matrices F_i and G_i are such that $A_i + B_iF_i$ and $A_i - G_iC_{1i}$ are Hurwitz and (7.22) holds, then the protocol (7.3) using these F_i and G_i yields **z**-output synchronization and $J < \gamma$. In the next theorem, we will provide a method for computing gain matrices F_i and G_i such that the above holds.

Theorem 7.4. Let $\gamma > 0$ be a given tolerance. For i = 1, 2, ..., N, assume that $D_{1i}E_i^{\top} = 0$, $D_{2i}^{\top}C_{2i} = 0$, $D_{1i}D_{1i}^{\top} = I_{r_i}$ and $D_{2i}^{\top}D_{2i} = I_{m_i}$. Let $P_i > 0$ satisfy

$$A_i^{\top} P_i + P_i A_i^{\top} - P_i B_i B_i^{\top} P_i + C_{2i}^{\top} C_{2i} < 0.$$
(7.27)

Let $Q_i > 0$ satisfy

$$A_{i}Q_{i} + Q_{i}A_{i}^{\top} - Q_{i}C_{1i}^{\top}C_{1i}Q_{i} + E_{i}E_{i}^{\top} < 0.$$
(7.28)

If, in addition, such P_i and Q_i satisfy

$$\operatorname{tr}(C_{1i}Q_iP_iQ_iC_{1i}^{\top}) + \operatorname{tr}(C_{2i}Q_iC_{2i}^{\top}) < \frac{\gamma}{N\lambda_N},\tag{7.29}$$

then the protocol (7.3) with $F_i := -B_i^{\top} P_i$ and $G_i := Q_i C_{1i}^{\top}$ achieves *z*-output synchronization for the network (7.8) and guarantees $J < \gamma$.

Proof. Note that (7.27) is equivalent to

$$(A_i - B_i B_i^{\top} P_i)^{\top} P_i + (A_i - B_i B_i^{\top} P_i) + P_i B_i B_i^{\top} P_i + C_{2i}^{\top} C_{2i} < 0$$
(7.30)

and (7.28) is equivalent to

$$(A_i - Q_i C_{1i}^{\top} C_{1i})Q_i + Q_i (A_i - Q_i C_{1i}^{\top} C_{1i})^{\top} + Q_i C_{1i}^{\top} C_{1i} Q_i + E_i E_i^{\top} < 0.$$
(7.31)

Taking $F_i := -B_i^\top P_i$ and $G_i := Q_i C_{1i}^\top$, it then follows that $A_i + B_i F_i$ and $A_i - G_i C_{1i}$ are Hurwitz.

Next, by (7.29), it follows from Lemma 7.1 that

$$J_i < \frac{\gamma}{N\lambda_N}, \quad i = 1, 2, \dots, N.$$

Thus we have (7.22), and the conclusion then follows from Lemma 7.3. \Box

We note that the conditions $D_{1i}E_i^{\top} = 0$, $D_{2i}^{\top}C_{2i} = 0$, $D_{1i}D_{1i}^{\top} = I_{r_i}$ and $D_{2i}^{\top}D_{2i} = I_{m_i}$ are made here to simplify notation, and can be relaxed to the regularity conditions $D_{1i}D_{1i}^{\top} > 0$ and $D_{2i}^{\top}D_{2i} > 0$ alone.

Remark 7.5. In Theorem 7.4, in order to select γ , the following steps could be taken. For i = 1, 2..., N:

- (i) Compute positive definite solutions P_i and Q_i of the Riccati inequalities (7.27) and (7.28). Such solutions exist.
- (ii) Denote $S_i = \operatorname{tr}(C_{1i}Q_iP_iQ_iC_{1i}^{\top}) + \operatorname{tr}(C_{2i}Q_iC_{2i}^{\top}).$
- (iii) Choose γ such that $N\lambda_N S_i < \gamma$.

Note that the smaller S_i or λ_N is, the smaller such feasible γ is allowed to be. Unfortunately, the problem of minimizing S_i over all $P_i > 0$ and $Q_i > 0$ that satisfy (7.27) and (7.28) is a non-convex optimization problem. However, since smaller Q_i leads to smaller $\text{tr}(C_{2i}Q_iC_{2i}^{\top})$ and smaller P_i and Q_i lead to smaller $\text{tr}(C_{1i}Q_iP_iQ_iC_{1i}^{\top})$, and consequently smaller feasible γ , we could try to find P_i and Q_i as small as possible. In fact, one can find $P_i = P_i(\epsilon_i) > 0$ to (7.27) by solving the Riccati equation

$$A_i^{\top} P_i + P_i A_i^{\top} - P_i B_i B_i^{\top} P_i + C_{2i}^{\top} C_{2i} + \epsilon_i I_{n_i} = 0$$

with $\epsilon_i > 0$ arbitrary. Similarly, one can find $Q_i = Q_i(\sigma_i) > 0$ to (7.28) by solving the dual Riccati equation

$$A_{i}Q_{i} + Q_{i}A_{i}^{\top} - Q_{i}C_{1i}^{\top}C_{1i}Q_{i} + E_{i}E_{i}^{\top} + \sigma_{i}I_{n_{i}} = 0$$

with $\sigma_i > 0$ arbitrary. By using a standard argument, it can be shown that $P_i(\epsilon_i)$ and $Q_i(\sigma_i)$ decrease as ϵ_i and σ_i decrease, respectively. So ϵ_i and σ_i should be taken close to 0 to get smaller P_i and Q_i .

7.5 H_2 suboptimal output synchronization protocols by state feedback

In this section, we consider a special case of Problem 7.1, namely, the case that full state information of the agents is available. More specifically, we consider the H_2 suboptimal output synchronization problem by relative state feedback for heterogeneous linear multi-agent systems.

Consider a heterogeneous multi-agent system consisting of N possibly distinct agents. The dynamics of the *i*th agent is represented by

$$\dot{x}_i = A_i x_i + B_i u_i + E_i d_i,
z_i = C_i x_i + D_i u_i,
i = 1, 2, \dots, N,$$
(7.32)

where $x_i \in \mathbb{R}^{n_i}$ is the state, $u_i \in \mathbb{R}^{m_i}$ the control input, $d_i \in \mathbb{R}^{q_i}$ the external disturbance and $z_i \in \mathbb{R}^p$ the output to be synchronized. The matrices A_i , B_i , C_i , D_i and E_i are of suitable dimensions. We assume that the pairs (A_i, B_i) are stabilizable. Also, as before, we assume that, for some positive integer r, the regulator equations

$$A_i \Pi_i + B_i \Gamma_i = \Pi_i S,$$

$$C_i \Pi_i + D_i \Gamma_i = R, \quad i = 1, 2, \dots, N$$
(7.33)

have solutions $\Pi_i \in \mathbb{R}^{n_i \times r}$, $\Gamma_i \in \mathbb{R}^{m_i \times r}$, $S \in \mathbb{R}^{r \times r}$ and $R \in \mathbb{R}^{p \times r}$, where all eigenvalues of S are on the imaginary axis and the pair (R, S) is observable.

Following [115] and [43], the agents will be interconnected by a protocol of the form

$$\dot{v}_i = Sv_i + \sum_{i=1}^N a_{ij}(v_j - v_i),$$

$$u_i = K_i(x_i - \Pi_i v_i) + \Gamma_i v_i, \quad i = 1, 2, \dots, N,$$
(7.34)

where $v_i \in \mathbb{R}^r$ is the state of the *i*th local controller, the matrices Π_i and Γ_i are solutions of (7.33) and the matrices $K_i \in \mathbb{R}^{m_i \times n_i}$ are control gains to be designed. The coefficients a_{ij} are the entries of the adjacency matrix \mathcal{A} of the communication graph.

Denote by $\mathbf{x} = (x_1^{\top}, x_2^{\top}, \dots, x_N^{\top})^{\top}$ the aggregate state vector and likewise define **u**, **z** and **d**. As before, denote by *A* the block diagonal matrix

$$A = \operatorname{blockdiag}(A_1, A_2, \ldots, A_N)$$

and likewise define B, C, D and E. The multi-agent system (7.32) can be written

in compact form as

$$\dot{\mathbf{x}} = A\mathbf{x} + B\mathbf{u} + E\mathbf{d},$$

$$\mathbf{z} = C\mathbf{x} + D\mathbf{u}.$$
 (7.35)

Denote

$$K = \mathsf{blockdiag}(K_1, K_2, \dots, K_N)$$

and likewise define Γ and Π . The protocol (7.34) can be written in compact form as

$$\dot{\mathbf{v}} = (I_N \otimes S - L \otimes I_r) \mathbf{v},
\mathbf{u} = K \mathbf{x} + (\Gamma - K \Pi) \mathbf{v}.$$
(7.36)

Denote

$$\mathbf{x}_s = (\mathbf{x}^{\top}, \mathbf{v}^{\top})^{\top}.$$

By interconnecting the system (7.35) and the protocol (7.36), the controlled network is given by

$$\begin{aligned} \dot{\mathbf{x}}_s &= A_s \mathbf{x}_s + E_s \mathbf{d}, \\ \mathbf{z} &= C_s \mathbf{x}_s, \end{aligned} \tag{7.37}$$

where

$$A_s = \begin{pmatrix} A + BK & B\Gamma - BK\Pi \\ 0 & I \otimes S - L \otimes I_r \end{pmatrix}, \ C_s = \begin{pmatrix} C + DK & D\Gamma - DK\Pi \end{pmatrix}, \ E_s = \begin{pmatrix} E \\ 0 \end{pmatrix}.$$

Similar to Definition 7.1, we have the following definition for output synchronization by relative state feedback:

Definition 7.2. The protocol (7.34) is said to achieve *z*-output synchronization for the network (7.37) if, whenever the disturbance is equal to zero, i.e., d = 0, then for all i, j = 1, 2, ..., N we have $z_i(t) - z_j(t) \rightarrow 0$ and $v_i(t) - v_j(t) \rightarrow 0$ as $t \rightarrow \infty$.

By introducing the performance output

$$\boldsymbol{\zeta}_s = (W^{\frac{1}{2}} R^\top \otimes I_p) \mathbf{z},$$

we have the following equations for the controlled network

$$\begin{aligned} \dot{\mathbf{x}}_s &= A_s \mathbf{x}_s + E_s \mathbf{d}, \\ \mathbf{z} &= C_s \mathbf{x}_s, \\ \boldsymbol{\zeta}_s &= C_c \mathbf{x}_s, \end{aligned} \tag{7.38}$$

where

$$C_c = (W^{\frac{1}{2}} R^\top \otimes I_p) C_s.$$

The impulse response from the disturbance **d** to the performance output ζ is

then given by $T_s(t) = C_c e^{A_s t} E_s$. Subsequently, we define the associated \mathcal{H}_2 cost functional as

$$J := \int_0^\infty \operatorname{tr}\left[T_s^\top(t)T_s(t)\right] dt.$$
(7.39)

Note that the cost functional (7.39) is a function of the gain matrices K_1, K_2, \ldots, K_N . The problem we want to address in this section is the following:

Problem 7.2. Let $\gamma > 0$ be a given tolerance. Design gain matrices K_1, K_2, \ldots, K_N such that the associated protocol (7.34) achieves *z*-output synchronization and $J(K) < \gamma$.

In the following proposition, we provide a design method for computing a protocol (7.34) such that the controlled network (7.38) achieves **z**-output synchronization and the associated cost is smaller than an a priori given upper bound.

Proposition 7.6. Let $\gamma > 0$ be a given tolerance. For i = 1, 2..., N, assume that $D_i^{\top} C_i = 0$ and $D_i^{\top} D_i = I_{m_i}$. Let $P_i > 0$ satisfy

$$A_{i}^{\top}P_{i} + P_{i}A_{i}^{\top} - P_{i}B_{i}B_{i}^{\top}P_{i} + C_{i}^{\top}C_{i} < 0.$$
(7.40)

If, in addition, such P_i *satisfy*

$$\operatorname{tr}(E_i^{\top} P_i E_i) < \frac{\gamma}{N\lambda_N},\tag{7.41}$$

then the protocol (7.34) with $K_i := -B_i^\top P_i$ achieves *z*-output synchronization for the network (7.38) and guarantees $J < \gamma$.

A proof of Proposition 7.6 can be given along the lines of the proof of Theorem 7.4, and is hence omitted here. The conditions $D_i^{\top} D_i = I_{m_i}$ are made here to simplify notation, and can be relaxed to the regularity conditions $D_i^{\top} D_i > 0$ alone.

7.6 Simulation example

In this section, we will give a simulation example based on the example in [115] to illustrate the design method of Theorem 7.4.

Consider a network of N = 6 heterogeneous agents. The dynamics of the agents are given by

$$\dot{x}_i = A_i x_i + B_i u_i + E_i d_i,$$

 $y_i = C_{1i} x_i + D_{1i} d_i,$ $i = 1, 2, \dots, 6,$
 $z_i = C_{2i} x_i + D_{2i} u_i,$

where
$$A_i = \begin{pmatrix} 0 & 1 & 0 \\ 0 & 0 & c_i \\ 0 & -f_i & -a_i \end{pmatrix}$$
, $B_i = \begin{pmatrix} 0 \\ 0 \\ b_i \end{pmatrix}$, $E_i = \begin{pmatrix} 0 & 0.2 \\ 0 & 0 \\ 0 & 0.2 \end{pmatrix}$, $C_{1i} = \begin{pmatrix} 1 & 0 & 0 \end{pmatrix}$,

 $D_{1i} = \begin{pmatrix} 1 & 0 \end{pmatrix}, C_{2i} = \begin{pmatrix} 1 & 1 & 0 \\ 0 & 0 & 0 \end{pmatrix}, D_{2i} = \begin{pmatrix} 0 \\ 1 \end{pmatrix}$. The parameters a_i, b_i, c_i and f_i are chosen to be

chosen to be

$$a_i = 2, c_i = 1, \quad i = 1, 2, \dots, 6,$$

 $b_1 = b_4 = 1, b_2 = b_5 = 2, b_3 = b_6 = 3,$
 $f_1 = f_4 = 1, f_2 = f_5 = 2, f_3 = f_6 = 3.$

The pairs (A_i, B_i) are stabilizable and the pairs (C_{1i}, A_i) are detectable. We also have that $D_{1i}E_i^{\top} = 0$, $D_{2i}^{\top}C_{2i} = 0$, $D_{1i}D_{1i}^{\top} = 1$ and $D_{2i}^{\top}D_{2i} = 1$. The communication graph between the six agents is assumed to be an undirected cycle graph. The largest eigenvalue of the corresponding Laplacian matrix L is $\lambda_6 = 4$.

We choose the matrices S and R in the regulator equations (7.2) to be

$$S = \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix}, \quad R = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}.$$

The eigenvalues of S are on the imaginary axis and the pair (R, S) is observable. We solve the equations (7.2) and compute

$$\Pi_i = \begin{pmatrix} 1 & 0 \\ 0 & 1 \\ 0 & 0 \end{pmatrix}, \quad \Gamma_i = \begin{pmatrix} 0 & 1 \end{pmatrix}, \quad i = 1, 2, \dots, 6.$$

The objective is to design a protocol of the form (7.3) such that the associated cost (7.10) satisfies $J < \gamma$ while achieving **z**-output synchronization. Let the desired upper bound be $\gamma = 18$.

Following the design method in Theorem 7.4, for i = 1, 2, ..., 6, we compute a positive definite solution P_i to (7.27) by solving the Riccati equation

$$A_i^{\top} P_i + P_i A_i^{\top} - P_i B_i B_i^{\top} P_i + C_{2i}^{\top} C_{2i} + \epsilon I_{n_i} = 0$$

with $\epsilon = 0.001$. We also compute a positive definite solution Q_i to (7.27) by solving the dual Riccati equation

$$A_{i}Q_{i} + Q_{i}A_{i}^{\top} - Q_{i}C_{1i}^{\top}C_{1i}Q_{i} + E_{i}E_{i}^{\top} + \sigma I_{n_{i}} = 0$$

with $\sigma = 0.001$. Accordingly, we compute the associated gain matrices F_i and G_i



Figure 7.1: Plots of trajectories of the first component of the output vectors z_1, z_2, \ldots, z_6

to be

$$F_1 = F_4 = \begin{pmatrix} -1.0005 & -1.7329 & -0.7326 \end{pmatrix},$$

$$F_2 = F_5 = \begin{pmatrix} -1.0005 & -1.2345 & -0.4951 \end{pmatrix},$$

$$F_3 = F_6 = \begin{pmatrix} -1.0005 & -1.0327 & -0.3982 \end{pmatrix},$$

and

$$G_1 = G_4 = \begin{pmatrix} 0.3290 & 0.0341 & 0.0028 \end{pmatrix}^{\top},$$

$$G_2 = G_5 = \begin{pmatrix} 0.2804 & 0.0193 & 0.0007 \end{pmatrix}^{\top},$$

$$G_3 = G_6 = \begin{pmatrix} 0.2578 & 0.0132 & 0.0002 \end{pmatrix}^{\top}.$$

As an example, we take the initial states of the agents to be $x_{10} = \begin{pmatrix} 1.0 & 1.4 & 1.6 \end{pmatrix}^{\top}$, $x_{20} = \begin{pmatrix} 1.2 & -1.7 & 0.5 \end{pmatrix}^{\top}$, $x_{30} = \begin{pmatrix} 1.3 & -1.2 & 1.3 \end{pmatrix}^{\top}$, $x_{40} = \begin{pmatrix} 0.6 & 1.6 & -1.3 \end{pmatrix}^{\top}$, $x_{50} = \begin{pmatrix} 1.8 & 1.5 & 1.6 \end{pmatrix}^{\top}$, $x_{60} = \begin{pmatrix} -1.1 & 1.7 & 0.9 \end{pmatrix}^{\top}$. We take the initial states w_i to be zero, and the initial states v_i to be $v_{10} = \begin{pmatrix} 0.9 & 1.1 \end{pmatrix}^{\top}$, $v_{20} = \begin{pmatrix} 0.8 & 1.4 \end{pmatrix}^{\top}$,



Figure 7.2: Plots of trajectories of the second component of the output vectors z_1, z_2, \ldots, z_6

 $v_{30} = \begin{pmatrix} -1.0 & 0.9 \end{pmatrix}^{\top}$, $v_{40} = \begin{pmatrix} 1.8 & 1.1 \end{pmatrix}^{\top}$, $v_{50} = \begin{pmatrix} -1.6 & 1.4 \end{pmatrix}^{\top}$, $v_{60} = \begin{pmatrix} 1.1 & -1.2 \end{pmatrix}^{\top}$. In Figures 7.1 and 7.2, we have plotted the trajectories of the output vectors z_i , i = 1, 2..., 6 of the controlled network. The proposed protocol indeed achieves **z**-output synchronization for the network.

Moreover, for $i = 1, 2, \ldots, 6$, we compute

$$S_i = \operatorname{tr}(C_{1i}Q_iP_iQ_iC_{1i}^{\top}) + \operatorname{tr}(C_{2i}Q_iC_{2i}^{\top}),$$

and obtain that

$$S_1 = S_4 = 0.6621, S_2 = S_5 = 0.4379, S_3 = S_6 = 0.3637.$$

Note that, for all $i = 1, 2, \ldots, 6$, we have

$$S_i < \frac{\gamma}{N\lambda_N} = 0.75,$$

it then follows from Theorem 7.4 that the designed protocol is suboptimal, i.e. the associated cost is indeed smaller than the desired tolerance $\gamma = 18$.

7.7 Conclusions

In this chapter, we have studied the \mathcal{H}_2 suboptimal output synchronization problem for heterogeneous linear multi-agent systems. Given a heterogeneous multiagent system and an associated \mathcal{H}_2 cost functional, we have provided a design method for computing dynamic output feedback based protocols that guarantee the associated cost to be smaller than a given upper bound while the controlled network achieves output synchronization. For each agent, its two local control gains are given in terms of solutions of two Riccati inequalities, each of dimension equal to that of the agent dynamics. The computation of the local control gains involves the largest eigenvalue of the Laplacian matrix of the communication graph. We have also considered the special case that full state information of the agents is available.

This chapter investigates the \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filtering problems for continuous time linear systems. Consider a linear system monitored by a number of local filters, where each of the filters receives only part of the measured output of the system. Each filter can communicate with the other filters according to an a priori given strongly connected weighted directed graph. The aim is to design filter gains that guarantee the \mathcal{H}_2 or \mathcal{H}_∞ norm of the transfer matrix from the disturbance input to the output estimation error to be smaller than an a priori given upper bound, while all local filters reconstruct the full system state asymptotically. We provide a centralized design method for obtaining such suboptimal distributed \mathcal{H}_2 and \mathcal{H}_∞ filters. The proposed design method is illustrated by a simulation example.

8.1 Introduction

Recent years have witnessed an increasing interest in problems of state estimation for spatially constrained large-scale systems. Such problems are relevant in applications, such as power grids [32], industrial plants [107] and wireless sensor networks [86]. Due to the physical constraints, the measured output of these systems is often monitored by a sensor network, consisting of a number of local sensors. Each of these local sensors makes use of its local measurements and then communicates with the other local sensors. In this way, all of these sensors together are able to estimate the state of the system asymptotically. In this problem setting, one of the main challenges is that none of the local sensors by itself is able to estimate the system state by using its own local measurements. Consequently, standard estimation methods do not directly apply anymore.

The distributed estimation problem has been mainly studied in two research directions, namely, distributed observer design and distributed Kalman filtering. In [88], an augmented state observer was proposed to cast the distributed observer design problem into a decentralized control problem for linear systems, using the

notion of 'fixed modes' [92]. Later on, in [113], the results in [88] were extended and a more general form of distributed observers was provided, allowing the rate of convergence of the observer to be freely assignable. In [114], for timevarying communication graphs, a hybrid observer was introduced to distributedly estimate the state of a linear system. Based on observability decompositions, the problem of distributed observer design was also investigated in [27, 49, 65]. In [66], an attack resilient algorithm was introduced to address the distributed estimation problem when certain nodes are compromised by adversaries.

On the other hand, much attention in the literature has also been devoted to distributed filtering problems. A Kalman-filter-based distributed filter was proposed in [81, 83, 84]. There, the proposed methods employ a two-step strategy: a state update rule based on a Kalman-filter and a data fusion step based on consensus. In [105], a distributed robust filtering problem was addressed using dissipativity theory. Later on in [106], the results of [105] were generalized to the case that the communication graph is allowed to randomly change. Recently, in [47], a distributed Kalman-Bucy filtering problem was studied, using the idea of averaging the dynamics of heterogeneous multi-agent systems [48].

Different from the existing work, in this chapter, we will consider two deterministic versions of the distributed optimal filtering problem for linear systems, i.e., the distributed \mathcal{H}_2 and \mathcal{H}_∞ filtering problems. Given a linear system and a network of local filters, each local filter receives a portion of the measured output of the system and then exchanges its state with that of its neighboring local filters. Together, these local filters form a distributed filter. We introduce \mathcal{H}_2 and \mathcal{H}_∞ performances to quantify the influence of the disturbances on the output estimation error. The distributed optimal filtering problem is then to find suitable filter gain matrices such that the associated \mathcal{H}_2 or \mathcal{H}_∞ performance is minimized, while the states of all local filters asymptotically track the system state. However, due to *non-convexity*, this problem is difficult to solve in general. Therefore, in this chapter we will address a *suboptimality* version of this problem. The objective of this chapter is then to design suitable filter gain matrices such that the \mathcal{H}_2 or \mathcal{H}_∞ performance is maller than an a priori given tolerance. The main contributions of this chapter are the following:

- 1. We establish conditions for the existence of suitable filter gains in terms of solvability of LMI's for both the \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filtering problem. For the \mathcal{H}_2 filtering problem, all except one of these LMI's will always turn out to be solvable.
- 2. We provide conceptual algorithms for obtaining suitable distributed \mathcal{H}_2 and \mathcal{H}_∞ suboptimal filters, respectively.

This chapter is organized as follows. In Section 8.2, we review some basic results on graph theory, detectability properties of linear systems, and the H_2 and

 \mathcal{H}_{∞} performance of linear systems. Subsequently, in Section 8.3 we formulate the \mathcal{H}_2 and \mathcal{H}_{∞} suboptimal distributed filtering problems. We then provide design methods for obtaining such distributed filters in Section 8.4. In Section 8.5 we provide a simulation example to illustrate our design method. Finally, in Section 8.6 we formulate our conclusions.

8.2 Preliminaries

8.2.1 Detectability and detectability decomposition

In this subsection, we will review detectability and the detectability decomposition of linear systems. Consider the linear system

$$\begin{aligned} \dot{x} &= Ax, \\ y &= Cx, \end{aligned} \tag{8.1}$$

where $x \in \mathbb{R}^n$ represents the state and $y \in \mathbb{R}^p$ the measured output. The matrices *A* and *C* are of suitable dimensions.

Let p(s) be the characteristic polynomial of A. Then p(s) can be factorized as

$$p(s) = p_-(s)p_+(s),$$

where $p_{-}(s)$ and $p_{+}(s)$ have roots in the open left half-plane and the closed right half-plane, respectively. The undetectable subspace of the pair (C, A) is defined as

$$\mathcal{S} := \mathcal{N} \cap \ker (p_+(A)),$$

where

$$\mathcal{N} := \ker \begin{pmatrix} C \\ CA \\ \vdots \\ CA^{n-1} \end{pmatrix}$$

The pair (C, A) is detectable if and only if $S = \{0\}$, see e.g. [116].

There exists an orthogonal matrix $T \in \mathbb{R}^{n \times n}$ such that the pair (C, A) is transformed into the detectability decomposition form

$$T^{\top}AT = \begin{pmatrix} A_{11} & 0 \\ A_{21} & A_{22} \end{pmatrix}, \quad CT = \begin{pmatrix} C_1 & 0 \end{pmatrix},$$

where $A_{11} \in \mathbb{R}^{v \times v}$, $A_{21} \in \mathbb{R}^{(n-v) \times v}$, $A_{22} \in \mathbb{R}^{(n-v) \times (n-v)}$, $C_1 \in \mathbb{R}^{p \times v}$ and the pair (C_1, A_{11}) is detectable. In addition, if we partition $T = (T_1 T_2)$, where T_1 contains

the first v columns, then the undetectable subspace is given by

$$\operatorname{im}(T_2) = \mathcal{S}.$$

Since T is orthogonal, we also have

$$\operatorname{im}(T_1) = \mathcal{S}^{\perp}.$$

8.2.2 \mathcal{H}_2 and \mathcal{H}_∞ performance of linear systems

In this subsection, we will review the \mathcal{H}_2 and \mathcal{H}_∞ performance of a linear system with external disturbances. Consider the linear system

$$\begin{aligned} \dot{x} &= Ax + Ed, \\ y &= Cx, \end{aligned} \tag{8.2}$$

where $x \in \mathbb{R}^n$ is the state, $d \in \mathbb{R}^q$ the external disturbance and $y \in \mathbb{R}^p$ the measured output. The matrices A, C and E are of suitable dimensions.

We will first review the \mathcal{H}_2 performance of the system (8.2). Let $T_d(t) = Ce^{At}E$ be the impulse response of (8.2). Then the associated \mathcal{H}_2 performance is defined to be the square of its \mathcal{L}_2 -norm, given by

$$J = \int_0^\infty \operatorname{tr} \left[T_d^\top(t) T_d(t) \right] dt.$$
(8.3)

Note that the performance (8.3) is finite if the system (8.2) is internally stable, i.e., A is Hurwitz.

The following well-known result provides a necessary and sufficient condition under which (8.2) is internally stable and (8.3) is smaller than a given upper bound (see e.g. [33], [96]):

Lemma 8.1. Let $\gamma > 0$. Then the system (8.2) is internally stable and $J < \gamma$ if and only if there exists P > 0 satisfying

$$A^{\top}P + PA + C^{\top}C < 0,$$

tr $(E^{\top}PE) < \gamma.$

Next, we will review the \mathcal{H}_{∞} performance of the system (8.2). Let $T_d(s) = C(sI_n - A)^{-1}E$ be the transfer matrix of (8.2). If A is Hurwitz, then the \mathcal{H}_{∞} performance of (8.2) is defined as the \mathcal{H}_{∞} norm of $T_d(s)$, given by

$$||T_d||_{\infty} := \sup_{\omega \in \mathbb{R}} \sigma(T(j\omega)), \tag{8.4}$$

where $\sigma(T_d(j\omega))$ is the maximum singular value of the complex matrix $T_d(j\omega)$.

The well-known bounded real lemma provides a necessary and sufficient condition under which (8.2) is stable and (8.4) is smaller than a given upper bound (see e.g. [124], [102]):

Lemma 8.2. Let $\gamma > 0$. Then the system (8.2) is internally stable and $||T_d||_{\infty} < \gamma$ if and only if there exists P > 0 such that

$$A^{\top}P + PA + \frac{1}{\gamma^2} PEE^{\top}P + C^{\top}C < 0.$$

In the next section, we will formulate the \mathcal{H}_2 and \mathcal{H}_∞ distributed filter design problems that will be addressed in this chapter.

8.3 **Problem formulation**

Consider the finite-dimensional linear time-invariant system

$$\begin{aligned} \dot{x} &= Ax + Ed, \\ y &= Cx + Dd, \\ z &= Hx, \end{aligned} \tag{8.5}$$

where $x \in \mathbb{R}^n$ is the state, $d \in \mathbb{R}^q$ the external disturbance, $y \in \mathbb{R}^r$ the measured output and $z \in \mathbb{R}^p$ the output to be estimated. The matrices *A*, *C*, *D*, *E* and *H* are of suitable dimensions.

The standard optimal filtering problem for the system (8.5) is to find a filter that takes y as input and returns an optimal estimate ζ of z, while the filter state asymptotically tracks the state x of (8.5). Here, 'optimal' means that the \mathcal{H}_2 or \mathcal{H}_∞ norm of the transfer matrix from d to the estimation error $z - \zeta$ is minimized over all such filters. In that problem setting, however, a standing assumption is that one single filter is able to acquire the complete measured output y of the system.

In this chapter, we relax this assumption. More specifically, we assume that the measured output y of (8.5) is not available to one single filter, but is observed by N local filters. Moreover, each local filter only acquires a certain portion of the measured output, namely,

$$y_i = C_i x + D_i d,$$

where $y_i \in \mathbb{R}^{r_i}$, $C_i \in \mathbb{R}^{r_i \times n}$ and $D_i \in \mathbb{R}^{r_i \times q}$, for i = 1, 2, ..., N. Here, the matrices

 C_i and D_i are obtained by partitioning

$$C = \begin{pmatrix} C_1 \\ C_2 \\ \vdots \\ C_N \end{pmatrix}, \quad D = \begin{pmatrix} D_1 \\ D_2 \\ \vdots \\ D_N \end{pmatrix}$$

Clearly, the original output y of (8.5) has then been partitioned as

$$y = \begin{pmatrix} y_1 \\ y_2 \\ \vdots \\ y_N \end{pmatrix}$$

and $\sum_{i=1}^{N} r_i = r$. In this chapter it will be a standing assumption that the pair (C, A) is detectable. We will also assume that none of the pairs (C_i, A) is detectable itself. If, for at least one *i*, the pair (C_i, A) is detectable, the distributed filtering problem boils down to the standard optimal filtering problem.

In our distributed case, each local filter makes use of the portion of the measured output that it acquires and will then communicate with its neighboring local filters by exchanging filter state information. In this way, the local filters will together form a distributed filter. Following [27] and [49], we propose a distributed filter of the form

$$\dot{w}_{i} = Aw_{i} + G_{i}(y_{i} - C_{i}w_{i}) + F_{i}\sum_{j=1}^{N} a_{ij}(w_{j} - w_{i}),$$

$$\zeta_{i} = Hw_{i}, \quad i = 1, 2, \dots, N,$$
(8.6)

where $w_i \in \mathbb{R}^n$ is the state of the *i*th local filter and $\zeta_i \in \mathbb{R}^p$ is the associated output. The matrices $G_i \in \mathbb{R}^{n \times r_i}$ and $F_i \in \mathbb{R}^{n \times n}$ are local filter gains to be designed. The coefficients a_{ij} are the entries of the adjacency matrix \mathcal{A} of the communication graph. In this chapter, it will be a standing assumption that this graph is a strongly connected weighted directed graph.

For the *i*th local filter, we introduce the associated local state estimation error e_i and local output estimation error η_i as

$$e_i := x - w_i,$$

 $\eta_i := z - \zeta_i, \quad i = 1, 2, ..., N.$

The dynamics of the *i*th local error system is then given by

$$\dot{e}_{i} = (A - G_{i}C_{i})e_{i} + F_{i}\sum_{j=1}^{N}a_{ij}(e_{j} - e_{i}) + (E - G_{i}D_{i})d,$$

$$\eta_{i} = He_{i}, \quad i = 1, 2, \dots, N.$$

Denote $e = (e_1^{\top}, e_2^{\top}, \dots, e_N^{\top})^{\top}$, $\eta = (\eta_1^{\top}, \eta_2^{\top}, \dots, \eta_N^{\top})^{\top}$ and

$$\bar{A} := \text{blockdiag}(A - G_i C_i) \in \mathbb{R}^{nN \times nN},$$

$$\bar{F} := \text{blockdiag}(F_i) \in \mathbb{R}^{nN \times nN},$$

$$\bar{E} := \text{col}(E - G_i D_i) \in \mathbb{R}^{nN \times q}.$$

The global error system is then given by

$$\dot{e} = \left(\bar{A} - \bar{F}(L \otimes I_n)\right)e + \bar{E}d,$$

$$\eta = (I_N \otimes H)e,$$
(8.7)

where $L \in \mathbb{R}^{N \times N}$ is the Laplacian matrix of the communication graph. The impulse response of the system (8.7) from the disturbance *d* to the output estimation error η is equal to

$$T_d(t) = (I_N \otimes H)e^{(\bar{A} - \bar{F}(L \otimes I_n))t}\bar{E}.$$

We introduce the global \mathcal{H}_2 cost functional

$$J = \int_0^\infty \operatorname{tr} \left[T_d^\top(t) T_d(t) \right] dt.$$
(8.8)

The distributed \mathcal{H}_2 optimal filtering problem is then the problem of minimizing the \mathcal{H}_2 cost functional (8.8) over all distributed filters (8.6) such that the global error system (8.7) is internally stable. Note that (8.8) is a function of the local gain matrices F_1, F_2, \ldots, F_N and G_1, G_2, \ldots, G_N .

Unfortunately, due to the particular form of (8.6), this optimization problem is, in general, non-convex and it is unclear whether a closed-form solution exists. Therefore, instead of trying to find an *optimal* solution, we will address a version of this problem that only requires *suboptimality*. More concretely, we aim at designing a distributed filter such that the error system (8.7) is internally stable and the \mathcal{H}_2 performance (8.8) is smaller than an a priori given tolerance γ . In that case, we say that the distributed filter (8.6) is $\mathcal{H}_2 \gamma$ -suboptimal:

Definition 8.1. Let $\gamma > 0$. The distributed filter (8.6) is called $\mathcal{H}_2 \gamma$ -suboptimal if:

1. for all i = 1, 2, ..., N, whenever d = 0, we have that $\lim_{t\to\infty} (x(t) - w_i(t)) \to 0$ for all initial conditions on (8.5) and (8.6).

2. the associated performance (8.8) satisfies $J < \gamma$.

Correspondingly, the \mathcal{H}_2 suboptimal distributed filtering problem that we will address is the following:

Problem 8.1. Let $\gamma > 0$. For i = 1, 2, ..., N, find gain matrices $G_i \in \mathbb{R}^{n \times r_i}$ and $F_i \in \mathbb{R}^{n \times n}$ such that the distributed filter (8.6) is $\mathcal{H}_2 \gamma$ -suboptimal.

In addition to the distributed filtering problem with \mathcal{H}_2 performance, in this chapter we will also consider the version of this problem with \mathcal{H}_∞ performance. Obviously, the transfer matrix of the system (8.7) from the disturbance *d* to the output estimation error η is equal to

$$T_d(s) = (I_N \otimes H) \left(sI_{nN} - (\bar{A} - \bar{F}(L \otimes I_n)) \right)^{-1} \bar{E}.$$

The \mathcal{H}_{∞} performance of the distributed filter (8.6) is given by the \mathcal{H}_{∞} norm $||T_d||_{\infty}$ of $T_d(s)$. The problem that we will then consider is to design a distributed filter (8.6) such that the error system (8.7) is internally stable and its \mathcal{H}_{∞} performance is smaller than an a priori given tolerance γ . In that case, we say that the distributed filter (8.6) is $\mathcal{H}_{\infty} \gamma$ -suboptimal:

Definition 8.2. Let $\gamma > 0$. The distributed filter (8.6) is called $\mathcal{H}_{\infty} \gamma$ -suboptimal if:

- 1. for all i = 1, 2, ..., N, whenever d = 0, we have that $\lim_{t\to\infty} (x(t) w_i(t)) \to 0$ for all initial conditions on (8.5) and (8.6).
- 2. $||T_d||_{\infty} < \gamma$.

Correspondingly, the \mathcal{H}_{∞} suboptimal distributed filtering problem that we will address is the following:

Problem 8.2. Let $\gamma > 0$. For i = 1, 2, ..., N, find gain matrices $G_i \in \mathbb{R}^{n \times r_i}$ and $F_i \in \mathbb{R}^{n \times n}$ such that the distributed filter (8.6) is $\mathcal{H}_{\infty} \gamma$ -suboptimal.

8.4 \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filter design

In this section, we will address Problems 8.1 and 8.2 introduced above and provide design methods for obtaining suboptimal distributed filters.

As we have explained before, the *i*th local filter (8.6) receives only a certain portion of the measured output, namely,

$$y_i = C_i x + D_i d, \quad i = 1, 2, \dots, N.$$

In order to proceed, we first apply orthogonal transformations to the pairs (C_i, A) . For i = 1, 2, ..., N, let T_i be an orthogonal matrix such that the pair (C_i, A) is transformed into the detectability decomposition form

$$T_i^{\top} A T_i = \begin{pmatrix} A_{i11} & 0 \\ A_{i21} & A_{i22} \end{pmatrix}, \quad C_i T_i = \begin{pmatrix} C_{i1} & 0 \end{pmatrix},$$
(8.9)

where $A_{i11} \in \mathbb{R}^{v_i \times v_i}$, $A_{i21} \in \mathbb{R}^{(n-v_i) \times v_i}$, $A_{i22} \in \mathbb{R}^{(n-v_i) \times (n-v_i)}$, $C_{i1} \in \mathbb{R}^{r_i \times v_i}$ and the pair (C_{i1}, A_{i11}) is detectable. The integer v_i is equal to the dimension of the othogonal complement of the undetectable subspace of the pair (C_i, A) . Accordingly, partition

$$T_i^{\top} E = \begin{pmatrix} E_{i1} \\ E_{i2} \end{pmatrix}, \quad HT_i = \begin{pmatrix} H_{i1} & H_{i2} \end{pmatrix}, \quad (8.10)$$

where $E_{i1} \in \mathbb{R}^{v_i \times q}$, $E_{i2} \in \mathbb{R}^{(n-v_i) \times q}$, $H_{i1} \in \mathbb{R}^{p \times v_i}$ and $H_{i1} \in \mathbb{R}^{p \times (n-v_i)}$.

Using the fact that (C_{i1}, A_{i11}) is detectable, let Q_{i1} be any positive definite solution to

$$A_{i11}Q_{i1} + Q_{i1}A_{i11}^{\top} - Q_{i1}C_{i1}^{\top}C_{i1}Q_{i1} < 0.$$
(8.11)

Then, by defining

$$G_{i1} := Q_{i1} C_{i1}^{\top}, \tag{8.12}$$

the matrix $A_{i11} - G_{i1}C_{i1}$ is Hurwitz.

In the sequel, we will make use of the transformed matrices (8.9) and (8.10) and the gain matrix (8.12) to obtain filter gains that solve Problems 8.1 and 8.2. Before presenting the main results of this chapter, we will first provide a lemma that will be essential for later use. This lemma is a generalization of [27, Lemma 4], and connects the Laplacian matrix of the communication graph with detectability properties of the system (8.5).

Lemma 8.3. Let $\mathcal{L} := \Theta L + L^{\top}\Theta$, where Θ is defined as in Lemma 1.2. Define T := blockdiag $(T_i) \in \mathbb{R}^{nN \times nN}$, where the T_i are the orthogonal matrices introduced in (8.9) and (8.10). Let $m_i > 0$ and

$$M_i := \begin{pmatrix} m_i I_{v_i} & 0\\ 0 & 0_{n-v_i} \end{pmatrix}, \quad i = 1, 2, \dots, N.$$

Define $M := blockdiag(M_i)$. Then,

$$T^{\top}(\mathcal{L} \otimes I_n)T + M > 0.$$
(8.13)

Proof. Note that the inequality (8.13) holds if and only if the inequality

$$\mathcal{L} \otimes I_n + TMT^+ > 0 \tag{8.14}$$

holds. Let $U \in \mathbb{R}^{N \times N}$ be an orthogonal matrix that diagonalizes the matrix \mathcal{L} , i.e.,

$$U^{\top}\mathcal{L}U = \Lambda = \operatorname{diag}(0, \lambda_2, \dots, \lambda_N).$$

Then, the inequality (8.14) holds if and only if

$$(U \otimes I_n)(\Lambda \otimes I_n)(U \otimes I_n)^\top + TMT^\top > 0$$
(8.15)

holds. The inequality (8.15) holds if

$$(U \otimes I_n) \left(\lambda_2 I_{nN} - \begin{pmatrix} \lambda_2 I_n & 0\\ 0 & 0_{n(N-1)} \end{pmatrix} \right) (U \otimes I_n)^\top + TMT^\top > 0$$
(8.16)

holds, where we use the fact that $\lambda_2 \leq \lambda_i$ for i = 2, 3, ..., N. Note that (8.16) is equal to

$$\lambda_2 I_{nN} - (U \otimes I_n) \begin{pmatrix} \lambda_2 I_n & 0\\ 0 & 0_{n(N-1)} \end{pmatrix} (U \otimes I_n)^\top + TMT^\top > 0.$$
(8.17)

Next, recall that

$$U = \begin{pmatrix} \frac{1}{\sqrt{N}} \mathbf{1}_N & U_2 \end{pmatrix}$$
 and $U^{\top} = \begin{pmatrix} \frac{1}{\sqrt{N}} \mathbf{1}_N^{\top} \\ U_2^{\top} \end{pmatrix}$,

the inequality (8.17) is equivalent to

$$\lambda_2 I_{nN} - \frac{\lambda_2}{N} (\mathbf{1}_N \otimes I_n) (\mathbf{1}_N \otimes I_n)^\top + TMT^\top > 0.$$
(8.18)

Now, by pre- and post-multiplying T^{\top} and T, the inequality (8.18) holds if and only if

$$\lambda_2 I_{nN} + M - \frac{\lambda_2}{N} \begin{pmatrix} T_1 & \dots & T_N \end{pmatrix}^\top \begin{pmatrix} T_1 & \dots & T_N \end{pmatrix} > 0.$$
(8.19)

Notice that, according to detectability decompositions, the matrix T_i can be partitioned as $T_i = (T_{i1} \ T_{i2})$ with $T_{i1}T_{i1}^{\top} + T_{i2}T_{i2}^{\top} = I_n$, for i = 1, 2, ..., N. By applying Schur complement, the inequality (8.19) holds if and only if

$$\begin{pmatrix} R_1 & \dots & 0 & T_1^\top \\ \vdots & \ddots & \vdots & \vdots \\ 0 & \dots & R_N & T_N^\top \\ T_1 & \dots & T_N & \frac{N}{\lambda_2} I_n \end{pmatrix} > 0$$
(8.20)

holds, where

$$R_i = \begin{pmatrix} (\lambda_2 + m_i)I_{v_i} & 0\\ 0 & \lambda_2 I_{n-v_i} \end{pmatrix}, \quad i = 1, 2, \dots, N.$$

Again by using Schur complement, the inequality (8.20) is equivalent to

$$\begin{pmatrix} \lambda_2 I_{n-v_1} & 0 & \dots & 0 & T_{12}^{\top} \\ 0 & R_2 & \dots & 0 & T_2^{\top} \\ \vdots & \vdots & \ddots & \vdots & & \vdots \\ 0 & 0 & \dots & R_N & T_N^{\top} \\ T_{12} & T_2 & \dots & T_N & \frac{N}{\lambda_2} I_n - \frac{1}{\lambda_2 + m_1} T_{11} T_{11}^{\top} \end{pmatrix} > 0.$$
 (8.21)

Now, by repeatedly applying Schur complement, we obtain that the inequality (8.19) holds if and only if

$$\frac{N}{\lambda_2}I_n - \sum_{i=1}^N \frac{1}{\lambda_2 + m_i} T_{i1}T_{i1}^\top - \sum_{i=1}^N \frac{1}{\lambda_2} T_{i2}T_{i2}^\top > 0.$$

holds. Since

$$\frac{N}{\lambda_2} I_n - \sum_{i=1}^N \frac{1}{\lambda_2 + m_i} T_{i1} T_{i1}^\top - \sum_{i=1}^N \frac{1}{\lambda_2} T_{i2} T_{i2}^\top$$
$$= \frac{N}{\lambda_2} I_n - \sum_{i=1}^N \frac{1}{\lambda_2} T_{i1} T_{i1}^\top - \sum_{i=1}^N \frac{1}{\lambda_2} T_{i2} T_{i2}^\top$$
$$+ \sum_{i=1}^N \frac{1}{\lambda_2} T_{i1} T_{i1}^\top - \sum_{i=1}^N \frac{1}{\lambda_2 + m_i} T_{i1} T_{i1}^\top$$
$$= \sum_{i=1}^N \left(\frac{1}{\lambda_2} - \frac{1}{\lambda_2 + m_i}\right) T_{i1} T_{i1}^\top$$

Let $m_{\min} = \min\{m_1, m_2, \dots, m_N\}$, then we have

$$\frac{N}{\lambda_2} I_n - \sum_{i=1}^N \frac{1}{\lambda_2 + m_i} T_{i1} T_{i1}^\top - \sum_{i=1}^N \frac{1}{\lambda_2} T_{i2} T_{i2}^\top$$

$$\geq \sum_{i=1}^N \left(\frac{1}{\lambda_2} - \frac{1}{\lambda_2 + m_{\min}} \right) T_{i1} T_{i1}^\top$$

$$= \left(\frac{1}{\lambda_2} - \frac{1}{\lambda_2 + m_{\min}} \right) \left(T_{11} \dots T_{N1} \right) \left(T_{11} \dots T_{N1} \right)^\top$$

Since $m_{\min} > 0$, then $\frac{1}{\lambda_2} - \frac{1}{\lambda_2 + m_{\min}} > 0$. Therefore, the inequality (8.13) holds if

 $\operatorname{rank} \begin{pmatrix} T_{11} & \ldots & T_{N1} \end{pmatrix} = n,$

i.e., the matrix $(T_{11} \ldots T_{N1})$ has full row rank. In the sequel, we will show that indeed rank $(T_{11} \ldots T_{N1}) = n$.

Define

$$\mathcal{N}_i := \ker \begin{pmatrix} C_i \\ C_i A \\ \vdots \\ C_i A^{n-1} \end{pmatrix} \in \mathbb{R}^n$$

and

$$\mathcal{S}_i := \mathcal{N}_i \cap \ker \left(p_+(A) \right),$$

where $p_+(A)$ is defined in Subsection 8.2.1. Note that

$$\operatorname{im}(T_{i1}) = \mathcal{S}_i^{\perp}.$$

Furthermore, if the pair (C, A) is detectable, then

$$\bigcap_{i=1}^{N} \mathcal{S}_i = 0.$$

Next, we find that

$$(\operatorname{im} (T_{11} \dots T_{N1}))^{\perp}$$

$$= (\operatorname{im}(T_{11}) + \dots + \operatorname{im}(T_{N1}))^{\perp}$$

$$= \bigcap_{i=1}^{N} (\operatorname{im}(T_{i1}))^{\perp}$$

$$= \bigcap_{i=1}^{N} S_{i}$$

$$= 0,$$

where the last step is due to the assumption that the pair (C, A) is detectable. It then follows from $(\operatorname{im}(T_{11} \ldots T_{N1}))^{\perp} = 0$ that rank $(T_{11} \ldots T_{N1}) = n$. Consequently, the inequality (8.3) holds. This completes the proof.

The lines of the proof of Lemma 8.3 is analogous to that of [27, Lemma 4], replacing the observability decomposition by the detectability decomposition. We provide the detailed proof here to make this chapter self-contained.

In the next two subsections, we will deal with the design of \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filters, respectively.

8.4.1 H_2 suboptimal distributed filter design

In this subsection, we will provide a design method for obtaining \mathcal{H}_2 suboptimal distributed filters. More specifically, we aim at finding a distributed filter such that the global error system (8.7) is stable and the associated \mathcal{H}_2 performance (8.8) is less than an a priori given tolerance.

The next lemma expresses the existence of suitable gain matrices F_i and G_i , i = 1, 2, ..., N in terms of solvability of LMI's.

Lemma 8.4. Let $\gamma > 0$. Let the matrices T, M and \mathcal{L} be as introduced in Lemma 8.3. Let $\epsilon > 0$ be such that

$$T^{\top}(\mathcal{L} \otimes I_n)T + M > \epsilon I_{nN}.$$
(8.22)

Let G_{i1} be as defined in (8.12). For i = 1, 2, ..., N, assume there exist $\kappa > 0$, $P_{i1} > 0$ and $P_{i2} > 0$ satisfying

$$\begin{pmatrix} \Phi_i + H_{i1}^\top H_{i1} + \kappa (m_i - \epsilon) I_{v_i} & A_{i21}^\top P_{i2} + H_{i1}^\top H_{i2} \\ P_{i2} A_{i21} + H_{i2}^\top H_{i1} & \Psi_i \end{pmatrix} < 0$$
(8.23)

and

$$\sum_{i=1}^{N} \operatorname{tr} \left[(E_{i1} - G_{i1}D_i)^{\top} P_{i1} (E_{i1} - G_{i1}D_i) + E_{i2}^{\top} P_{i2} E_{i2} \right] < \gamma,$$
(8.24)

where

$$\Phi_i := A_{i11}^{\top} P_{i1} + P_{i1} A_{i11} - C_{i1}^{\top} G_{i1}^{\top} P_{i1} - P_{i1} G_{i1} C_{i1}, \qquad (8.25)$$

$$\Psi_i := P_{i2}A_{i22} + A_{i22}^\top P_{i2} + H_{i2}^\top H_{i2} - \kappa \epsilon I_{n-v_i}.$$
(8.26)

For i = 1, 2, ..., N, define gain matrices F_i and G_i by

$$F_{i} := \kappa \theta_{i} T_{i} \begin{pmatrix} P_{i1}^{-1} & 0\\ 0 & P_{i2}^{-1} \end{pmatrix} T_{i}^{\top}$$
(8.27)

and

$$G_i := T_i \begin{pmatrix} G_{i1} \\ 0 \end{pmatrix}. \tag{8.28}$$

Then the corresponding distributed filter (8.6) is $\mathcal{H}_2 \gamma$ -suboptimal.

Proof. First, it follows from (8.13) in Lemma 8.3 that there exists $\epsilon > 0$ such that

(8.22) holds. Next, note that (8.23) is equivalent to

blockdiag
$$\begin{pmatrix} \Phi_{i} + H_{i1}^{\top}H_{i1} & A_{i21}^{\top}P_{i2} + H_{i1}^{\top}H_{i2} \\ P_{i2}A_{i21} + H_{i2}^{\top}H_{i1} & P_{i2}A_{i22} + A_{i22}^{\top}P_{i2} + H_{i2}^{\top}H_{i2} \end{pmatrix} + \kappa (M - \epsilon I_{nN}) < 0.$$
 (8.29)

Using (8.22), it follows from (8.29) that

blockdiag
$$\begin{pmatrix} \Phi_{i} + H_{i1}^{\top}H_{i1} & A_{i21}^{\top}P_{i2} + H_{i1}^{\top}H_{i2} \\ P_{i2}A_{i21} + H_{i2}^{\top}H_{i1} & P_{i2}A_{i22} + A_{i22}^{\top}P_{i2} + H_{i2}^{\top}H_{i2} \end{pmatrix}$$

 $-\kappa T^{\top}(\mathcal{L} \otimes I_{n})T < 0.$ (8.30)

Let

$$P := \operatorname{blockdiag}(P_i), \quad P_i := T_i \begin{pmatrix} P_{i1} & 0\\ 0 & P_{i2} \end{pmatrix} T_i^{\top}.$$
(8.31)

Clearly, P > 0. By using (8.27), (8.28), (8.31), (8.9) and (8.10), then (8.30) holds if and only if

$$\bar{A}^{\top}P + P\bar{A} - (L^{\top} \otimes I_n)\bar{F}^{\top}P + P\bar{F}(L \otimes I_n) + I_N \otimes H^{\top}H < 0$$
(8.32)

holds, where \overline{F} := blockdiag(F_i) and F_i is defined by (8.27). Therefore, there exist $\kappa > 0$, $P_{i1} > 0$ and $P_{i2} > 0$ such that (8.23) holds for i = 1, 2, ..., N if and only if there exists P > 0 of the form (8.31) such that (8.32) holds. Since the solutions of (8.23) also satisfy (8.24), we obtain

$$\operatorname{tr}\left(\bar{E}^{\top}P\bar{E}\right) < \gamma. \tag{8.33}$$

Finally, since (8.32) and (8.33) have a solution P > 0, it follows from Lemma 8.1 that the error system (8.7) is internally stable and $J < \gamma$. Thus the distributed filter (8.6) with (8.28) and (8.27) is $\mathcal{H}_2 \gamma$ -suboptimal.

Remark 8.5. In Lemma 8.4, the choice of the parameters $m_i > 0$ is arbitrary. The parameter $\epsilon > 0$ should be chosen sufficiently small so that (8.22) holds. The gain G_i is defined by (8.28). Then, of course, the question arises: for chosen $m_i > 0$, $\epsilon > 0$ and G_i , how can we find the smallest $\gamma > 0$ such that the corresponding distributed filter (8.6) is $\mathcal{H}_2 \gamma$ -suboptimal? This requires to find the smallest γ such that the LMI's (8.23) and (8.24) are solvable. It is well known that this can be done by using a standard bisection algorithm, see e.g. [124, page 115].

Remark 8.6. Lemma 8.4 states that if there exist solutions $\kappa > 0$, $P_{i1} > 0$ and $P_{i2} > 0$ satisfying (8.23) and (8.24), then the distributed filter (8.6) with gain matrices (8.27) and (8.28) is $\mathcal{H}_2 \gamma$ -suboptimal. There, the inequality (8.24) is a

global condition for checking suboptimality. In fact, such suboptimality condition can also be checked locally. Indeed, if for i = 1, 2, ..., N there exist solutions satisfying (8.23) and

$$\operatorname{tr}\left[(E_{i1} - G_{i1}D_i)^{\top} P_{i1}(E_{i1} - G_{i1}D_i) + E_{i2}^{\top} P_{i2}E_{i2}\right] < \frac{\gamma}{N},$$

then the corresponding distributed filter (8.6) with (8.27) and (8.28) is $\mathcal{H}_2 \gamma$ -suboptimal.

Lemma 8.4 provides a condition for the existence of suitable gain matrices F_i and G_i in terms of solvability of LMI's. In the next theorem, we show that, in fact, the LMI's (8.23) in Lemma 8.4 always have solutions. In fact, we can take P_{i2} to be the identity matrix of dimension $n - v_i$ and P_{i1} to be the unique solution of a given Lyapunov equation. In this way we obtain the following conceptual algorithm for computing suitable gain matrices.

Theorem 8.7. Let $\gamma > 0$. Then an $\mathcal{H}_2 \gamma$ -suboptimal distributed filter of the form (8.6) is obtained as follows:

(i) Compute $\theta = (\theta_1, \theta_2, \dots, \theta_N)$ with $\theta_i > 0$ such that $\theta L = 0$ and $\theta \mathbf{1}_N = N$.

Then, for i = 1, 2, ..., N:

- (ii) Compute orthogonal matrices T_i that put A, C_i, E and H into the form (8.9) and (8.10).
- (iii) Take $m_i = 1$ and compute $\epsilon > 0$ such that

$$T^{\top}(\mathcal{L} \otimes I_n)T + M > \epsilon I_{nN}.$$
(8.34)

- (iv) Compute $Q_{i1} > 0$ satisfying (8.11). Define $G_{i1} := Q_{i1}C_{i1}^{\top}$.
- (v) Take $\kappa > 0$ sufficiently large such that

$$A_{i22} + A_{i22}^{\top} + H_{i2}^{\top} H_{i2} - \kappa \epsilon I_{n-v_i} + \frac{1}{\kappa \epsilon} (A_{i21} + H_{i2}^{\top} H_{i1}) (A_{i21} + H_{i2}^{\top} H_{i1})^{\top} < 0.$$
(8.35)

(vi) Compute $P_{i1} > 0$ satisfying the Lyapunov equation

$$(A_{i11} - G_{i1}C_{i1})^{\top}P_{i1} + P_{i1}(A_{i11} - G_{i1}C_{i1}) + H_{i1}^{\top}H_{i1} + \kappa I_{v_i} = 0.$$
(8.36)

(vii) Define gain matrices F_i and G_i by

$$F_{i} := \kappa \theta_{i} T_{i} \begin{pmatrix} P_{i1}^{-1} & 0\\ 0 & I_{n-v_{i}} \end{pmatrix} T_{i}^{\top}, \quad G_{i} := T_{i} \begin{pmatrix} G_{i1}\\ 0 \end{pmatrix}.$$
(8.37)

Then for all $\gamma > 0$ satisfying

$$\sum_{i=1}^{N} \operatorname{tr} \left[(E_{i1} - G_{i1}D_i)^{\top} P_{i1} (E_{i1} - G_{i1}D_i) + E_{i2}^{\top} E_{i2} \right] < \gamma,$$
(8.38)

the corresponding distributed filter (8.6) with gain matrices (8.37) is $\mathcal{H}_2 \gamma$ -suboptimal.

Proof. Using Lemma 8.3, by choosing $m_i = 1$ for i = 1, 2, ..., N, there exists $\epsilon > 0$ such that (8.34) holds. Next, for i = 1, 2, ..., N, we choose $\kappa > 0$ sufficiently large such that (8.35) holds. Since Q_{i1} is a positive definite solution of (8.11) and $G_{i1} := Q_{i1}C_{i1}^{\top}$, then $A_{i11} - G_{i1}C_{i1}$ is Hurwitz. Consequently, for i = 1, 2, ..., N, the Lyapunov equation (8.36) has unique solution $P_{i1} > 0$. Since (8.35) holds and $-\kappa \epsilon I_{v_i} < 0$, by using the Schur complement, we obtain

$$\begin{pmatrix} -\kappa\epsilon I_{v_i} & A_{i21}^{\top} + H_{i1}^{\top}H_{i2} \\ A_{i21} + H_{i2}^{\top}H_{i1} & \tilde{\Psi}_i \end{pmatrix} < 0, \quad i = 1, 2, \dots, N,$$
(8.39)

where $\tilde{\Psi}_i := A_{i22} + A_{i22}^{\top} + H_{i2}^{\top} H_{i2} - \kappa \epsilon I_{n-v_i}$. Using (8.36) and $P_{i2} = I_{n-v_i}$, it then follows that (8.23) holds.

On the other hand, by taking $P_{i2} = I_{n-v_i}$ in (8.24), we obtain (8.38). It then follows from Lemma 8.4 that the corresponding distributed filter is $\mathcal{H}_2 \gamma$ -suboptimal.

Remark 8.8. Note that, in step (i) of Theorem 8.7, we need to compute the left eigenvector θ of the Laplacian matrix corresponding to the eigenvalue 0. This requires so-called global information on the communication graph. This dependency on global information can be removed using algorithms that compute left eigenvectors of the Laplacian matrix in a distributed fashion, see e.g. [12] or [24]. On the other hand, in step (iii) we need to compute ϵ . To do so, we need knowledge of the orthogonal matrices T_i , the matrix M and the Laplacian matrix \mathcal{L} , which is global information. Also in step (v), we need to find one κ that satisfy (8.35) for $i = 1, 2, \ldots, N$. Note that, however, we can always take $\epsilon > 0$ sufficiently small and $\kappa > 0$ sufficiently large such that (8.34) and (8.35) hold, respectively. This might however lead to an achievable tolerance γ that is very large, giving poor suboptimality of the corresponding distributed filter.

In general, the computation of our suboptimal filters requires global information, so cannot be performed in a decentralized fashion. This is in contrast with the decentralized computation of distributed state observers as described in [50].

8.4.2 \mathcal{H}_{∞} suboptimal distributed filter design

In this subsection, we will provide a method for obtaining \mathcal{H}_{∞} suboptimal distributed filters. More concretely, we aim at finding, for a given tolerance $\gamma > 0$, a distributed filter such that the global error system (8.7) is stable and $||T_d||_{\infty} < \gamma$.

The next lemma expresses the existence of suitable gain matrices F_i and G_i , i = 1, 2, ..., N in terms of solvability of N nonlinear matrix inequalities.

Lemma 8.9. Let $\gamma > 0$. Let the matrices T, M and \mathcal{L} be as introduced in Lemma 8.3. Let $\epsilon > 0$ be such that

$$T^{\top}(\mathcal{L} \otimes I_n)T + M > \epsilon I_{nN}.$$
(8.40)

Let G_{i1} be as defined in (8.12). For i = 1, 2, ..., N, assume there exist $\kappa > 0$, $P_{i1} > 0$ and $P_{i2} > 0$ satisfying

$$\begin{pmatrix} \Phi_i + \kappa (m_i - \epsilon) I_{v_i} & \Omega_i \\ \Omega_i^\top & \Psi_i - \kappa \epsilon I_{n-v_i} \end{pmatrix} < 0,$$
(8.41)

where

$$\Phi_{i} = (A_{i11} - G_{i1}^{\top}C_{i1})^{\top}P_{i1} + P_{i1}(A_{i11} - G_{i1}^{\top}C_{i1}) + \frac{1}{\gamma^{2}}P_{i1}(E_{i1} - G_{i1}D_{i})(E_{i1} - G_{i1}D_{i})^{\top}P_{i1} + H_{i1}^{\top}H_{i1},$$
(8.42)

$$\Omega_{i} = A_{i21}^{\top} P_{i2} + H_{i1}^{\top} H_{i2} + \frac{1}{\gamma^{2}} P_{i1} (E_{i1} - G_{i1} D_{i}) E_{i2}^{\top} P_{i2}, \qquad (8.43)$$

$$\Psi_i = P_{i2}A_{i22} + A_{i22}^{\top}P_{i2} + \frac{1}{\gamma^2}P_{i2}E_{i2}E_{i2}^{\top}P_{i2} + H_{i2}^{\top}H_{i2}$$

For i = 1, 2, ..., N, define gain matrices F_i and G_i by

$$F_{i} := \kappa \theta_{i} T_{i} \begin{pmatrix} P_{i1}^{-1} & 0\\ 0 & P_{i2}^{-1} \end{pmatrix} T_{i}^{\top}$$
(8.44)

and

$$G_i := T_i \begin{pmatrix} G_{i1} \\ 0 \end{pmatrix}. \tag{8.45}$$

Then the corresponding distributed filter (8.6) is $\mathcal{H}_{\infty} \gamma$ -suboptimal.

Proof. First, it follows from (8.13) in Lemma 8.3 that there exists $\epsilon > 0$ such that (8.22) holds. Next, note that (8.41) is equivalent to

blockdiag
$$\begin{pmatrix} \Phi_i & \Omega_i \\ \Omega_i^\top & \Psi_i \end{pmatrix} + \kappa (M - \epsilon I_{nN}) < 0.$$
 (8.46)
Using (8.22), it then follows from (8.46) that

blockdiag
$$\begin{pmatrix} \Phi_i & \Omega_i \\ \Omega_i^\top & \Psi_i \end{pmatrix} - \kappa T^\top (\mathcal{L} \otimes I_n) T < 0.$$
 (8.47)

Let

$$P := \operatorname{blockdiag}(P_i), \quad P_i := T_i \begin{pmatrix} P_{i1} & 0\\ 0 & P_{i2} \end{pmatrix} T_i^{\top}.$$
(8.48)

Clearly, P > 0. By using (8.44), (8.45), (8.48), (8.9) and (8.10), then (8.47) holds if and only if

$$\bar{A}^{\top}P + P\bar{A} - (L^{\top} \otimes I_n)\bar{F}^{\top}P - P\bar{F}(L \otimes I_n) + \frac{1}{\gamma^2}P\bar{E}\bar{E}^{\top}P + I_N \otimes H^{\top}H < 0$$
(8.49)

holds, where \overline{F} := blockdiag(F_i) and F_i is defined by (8.44). Therefore, there exist $\kappa > 0$, $P_{i1} > 0$ and $P_{i2} > 0$ such that (8.41) holds for i = 1, 2, ..., N if and only if there exists P > 0 of the form (8.31) such that (8.49) holds. Finally, since (8.49) has a solution P > 0, it follows from Lemma 8.2 that the error system (8.7) is internally stable and $||T_d||_{\infty} < \gamma$. Thus the distributed filter (8.6) with (8.45) and (8.44) is $\mathcal{H}_{\infty} \gamma$ -suboptimal.

Lemma 8.9 provides a condition for the existence of suitable gain matrices F_i and G_i in terms of solvability of the nonlinear matrix inequalities (8.41). However, these inequalities are not LMI's. However, by using suitable Schur complements, we can transform the inequalities (8.41) into LMI's. In this way we obtain the following conceptual algorithm for computing suitable gain matrices.

Theorem 8.10. Let $\gamma > 0$. Then an \mathcal{H}_{∞} suboptimal distributed filter of the form (8.6) is obtained as follows:

(i) Compute $\theta = (\theta_1, \theta_2, \dots, \theta_N)$ with $\theta_i > 0$ such that $\theta L = 0$ and $\theta \mathbf{1}_N = N$.

For i = 1, 2, ..., N:

- (ii) Compute an orthogonal matrix T_i that puts A, C_i, E and H into the form (8.9) and (8.10).
- (iii) Take arbitrary $m_i > 0$ and compute $\epsilon > 0$ such that

$$T^{\top}(\mathcal{L} \otimes I_n)T + M > \epsilon I_{nN}.$$
(8.50)

(iv) Compute $Q_{i1} > 0$ satisfying (8.11). Define $G_{i1} := Q_{i1}C_{i1}^{\top}$.

(v) Compute $P_{i1} > 0$, $P_{i2} > 0$ and $\kappa > 0$ such that the inequality

$$\begin{pmatrix} \Delta_i & A_{i21}^\top P_{i2} + H_{i1}^\top H_{i2} & P_{i1}(E_{i1} - G_{i1}D_i) \\ P_{i2}A_{i21} + H_{i2}^\top H_{i1} & P_{i2}A_{i22} + A_{i22}^\top P_{i2} & P_{i2}E_{i2} \\ (E_{i1} - G_{i1}D_i)^\top P_{i1} & E_{i2}^\top P_{i2} & -\gamma^2 I_q \end{pmatrix} < 0 \quad (8.51)$$

with $\Delta_i = (A_{i11} - G_{i1}^{\top}C_{i1})^{\top}P_{i1} + P_{i1}(A_{i11} - G_{i1}^{\top}C_{i1}) + \kappa(m_i - \epsilon)I_{v_i} + H_{i1}^{\top}H_{i1}$ holds.

(vi) Define gain matrices F_i and G_i by

$$F_{i} := \kappa \theta_{i} T_{i} \begin{pmatrix} P_{i1}^{-1} & 0\\ 0 & I_{n-v_{i}} \end{pmatrix} T_{i}^{\top}, \quad G_{i} := T_{i} \begin{pmatrix} G_{i1}\\ 0 \end{pmatrix}.$$
(8.52)

Then the corresponding distributed filter (8.6) is $\mathcal{H}_{\infty} \gamma$ -suboptimal.

Proof. By taking the appropriate Schur complements in (8.51), it follows that (8.51) hold if and only if (8.41) hold. The rest follows from Lemma 8.9. \Box

We conclude this section by noting that remarks similar to Remark 8.5 and Remark 8.8 hold in the \mathcal{H}_{∞} case.

8.5 Simulation example

In this section, we will use a simulation example borrowed from [49] to illustrate the conceptual algorithm in Theorem 8.7 for designing \mathcal{H}_2 suboptimal distributed filters. Consider the linear time-invariant system

$$\begin{aligned} \dot{x} &= Ax + Ed, \\ y &= Cx + Dd, \\ z &= Hx, \end{aligned} \tag{8.53}$$

where

$$A = \begin{pmatrix} 0 & 1 & 0 & 0 \\ -1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 2 \\ 0 & 0 & -2 & 0 \end{pmatrix}, \quad E = \begin{pmatrix} 0.1 \\ 0.1 \\ 0 \\ 0.1 \end{pmatrix}, \quad C = I_4,$$
$$D = \begin{pmatrix} 0.1 \\ 0 \\ 0.1 \\ 0.1 \end{pmatrix}, \quad H = \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & 2 & 0 & 0 \\ 0 & 0 & 0 & 1 \end{pmatrix}.$$



Figure 8.1: The communication graph between the local filters.

The system (8.53) is monitored by four local filters, and each local filter acquires a portion of the measured output y, namely,

$$y_i = C_i x + D_i d, \quad i = 1, 2, 3, 4,$$

where the matrices C_i and D_i are obtained by partitioning

$$C = \begin{pmatrix} \frac{C_1}{C_2} \\ \frac{C_3}{C_4} \end{pmatrix} = \begin{pmatrix} \frac{1 & 0 & 0 & 0}{0 & 1 & 0 & 0} \\ \frac{0 & 0 & 1 & 0}{0 & 0 & 0 & 1} \end{pmatrix}, \quad D = \begin{pmatrix} \frac{D_1}{D_2} \\ \frac{D_3}{D_4} \end{pmatrix} = \begin{pmatrix} \frac{0.1}{0} \\ \frac{0.1}{0.1} \\ \frac{0.1}{0.1} \end{pmatrix}.$$

The pair (C, A) is detectable but none of the pairs (C_i, A) is detectable.

We assume the four local filters to be of the form (8.6).

The communication graph between the four local filters is depicted in Figure 8.1. The graph is strongly connected and the associated Laplacian matrix is given by

$$L = \begin{pmatrix} 2 & -1 & 0 & -1 \\ 0 & 1 & -1 & 0 \\ -1 & 0 & 1 & 0 \\ -1 & 0 & 0 & 1 \end{pmatrix}.$$

The normalized left eigenvector θ of *L* associated with eigenvalue 0 is computed to be $\theta = \begin{pmatrix} 1 & 1 & 1 & 1 \end{pmatrix}$.

Next, for i = 1, 2, 3, 4, we compute an orthogonal matrix T_i such that the matrices A, C_i , E and H are transformed into the form (8.9) and (8.10). For i = 1, 2, 3, 4, we take $m_i = 1$. We also compute $\epsilon = 0.42$ such that (8.34) holds. Subsequently, for i = 1, 2, 3, 4, we solve (8.35) and compute $\kappa = 9.6$. Following the



Figure 8.2: Plots of trajectories of x_1 (dashed lines) and the corresponding filter state component (solid lines).



Figure 8.3: Plots of trajectories of x_2 (dashed lines) and the corresponding filter state component (solid lines).



Figure 8.4: Plots of trajectories of x_3 (dashed lines) and the corresponding filter state component (solid lines).



Figure 8.5: Plots of trajectories of x_4 (dashed lines) and the corresponding filter state component (solid lines).

steps in Theorem 8.7, gain matrices F_i and G_i are then computed as

$$F_{1} = \begin{pmatrix} 0.3636 & 0.0837 & 0 & 0 \\ 0.0837 & 0.3442 & 0 & 0 \\ 0 & 0 & 9.6000 & 0 \\ 0 & 0 & 0 & 9.6000 \end{pmatrix},$$

$$F_{2} = \begin{pmatrix} 0.3460 & -0.0660 & 0 & 0 \\ -0.0660 & 0.3579 & 0 & 0 \\ 0 & 0 & 9.6000 & 0 \\ 0 & 0 & 0 & 9.6000 \end{pmatrix},$$

$$F_{3} = \begin{pmatrix} 9.6000 & 0 & 0 & 0 \\ 0 & 9.6000 & 0 & 0 \\ 0 & 0 & 0.4274 & 0.0491 \\ 0 & 0 & 0.0491 & 0.4217 \end{pmatrix},$$

$$F_{4} = \begin{pmatrix} 9.6000 & 0 & 0 & 0 \\ 0 & 9.6000 & 0 & 0 \\ 0 & 9.6000 & 0 & 0 \\ 0 & 0 & 0.4220 & -0.0445 \\ 0 & 0 & -0.0445 & 0.4266 \end{pmatrix},$$

and

$$G_{1} = \begin{pmatrix} 0.4445\\ 0.0488\\ 0\\ 0 \end{pmatrix}, \quad G_{2} = \begin{pmatrix} -0.0488\\ 0.4445\\ 0\\ 0 \end{pmatrix},$$
$$G_{3} = \begin{pmatrix} 0\\ 0\\ 0.4465\\ 0.0248 \end{pmatrix}, \quad G_{4} = \begin{pmatrix} 0\\ 0\\ -0.0248\\ 0.4465 \end{pmatrix}.$$

As an example, we take the initial state of the system (8.53) to be $x_0 = \begin{pmatrix} 1 & -0.5 & -1 & 0 \end{pmatrix}^{\top}$ and the initial state of the distributed filter to be zero. In Figures 8.2, 8.3, 8.4 and 8.5, we have plotted the state trajectories of the system and that of the distributed filter in absence of external disturbances. It can be seen that the states of the local filters asymptotically track the state of the system (8.53). Moreover, we compute

$$\sum_{i=1}^{N} \operatorname{tr} \left[(E_{i1}^{\top} - D_{i}^{\top} G_{i1}^{\top}) P_{i1} (E_{i1} - G_{i1} D_{i}) + E_{i2}^{\top} E_{i2} \right] = 1.3717$$

Thus, for all $\gamma > 1.3717$, the distributed filter (8.6) with gain matrices F_i and G_i is $\mathcal{H}_2 \gamma$ -suboptimal.

8.6 Conclusions

In this chapter, we have studied the \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filtering problems for linear systems. We have established conditions for the existence of suitable filter gains. These are expressed in terms of solvability of LMI's. Based on these conditions, we have provided conceptual algorithms for obtaining \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filters, respectively.

9

Conclusions and future research

In this thesis, we have studied the distributed linear quadratic suboptimal control problem, the distributed \mathcal{H}_2 suboptimal control problem, and the \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filtering problems. In the present chapter we will discuss the main results and contributions presented in Chapters 2 - 8. We will also provide directions for possible future research.

9.1 Conclusions

In Chapter 2, we have studied a suboptimality version of the distributed linear quadratic optimal control for leaderless homogeneous multi-agent systems. Given a number of homogeneous agents, we have introduced a global linear quadratic cost functional. The communication graph between the agents is represented by a connected undirected graph. We have provided a *centralized* design method for computing distributed control laws, whose cost is bounded by a given upper bound for all initial states in a closed ball of a given radius, such that the controlled network achieves synchronization. The computation of the local control gain uses the exact knowledge of the smallest nonzero and largest eigenvalue of the Laplacian matrix. As an extension, we have also provided a design method for computing distributed control laws which does not depend on exact knowledge of the smallest nonzero and largest eigenvalue of the largest eigenvalue of the smallest nonzero and largest eigenvalue of the smallest nonzero and largest eigenvalue of the smallest nonzero and largest eigenvalue of the largest eigenval

In Chapter 3, we have extended the results in Chapter 2 on distributed linear quadratic optimal control for leaderless homogeneous multi-agent systems to the case of distributed linear quadratic optimal tracking control for leader-follower homogeneous multi-agent systems. Given one autonomous leader and a number of followers, we have introduced an associated global linear quadratic cost functional. We have established a *centralized* design method for computing distributed control laws, whose cost is smaller than an upper bound for all initial states bounded in a norm by given radius, such that the controlled followers reach tracking synchronization. The computation of the local control gain depends on

the smallest and largest eigenvalue of a given positive definite matrix associated with the communication graph between the agents.

The computation of the local control gains in Chapters 2 and 3 requires exactly knowledge of eigenvalues of the Laplacian matrix or of a given positive definite matrix associated with the communication graph, which is *global information*. Consequently, the design methods for obtaining such distributed control laws are in a *centralized* fashion. In Chapter 4, we have aimed at removing this dependence on global information. More specifically, we have considered the distributed linear quadratic optimal control problem for leaderless multi-agent systems with *single integrator* dynamics. We have first shown that the optimal local control gain can *only* be obtained in a centralized fashion, i.e. the computation of the local gain depends on global information. We have then established a *decentralized* computes its local control gain using sampled information of its neighboring agents. This decentralized computation leads to a suboptimal overall network behavior.

In Chapter 5, we have studied the distributed \mathcal{H}_2 suboptimal control problem for homogeneous multi-agent systems by *static relative state feedback*. Given a number of identical agents, an associated \mathcal{H}_2 cost functional and a connected weighted undirected communication graph interconnecting the agents. We have provided a centralized design method for obtaining distributed protocols such that the associated \mathcal{H}_2 cost is smaller than an a priori given upper bound while the controlled network achieves state synchronization.

In Chapter 6, we have generalized the results in Chapter 5 on the distributed \mathcal{H}_2 suboptimal control problem for homogeneous multi-agent systems by static state feedback to the case of *dynamic relative output feedback*. We have first solved an open problem of finding, for a single continuous-time linear system, a *separation principle* based \mathcal{H}_2 suboptimal dynamic output feedback controller. We have then made use of these results to establish a centralized design method for computing distributed \mathcal{H}_2 suboptimal protocols such that the controlled network reaches state synchronization.

In Chapter 7, we have further generalized the results in Chapters 5 and 6 on distributed \mathcal{H}_2 suboptimal control of homogeneous multi-agent systems to the case of *heterogeneous* multi-agent systems. For heterogeneous systems, since the agent dynamics is possibly non-identical, in particular the state space dimensions of the agents may differ. Therefore, instead of state synchronization as considered in Chapters 5 and 6, it is more interesting and natural to consider *output synchronization*. Given a heterogeneous multi-agent system and an \mathcal{H}_2 cost functional, we have proposed a centralized method for computing distributed protocols such that the associated \mathcal{H}_2 cost is smaller than an a priori given upper bound while the controlled agents reach output synchronization.

In Chapter 8, we have studied the \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filtering problem for linear systems. Given a linear system monitored by a number of local filters, where each local filter receives only a part of the system output according to a given strongly connected directed graph. We have established conceptual algorithms for obtaining \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filters, in the sense that the \mathcal{H}_2 or \mathcal{H}_∞ norm of the transfer matrix of the disturbance input to the output estimation error is smaller than an a priori given upper bound while all local filters reconstruct the system state asymptotically.

9.2 Future research

The results presented in this thesis can be extended in several directions.

- In this thesis, we have investigated *suboptimality* versions of the distributed linear quadratic optimal control problem, the distributed H₂ optimal control problem, and the distributed H₂ and H_∞ optimal filtering problem. Future research could investigate whether, in general, there exist closed form solutions for the genuine *optimal* control and filtering problems.
- The computation of the local control gains in Chapters 2, 3, 5 7 requires exact knowledge of the eigenvalues of the Laplacian matrix or of a given positive definite matrix associated with the communication graph, which is global information. As a possibility for future research, we mention the development of methods for *decentralized* computation of the control gains.
- In Chapter 4, the model of the agents is represented by single integrator dynamics. One possibility for future research is to generalize the results in Chapter 4 to the case of *general higher dimensional* agent dynamics. Another possibility for future research is to extend the results in Chapter 4, in which the time clock of sampling is synchronized for all agents, to the case that the sampling takes place in an *asynchronous* way, using results, for example, in [9, 58].
- Chapters 5 7 are concerned with the distributed H₂ suboptimal control problem. It would be interesting to extend the results in these chapters to that of distributed H_∞ suboptimal control.
- In Chapter 8, the computation of the distributed filters requires centralized computation. As a possibility for future research, we mention the extension of the results in this chapter to the case that filter gains are computed in a *decentralized* fashion, see for example [50]. Another possibility for future research is to allow the local filters to estimate, instead of the complete state, only part of the state of the monitored system.

Bibliography

- W. N. Anderson and T. D. Morley. Eigenvalues of the Laplacian of a graph. Linear and Multilinear Algebra, 18(2):141–145, 1985.
- [2] R. Aragues, G. Shi, D. V. Dimarogonas, C. Sagüés, K. H. Johansson, and Y. Mezouar. Distributed algebraic connectivity estimation for undirected graphs with upper and lower bounds. *Automatica*, 50(12):3253–3259, 2014.
- [3] S. Baldi and P. Frasca. Leaderless synchronization of heterogeneous oscillators by adaptively learning the group model. *IEEE Transactions on Automatic Control*, 65(1):412–418, 2020.
- [4] B. Besselink and K. H. Johansson. String stability and a delay-based spacing policy for vehicle platoons subject to disturbances. *IEEE Transactions on Automatic Control*, 62(9):4376–4391, 2017.
- [5] B. Besselink, V. Turri, S. H. van de Hoef, K.-Y. Liang, A. Alam, J. Mårtensson, and K. H. Johansson. Cyber–physical control of road freight transport. *Proceedings of the IEEE*, 104(5):1128–1141, 2016.
- [6] D. Biro, D. J.T. Sumpter, J. Meade, and T. Guilford. From compromise to leadership in pigeon homing. *Current Biology*, 16(21):2123–2128, 2006.
- [7] F. Borrelli and T. Keviczky. Distributed LQR design for identical dynamically decoupled systems. *IEEE Transactions on Automatic Control*, 53(8):1901–1912, 2008.
- [8] F. Bullo, J. Cortés, and S. Martínez. Distributed Control of Robotic Networks. Applied Mathematics Series. Princeton University Press, 2009. Electronically available at http://coordinationbook.info.
- [9] M. Cao, A. S. Morse, and B. D. O. Anderson. Reaching a consensus in a dynamically changing environment: convergence rates, measurement delays, and asynchronous events. *SIAM Journal on Control and Optimization*, 47(2):601–623, 2008.

- [10] Y. Cao and W. Ren. Optimal linear-consensus algorithms: an LQR perspective. *IEEE Transactions on Systems, Man, and Cybernetics, Part B (Cybernetics)*, 40(3):819–830, 2010.
- [11] Y. Cao, W. Yu, W. Ren, and G. Chen. An overview of recent progress in the study of distributed multi-agent coordination. *IEEE Transactions on Industrial Informatics*, 9(1):427–438, 2013.
- [12] T. Charalambous, M. G. Rabbat, M. Johansson, and C. N. Hadjicostis. Distributed finite-time computation of digraph parameters: left-eigenvector, out-degree and spectrum. *IEEE Transactions on Control of Network Systems*, 3 (2):137–148, 2016.
- [13] Y. Cheng and V. Ugrinovskii. Leader–follower tracking control with guaranteed consensus performance for interconnected systems with linear dynamic uncertain coupling. *International Journal of Control*, 88(8):1663–1677, 2015.
- [14] F. Cucker and S. Smale. Emergent behavior in flocks. *IEEE Transactions on Automatic Control*, 52(5):852–862, 2007.
- [15] O. Demir and J. Lunze. Optimal and event-based networked control of physically interconnected systems and multi-agent systems. *International Journal of Control*, 87(1):169–185, 2014.
- [16] P. Deshpande, P. P. Menon, C. Edwards, and I. Postlethwaite. A distributed control law with guaranteed LQR cost for identical dynamically coupled linear systems. In *Proc. American Control Conf. (ACC)*, pages 5342–5347, 2011.
- [17] F. Dörfler, M. Chertkov, and F. Bullo. Synchronization in complex oscillator networks and smart grids. *Proceedings of the National Academy of Sciences*, 110 (6):2005–2010, 2013.
- [18] S. Fattahi, G. Fazelnia, J. Lavaei, and M. Arcak. Transformation of optimal centralized controllers into near-globally optimal static distributed controllers. *IEEE Transactions on Automatic Control*, 64(1):66–80, 2019.
- [19] G. Fazelnia, R. Madani, A. Kalbat, and J. Lavaei. Convex relaxation for optimal distributed control problems. *IEEE Transactions on Automatic Control*, 62(1):206–221, 2017.
- [20] F. E. Fish. Kinematics of ducklings swimming in formation: consequences of position. *Journal of Experimental Zoology*, 273(1):1–11, 1995.
- [21] C. Godsil and G. F. Royle. Algebraic Graph Theory. Graduate Texts in Mathematics. Springer, 2001.

- [22] H. F. Grip, T. Yang, A. Saberi, and A. A. Stoorvogel. Output synchronization for heterogeneous networks of non-introspective agents. *Automatica*, 48(10): 2444–2453, 2012.
- [23] V. Gupta, B. Hassibi, and R. M. Murray. A sub-optimal algorithm to synthesize control laws for a network of dynamic agents. *International Journal of Control*, 78(16):1302–1313, 2005.
- [24] A. Gusrialdi and Z. Qu. Distributed estimation of all the eigenvalues and eigenvectors of matrices associated with strongly connected digraphs. *IEEE Control Systems Letters*, 1(2):328–333, 2017.
- [25] S. Haesaert, S. Weiland, and C. W. Scherer. A separation theorem for guaranteed H_2 performance through matrix inequalities. *Automatica*, 96:306–313, 2018.
- [26] W. Han, H. L. Trentelman, Z. Wang, and Y. Shen. Towards a minimal order distributed observer for linear systems. *Systems & Control Letters*, 114:59–65, 2018.
- [27] W. Han, H. L. Trentelman, Z. Wang, and Y. Shen. A simple approach to distributed observer design for linear systems. *IEEE Transactions on Automatic Control*, 64(1):329–336, 2019.
- [28] K. Hengster-Movric, F. L. Lewis, and M. Sebek. Distributed static outputfeedback control for state synchronization in networks of identical LTI systems. *Automatica*, 53:282–290, 2015.
- [29] Y. Hong, J. Hu, and L. Gao. Tracking control for multi-agent consensus with an active leader and variable topology. *Automatica*, 42(7):1177–1182, 2006.
- [30] R.A. Horn and C.R. Johnson. *Matrix Analysis*. Cambridge University Press, 1990.
- [31] G. Hu. Robust consensus tracking of a class of second-order multi-agent dynamic systems. *Systems & Control Letters*, 61(1):134–142, 2012.
- [32] Y. Huang, S. Werner, J. Huang, N. Kashyap, and V. Gupta. State estimation in electric power grids: meeting new challenges presented by the requirements of the future grid. *IEEE Signal Processing Magazine*, 29(5):33–43, 2012.
- [33] T. Iwasaki, R.E. Skelton, and J.C. Geromel. Linear quadratic suboptimal control with static output feedback. *Systems & Control Letters*, 23(6):421–430, 1994.

- [34] A. Jadbabaie, J. Lin, and A. S. Morse. Coordination of groups of mobile autonomous agents using nearest neighbor rules. *IEEE Transactions on Automatic Control*, 48(6):988–1001, 2003.
- [35] J. Jiao, H. L. Trentelman, and M. K. Camlibel. Distributed linear quadratic tracking control for leader-follower multi-agent systems: a suboptimality approach. *IFAC Congress* 2020.
- [36] J. Jiao, H. L. Trentelman, and M. K. Camlibel. A suboptimality approach to distributed *H*² optimal control. *IFAC-PapersOnLine*, 51(23):154–159, 2018.
 7th IFAC Workshop on Distributed Estimation and Control in Networked Systems NECSYS 2018.
- [37] J. Jiao, H. L. Trentelman, and M. K. Camlibel. A suboptimality approach to distributed linear quadratic optimal control. *IEEE Transactions on Automatic Control*, 65(3):1218–1225, 2020.
- [38] J. Jiao, H. L. Trentelman, and M. K. Camlibel. Distributed linear quadratic optimal control: compute locally and act globally. *IEEE Control Systems Letters*, 4(1):67–72, 2020.
- [39] J. Jiao, H. L. Trentelman, and M. K. Camlibel. A suboptimality approach to distributed \mathcal{H}_2 control by dynamic output feedback. *Automatica*, 121:109164, 2020.
- [40] J. Jiao, H. L. Trentelman, and M. K. Camlibel. \mathcal{H}_2 and \mathcal{H}_{∞} suboptimal distributed filter design for linear systems. 2020. [Online]. Available: https://arxiv.org/abs/2002.02830v1.
- [41] J. Jiao, H. L. Trentelman, and M. K. Camlibel. *H*₂ suboptimal output synchronization of heterogeneous multi-agent systems. Submitted for publication, 2020.
- [42] Q. Jiao, H. M., F. L. Lewis, S. Xu, and L. Xie. Distributed L₂-gain outputfeedback control of homogeneous and heterogeneous systems. *Automatica*, 71:361–368, 2016.
- [43] Q. Jiao, H. Zhang, S. Xu, F. L. Lewis, and L. Xie. Bipartite tracking of homogeneous and heterogeneous linear multi-agent systems. *International Journal of Control*, 92(12):2963–2972, 2019.
- [44] H.-J. Jongsma, P. Mlinarić, S. Grundel, P. Benner, and H. L. Trentelman. Model reduction of linear multi-agent systems by clustering with H_2 and H_{∞} error bounds. *Mathematics of Control, Signals, and Systems*, 30:6, 2018.

- [45] H.-J. Jongsma, H. L. Trentelman, and M. K. Camlibel. Model reduction of networked multiagent systems by cycle removal. *IEEE Transactions on Automatic Control*, 63(3):657–671, 2018.
- [46] H. Kim, H. Shim, and J. H. Seo. Output consensus of heterogeneous uncertain linear multi-agent systems. *IEEE Transactions on Automatic Control*, 56 (1):200–206, 2011.
- [47] J. Kim, H. Shim, and J. Wu. On distributed optimal Kalman-Bucy filtering by averaging dynamics of heterogeneous agents. In 2016 IEEE 55th Conference on Decision and Control (CDC), pages 6309–6314, 2016.
- [48] J. Kim, J. Yang, H. Shim, J. Kim, and J. H. Seo. Robustness of synchronization of heterogeneous agents by strong coupling and a large number of agents. *IEEE Transactions on Automatic Control*, 61(10):3096–3102, 2016.
- [49] T. Kim, H. Shim, and D. D. Cho. Distributed Luenberger observer design. In 2016 IEEE 55th Conference on Decision and Control (CDC), pages 6928–6933, 2016.
- [50] T. Kim, C. Lee, and H. Shim. Completely decentralized design of distributed observer for linear systems. *IEEE Transactions on Automatic Control*, to appear, 2020.
- [51] J. G. Lee, S. Trenn, and H. Shim. Synchronization with prescribed transient behavior: heterogeneous multi-agent systems under funnel coupling. 2020. [Online]. Available: https://stephantrenn.net/wpcontent/uploads/2019/07/Preprint-LTS190719.pdf.
- [52] Z. Li and Z. Ding. Fully distributed adaptive consensus control of multiagent systems with LQR performance index. In *Proc. IEEE Conference on Decision and Control (CDC)*, pages 386–391, 2015.
- [53] Z. Li and Z. Duan. Cooperative Control of Multi-Agent Systems: A Consensus Region Approach. CRC Press, 2014.
- [54] Z. Li, Z. Duan, G. Chen, and L. Huang. Consensus of multiagent systems and synchronization of complex networks: a unified viewpoint. *IEEE Transactions on Circuits and Systems I: Regular Papers*, 57(1):213–224, 2010.
- [55] Z. Li, Z. Duan, and G. Chen. On H_{∞} and H_2 performance regions of multiagent systems. *Automatica*, 47(4):797–803, 2011.
- [56] Z. Li, X. Liu, W. Ren, and L. Xie. Distributed tracking control for linear multiagent systems with a leader of bounded unknown input. *IEEE Transactions* on Automatic Control, 58(2):518–523, 2013.

- [57] F. Lin, M. Fardad, and M. R. Jovanović. Design of optimal sparse feedback gains via the alternating direction method of multipliers. *IEEE Transactions* on Automatic Control, 58(9):2426–2431, 2013.
- [58] J. Liu, S. Mou, and A. S. Morse. Asynchronous distributed algorithms for solving linear algebraic equations. *IEEE Transactions on Automatic Control*, 63(2):372–385, 2018.
- [59] J. Lunze. Synchronization of heterogeneous agents. *IEEE Transactions on Automatic Control*, 57(11):2885–2890, 2012.
- [60] J. Lunze. Networked Control of Multi-Agent Systems. Edition MoRa. Bookmundo Direct, 2019.
- [61] Y. Lv, Z. Li, Z. Duan, and J. Chen. Distributed adaptive output feedback consensus protocols for linear systems on directed graphs with a leader of bounded input. *Automatica*, 74:308 – 314, 2016.
- [62] J. Mei, W. Ren, and J. Chen. Distributed consensus of second-order multiagent systems with heterogeneous unknown inertias and control gains under a directed graph. *IEEE Transactions on Automatic Control*, 61(8):2019– 2034, 2016.
- [63] M. Mesbahi and M. Egerstedt. Graph Theoretic Methods in Multiagent Networks. Princeton Series in Applied Mathematics. Princeton University Press, 2010.
- [64] M. Milinski and R. Heller. Influence of a predator on the optimal foraging behaviour of sticklebacks (gasterosteus aculeatus l.). *Nature*, 275(5681): 642–644, 1978.
- [65] A. Mitra and S. Sundaram. Distributed observers for LTI systems. *IEEE Transactions on Automatic Control*, 63(11):3689–3704, 2018.
- [66] A. Mitra and S. Sundaram. Byzantine-resilient distributed observers for LTI systems. *Automatica*, 108:108487, 2019.
- [67] H. Modares and F. L. Lewis. Linear quadratic tracking control of partiallyunknown continuous-time systems using reinforcement learning. *IEEE Transactions on Automatic Control*, 59(11):3051–3056, 2014.
- [68] H. Modares, S. P. Nageshrao, G. A. D. Lopes, R. Babuška, and F. L. Lewis. Optimal model-free output synchronization of heterogeneous systems using off-policy reinforcement learning. *Automatica*, 71:334–341, 2016.
- [69] N. Monshizadeh, H. L. Trentelman, and M. K. Camlibel. Projection-based model reduction of multi-agent systems using graph partitions. *IEEE Transactions on Control of Network Systems*, 1(2):145–154, 2014.

- [70] J. M. Montenbruck, G. S. Schmidt, G. S. Seyboth, and F. Allgöwer. On the necessity of diffusive couplings in linear synchronization problems with quadratic cost. *IEEE Transactions on Automatic Control*, 60(11):3029–3034, 2015.
- [71] L. Moreau. Stability of multiagent systems with time-dependent communication links. *IEEE Transactions on Automatic Control*, 50(2):169–182, 2005.
- [72] A. Mosebach and J. Lunze. Synchronization of multi-agent systems with similar dynamics. *IFAC Proceedings Volumes*, 46(27):102–109, 2013. 4th IFAC Workshop on Distributed Estimation and Control in Networked Systems (2013).
- [73] A. Mosebach and J. Lunze. Optimal synchronization of circulant networked multi-agent systems. In *Proc. European Control Conf. (ECC)*, pages 3815–3820, 2013.
- [74] A. Mosebach and J. Lunze. Synchronization of autonomous agents by an optimal networked controller. In *Proc. European Control Conf. (ECC)*, pages 208–213, 2014.
- [75] K. H. Movric and F. L. Lewis. Cooperative optimal control for multi-agent systems on directed graph topologies. *IEEE Transactions on Automatic Control*, 59(3):769–774, 2014.
- [76] D. H. Nguyen. A sub-optimal consensus design for multi-agent systems based on hierarchical LQR. *Automatica*, 55:88–94, 2015.
- [77] D. H. Nguyen. Reduced-order distributed consensus controller design via edge dynamics. *IEEE Transactions on Automatic Control*, 62(1):475–480, 2017.
- [78] W. Ni and D. Cheng. Leader-following consensus of multi-agent systems under fixed and switching topologies. *Systems & Control Letters*, 59(3):209– 217, 2010.
- [79] K.-K. Oh, M.-C. Park, and H.-S. Ahn. A survey of multi-agent formation control. *Automatica*, 53:424–440, 2015.
- [80] A. Okubo. Dynamical aspects of animal grouping: swarms, schools, flocks, and herds. *Advances in Biophysics*, 22:1–94, 1986.
- [81] R. Olfati-Saber. Distributed Kalman filter with embedded consensus filters. In Proceedings of the 44th IEEE Conference on Decision and Control, pages 8179–8184, 2005.

- [82] R. Olfati-Saber. Flocking for multi-agent dynamic systems: algorithms and theory. *IEEE Transactions on Automatic Control*, 51(3):401–420, 2006.
- [83] R. Olfati-Saber. Distributed Kalman filtering for sensor networks. In 2007 46th IEEE Conference on Decision and Control, pages 5492–5498, 2007.
- [84] R. Olfati-Saber. Kalman-consensus filter : optimality, stability, and performance. In Proceedings of the 48h IEEE Conference on Decision and Control (CDC) held jointly with 2009 28th Chinese Control Conference, pages 7036–7042, 2009.
- [85] R. Olfati-Saber and R. M. Murray. Consensus problems in networks of agents with switching topology and time-delays. *IEEE Transactions on Automatic Control*, 49(9):1520–1533, 2004.
- [86] R. Olfati-Saber and J. S. Shamma. Consensus filters for sensor networks and distributed sensor fusion. In *Proceedings of the 44th IEEE Conference on Decision and Control*, pages 6698–6703, 2005.
- [87] R. Olfati-Saber, J. A. Fax, and R. M. Murray. Consensus and cooperation in networked multi-agent systems. *Proceedings of the IEEE*, 95(1):215–233, 2007.
- [88] S. Park and N. C. Martins. Design of distributed LTI observers for state omniscience. *IEEE Transactions on Automatic Control*, 62(2):561–576, 2017.
- [89] C. W. Reynolds. Flocks, herds and schools: a distributed behavioral model. In Proceedings of the 14th Annual Conference on Computer Graphics and Interactive Techniques, SIGGRAPH '87, page 25–34, New York, NY, USA, 1987. Association for Computing Machinery.
- [90] V. Rezaei and M. Stefanovic. Distributed decoupling of linear multiagent systems with mixed matched and unmatched state-coupled nonlinear uncertainties. In *Proc. American Control Conf. (ACC)*, pages 2693–2698, 2017.
- [91] M. Rotkowitz and S. Lall. A characterization of convex problems in decentralized control. *IEEE Transactions on Automatic Control*, 51(2):274–286, 2006.
- [92] S.-H. Wang and E. Davison. On the stabilization of decentralized control systems. *IEEE Transactions on Automatic Control*, 18(5):473–478, 1973.
- [93] T. Sato and K.-Z. Liu. LMI solution to general H₂ suboptimal control problems. Systems & Control Letters, 36(4):295–305, 1999.
- [94] L. Scardovi and R. Sepulchre. Synchronization in networks of identical linear systems. *Automatica*, 45(11):2557 2562, 2009.

- [95] C. Scherer and S. Weiland. *Linear Matrix Inequalities in Control (Lecture Notes)*. Delft: The Netherlands, 2000. [Online]. Available: https://www.imng.unistuttgart.de/mst/files/LectureNotes.pdf.
- [96] C. Scherer, P. Gahinet, and M. Chilali. Multiobjective output-feedback control via LMI optimization. *IEEE Transactions on Automatic Control*, 42(7): 896–911, 1997.
- [97] E. Semsar-Kazerooni and K. Khorasani. Optimal consensus algorithms for cooperative team of agents subject to partial information. *Automatica*, 44(11): 2766–2777, 2008.
- [98] E. Semsar-Kazerooni and K. Khorasani. Multi-agent team cooperation: a game theory approach. *Automatica*, 45(10):2205–2213, 2009.
- [99] G. S. Seyboth, D. V. Dimarogonas, K. H. Johansson, P. Frasca, and F. Allgöwer. On robust synchronization of heterogeneous linear multi-agent systems with static couplings. *Automatica*, 53:392 – 399, 2015.
- [100] R. E. Skelton, T. Iwasaki, and D. E. Grigoriadis. A Unified Algebraic Approach to Control Design. Boca Raton, FL, USA: CRC Press, 1997.
- [101] M. Tillerson, G. Inalhan, and J. P. How. Co-ordination and control of distributed spacecraft systems using convex optimization techniques. *International Journal of Robust and Nonlinear Control*, 12(2-3):207–242, 2002.
- [102] H. L. Trentelman, A. A. Stoorvogel, and M. Hautus. Control Theory for Linear Systems. Springer, 2001.
- [103] H. L. Trentelman, K. Takaba, and N. Monshizadeh. Robust synchronization of uncertain linear multi-agent systems. *IEEE Transactions on Automatic Control*, 58(6):1511–1523, 2013.
- [104] S. E. Tuna. LQR-based coupling gain for synchronization of linear systems.2008. [Online]. Available: https://arxiv.org/abs/0801.3390.
- [105] V. Ugrinovskii. Distributed robust filtering with H_{∞} consensus of estimates. Automatica, 47(1):1–13, 2011.
- [106] V. Ugrinovskii. Distributed robust estimation over randomly switching networks using H_{∞} consensus. *Automatica*, 49(1):160–168, 2013.
- [107] R. Vadigepalli and F. J. Doyle. A distributed state estimation and control algorithm for plantwide processes. *IEEE Transactions on Control Systems Technology*, 11(1):119–127, 2003.

- [108] K. G. Vamvoudakis, F. L. Lewis, and G. R. Hudas. Multi-agent differential graphical games: online adaptive learning solution for synchronization with optimality. *Automatica*, 48(8):1598–1611, 2012.
- [109] H. J. van Waarde, M. K. Camlibel, and H. L. Trentelman. Comments on "On the necessity of diffusive couplings in linear synchronization problems with quadratic cost". *IEEE Transactions on Automatic Control*, 62(6):3099–3101, 2017.
- [110] T. Vicsek, A. Czirók, E. Ben-Jacob, I. Cohen, and O. Shochet. Novel type of phase transition in a system of self-driven particles. *Phys. Rev. Lett.*, 75: 1226–1229, 1995.
- [111] W. Ren and R. W. Beard. Consensus seeking in multiagent systems under dynamically changing interaction topologies. *IEEE Transactions on Automatic Control*, 50(5):655–661, 2005.
- [112] J. Wang, Z. Duan, Z. Li, and G. Wen. Distributed H_{∞} and H_2 consensus control in directed networks. *IET Control Theory Applications*, 8(3):193–201, 2014.
- [113] L. Wang and A. S. Morse. A distributed observer for a time-invariant linear system. *IEEE Transactions on Automatic Control*, 63(7):2123–2130, 2018.
- [114] L. Wang, A. S. Morse, D. Fullmer, and J. Liu. A hybrid observer for a distributed linear system with a changing neighbor graph. In 2017 IEEE 56th Annual Conference on Decision and Control (CDC), pages 1024–1029, 2017.
- [115] P. Wieland, R. Sepulchre, and F. Allgöwer. An internal model principle is necessary and sufficient for linear output synchronization. *Automatica*, 47 (5):1068–1074, 2011.
- [116] W. M. Wonham. Linear Multivariable Control: A Geometric Approach. Springer-Verlag, New York, 1985.
- [117] J. Xi, Y. Yu, G. Liu, and Y. Zhong. Guaranteed-cost consensus for singular multi-agent systems with switching topologies. *IEEE Transactions on Circuits* and Systems I: Regular Papers, 61(5):1531–1542, 2014.
- [118] Q. Yang, M. Cao, H. Fang, and J. Chen. Constructing universally rigid tensegrity frameworks with application in multiagent formation control. *IEEE Transactions on Automatic Control*, 64(1):381–388, 2019.
- [119] S. Zeng and F. Allgöwer. Structured optimal feedback in multi-agent systems: a static output feedback perspective. *Automatica*, 76:214–221, 2017.

- [120] F. Zhang, H. L. Trentelman, and J. M.A. Scherpen. Fully distributed robust synchronization of networked Lur'e systems with incremental nonlinearities. *Automatica*, 50(10):2515–2526, 2014.
- [121] F. Zhang, H. L. Trentelman, and J. M. A. Scherpen. Dynamic feedback synchronization of Lur'e networks via incremental sector boundedness. *IEEE Transactions on Automatic Control*, 61(9):2579–2584, 2016.
- [122] H. Zhang, F. L. Lewis, and A. Das. Optimal design for synchronization of cooperative systems: state feedback, observer and output feedback. *IEEE Transactions on Automatic Control*, 56(8):1948–1952, 2011.
- [123] H. Zhang, T. Feng, G. Yang, and H. Liang. Distributed cooperative optimal control for multiagent systems on directed graphs: an inverse optimal approach. *IEEE Transactions on Cybernetics*, 45(7):1315–1326, 2015.
- [124] K. Zhou, J. C. Doyle, and K. Glover. *Robust and Optimal Control.* New Jersey, NJ, USA: Prentice-Hall, 1996.

Summary

The thesis is concerned with several problems related to distributed linear quadratic control and filtering. In particular, we consider the distributed linear quadratic optimal control problem, the distributed \mathcal{H}_2 optimal control problem, and the \mathcal{H}_2 and \mathcal{H}_∞ optimal distributed filtering problems. Due the the *non-convex* nature of these problem, the *actual* optimization problems are in general very challenging and it is not clear whether closed form solutions exist. Therefore, in this thesis, instead we consider *suboptimality* versions of these problems.

We first consider the distributed linear quadratic suboptimal control problem for *leaderless* linear multi-agent systems. Given a multi-agent system with identical agent dynamics and a global linear quadratic cost functional, we establish a *centralized* design method for computing distributed control laws such that the associated cost is smaller than a given upper bound while the controlled network achieves state synchronization. We then extend these results to the problem of distributed linear quadratic suboptimal tracking control for *leader-follower* multiagent systems. We provide a *centralized* design method for computing distributed suboptimal control laws such that the states of the controlled followers track that of the leader asymptotically.

In the two problems above, the proposed design methods are centralized, i.e. the computation of the local gains depends on so-called *global information*. To remove the dependence on this global information, as the third problem, we aim at *decentralized computation* of the local gains. For multi-agent systems with *single integrator* agent dynamics, we establish a decentralized computation method for obtaining distributed suboptimal control laws, where each agent computes its own local control gains using sampled state information of its neighboring agents.

As the fourth problem, we consider the distributed \mathcal{H}_2 suboptimal control problem for leaderless *homogeneous* multi-agent systems by *static relative state feedback*. Given a multi-agent system and an associated \mathcal{H}_2 cost functional, we provide a centralized design method for obtaining distributed protocols such that the associated \mathcal{H}_2 cost functional is smaller than a given upper bound while the controlled network achieves *state* synchronization. We then generalize the results on distributed \mathcal{H}_2 suboptimal control using static state feedback to the case of *dynamic relative output feedback.* The results on distributed H_2 suboptimal control for state synchronization of homogeneous multi-agent systems are then further generalized to the case of *output* synchronization of *heterogeneous* multi-agent systems.

Finally, we study the \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filtering problems. Given is a linear system monitored by a number of local filters, where each local filter receives only a certain part of the system output. The local filters exchange information about their estimates of the system state with their neighbors according to a given communication graph. We establish conceptual algorithms for obtaining \mathcal{H}_2 and \mathcal{H}_∞ suboptimal distributed filters, in the sense that the \mathcal{H}_2 or \mathcal{H}_∞ norm of the transfer matrix of the disturbance input to the output estimation error is smaller than an a priori given upper bound while all local filters reconstruct the system state asymptotically.

Samenvatting

In dit proefschrift worden verschillende gedistribueerde lineair-kwadratische regelproblemen en filterproblemen bestudeerd. In het bijzonder bestuderen we het gedistribueerde lineair-kwadratische optimale regelprobleem, het gedistribueerde \mathcal{H}_2 optimale regelprobleem en de \mathcal{H}_2 en \mathcal{H}_∞ optimale gedistribueerde filterproblemen. Vanwege de *niet-convexe* aard van deze problemen zijn de *eigenlijke* optimalisatieproblemen over het algemeen erg uitdagend en is het niet duidelijk of er oplossingen in gesloten vorm bestaan. Daarom bestuderen we in dit proefschrift *suboptimale* versies van deze problemen.

Het eerste probleem waar we ons op richten is het gedistribueerde lineairkwadratische suboptimale regelprobleem voor lineaire multi-agent systemen *zonder leiders*. Voor een multi-agent systeem met identieke agent-dynamica en een globale lineair-kwadratische kostenfunctionaal stellen we een *gecentraliseerde* ontwerpmethode vast om gedistribueerde regelwetten te berekenen, zodanig dat de geassocieerde kosten kleiner zijn dan een gegeven bovengrens, terwijl het geregelde netwerk toestandsynchronisatie bereikt. Vervolgens breiden we deze resultaten uit naar het gedistribueerde lineair-kwadratische suboptimale tracking regelprobleem voor *leider-volger* multi-agent systemen. We bieden een *gecentraliseerde* ontwerpmethode om gedistribueerde suboptimale regelwetten te berekenen zodanig dat de toestanden van de geregelde volgers die van de leider asymptotisch tracken.

In de twee hierboven beschreven problemen zijn de voorgestelde methoden gecentraliseerd, dat wil zeggen dat de berekening van de lokale versterkingsfactoren afhankelijk is van zogenaamde *globale informatie*. Om de afhankelijkheid van deze globale informatie te omzeilen, streven we bij wijze van het derde probleem naar *gedecentraliseerde berekeningen* van de lokale versterkingsfactoren. Voor multi-agent systemen met *enkelvoudige integrator* agent-dynamica stellen we een gedecentraliseerde methode vast om gedistribueerde suboptimale regelwetten te verkrijgen, waar elke agent zijn eigen lokale regel versterkingsfactoren berekent met behulp van bemonsterde toestandinformatie van de aangrenzende agenten.

Het vierde probleem dat wordt onderzocht is het gedistribueerde \mathcal{H}_2 suboptimale regelprobleem voor *homogene* multi-agent systemen zonder leiders door middel van *statische relatieve toestandsterugkoppeling*. Voor een multi-agent systeem en een bijbehorende \mathcal{H}_2 kostenfunctionaal bieden we een gecentraliseerde ontwerpmethode om gedistribueerde protocollen te verkrijgen zodanig dat de bijbehorende \mathcal{H}_2 kostenfunctionaal kleiner is dan een gegeven bovengrens, terwijl het geregelde netwerk *toestandssynchronisatie* bereikt. Vervolgens generaliseren we de resultaten van gedistribueerde \mathcal{H}_2 suboptimale regeling met behulp van statische toestandsterugkoppeling naar het geval van *dynamische relatieve uitgangsterugkoppeling*. De resultaten van gedistribueerde \mathcal{H}_2 suboptimale regeling voor toestandssynchronisatie van homogene multi-agent systemen worden vervolgens verder gegeneraliseerd naar het geval van *uitgangssynchronisatie* van *heterogene* multi-agent systemen.

Tenslotte bestuderen we de \mathcal{H}_2 en \mathcal{H}_∞ suboptimale gedistribueerde filterproblemen. Gegeven is een lineair systeem dat gemonitord wordt door een aantal lokale filters, waar elk lokaal filter slechts een bepaald gedeelte van de uitgang van het systeem ontvangt. De lokale filters wisselen informatie over hun schatting van de toestand van het systeem uit met die van de aangrenzende filters volgens een gegeven communicatiegraaf. We stellen conceptuele algoritmes vast voor het verkrijgen van \mathcal{H}_2 en \mathcal{H}_∞ suboptimale gedistribueerde filters, in die zin dat de \mathcal{H}_2 of \mathcal{H}_∞ norm van de transfermatrix van de verstoringsingang naar de uitgangsschattingsfout kleiner is dan een a priori gegeven bovengrens, terwijl alle lokale filters de systeemtoestand asymptotisch reconstrueren.